

Water and carbon fluxes in young plantations: observations and modelling

Marcela Rafaela de Freitas Silva

BSc (Hons) in Environmental Engineering.

A thesis submitted for the degree of Doctor of Philosophy at Monash University in 2022

Department of Civil Engineering

Copyright notice

©Marcela Rafaela de Freitas Silva (2022)

I certify that I have made all reasonable efforts to secure copyright permissions for third-party content included in this thesis and have not knowingly added copyright content to my work without the owner's permission.

Abstract

The establishment and expansion of commercial plantations for timber production and carbon sequestration raise concerns because of their large water use. The dearth of high resolution experimental observations and knowledge of the dynamics of water use and growth in plantations, especially in their early years after establishment, makes it difficult to develop better management practices. Investigating ecosystem responses to environmental variables can provide insights on how plantation trees adjust their water use according to local conditions.

Combining field measurements and hydrodynamic modelling, this project provides a unique and comprehensive quantification of the trade-offs between water use and carbon assimilation in a *Eucalyptus globulus* (blue gum) plantation in the first 4 years after establishment. The blue gum is one of the most planted species globally, as it grows rapidly and is adaptable to a range of climatic conditions. The experimental measurements were performed in a study site located in southwest Victoria, Australia, where energy and CO_2 fluxes were continuously measured above the tree canopy for 3.5 years after the trees were planted. During the first year after establishment, understory vegetation and ecosystem respiration had a major impact on the net ecosystem exchange (NEE), with the plantation being a net carbon source. Subsequently, the trees started dominating the contributions to NEE and, after approximately 2 years, the plantation became a consistent carbon sink. These shifts in NEE were accompanied by smaller increases in annual evapotranspiration rates, which was 70% of the annual precipitation in the first year of measurements. This shows a remarkable increase in productivity at the expense of a small amount of water.

To further understand and estimate transpiration dynamics in plantation trees, a tree hydrodynamic model (FETCH3) was developed. FETCH3 assumes that the water flow through the soil, roots, and above-ground xylem can be approximated as flow in porous media. Through the application of the Richardson-Richards equation, FETCH3 describes the transport of water through the plant system by applying a rigorous coupling of the soil, roots xylem, and stem xylem pathways. The numerical scheme of the model, developed in Python 3, was verified and produced considerably small errors against exact analytical solutions for steady state and transient conditions using simplified but realistic model parametrizations. The model was also used to simulate a previously published case study where observed transpiration rates were available and the results were in agreement with observations. Finally, the model was calibrated using transpiration data collected from sap flow sensors installed in growing blue gum trees in the study site.

The results from this project emphasized the highly dynamic water and carbon fluxes in young and recently established plantations trees. Specifically, the continued cultivation of large scale commercial plantations would benefit from management practices that are inclusive of information and data from all the years of the management cycle, including the first few years after establishment.

Declaration

This thesis is an original work of my research and contains no material which has been accepted for the award of any other degree or diploma at any university or equivalent institution and that, to the best of my knowledge and belief, this thesis contains no material previously published or written by another person, except where due reference is made in the text of the thesis.

Signature:

Print Name: Marcela Rafaela de Freitas Silva

Date: 27th July 2022

Publications during enrollment

- Silva, M., Matheny, A. M., Pauwels, V. R. N., Triadis, D., Missik, J. E., Bohrer, G., and Daly, E.: Tree hydrodynamic modelling of the soil plant atmosphere continuum using FETCH3, Geosci. Model Dev., 2022. (I.F. 7.708, Q1 in Earth and Planetary Sciences and Model and Simulation)
- Silva, M., McHugh, I., Peixoto Neto, A., Pauwels, V.R.N., Cartwright, I., Daly, E.: Trading a little water for substantial carbon gains during the first years of a Eucalyptus globulus plantation, Agricultural and Forest Meteorology. (I.F. 5.734, Q1 in Agronomy and Crop Science, Atmospheric Science, Forestry, and Global and Planetary Change)

Acknowledgements

I am very grateful to everyone that helped me throughout my PhD journey. The continuous support from colleagues and friends has been a major key to the completion of my project.

I would like to start by expressing my gratitude for my supervisors, Edoardo Daly and Valentijn Pauwels, who guided me to improve not only as a researcher, but whose work ethic and dedication assisted me in growing as an individual. Their support and patience were fundamental for the completion of this project. They always provided encouragement, even when my research found some dead-ends throughout the years.

I would like to express sincere gratitude to my parents, Marcelo (*in memoriam*) and Catia, for all the efforts and sacrifices they made for my education. Although they did not have the opportunity of higher education themselves, they always encouraged me in my studies. Both my parents are examples of hardworking, brave, and honest people, who have truly inspired me to become a better person.

I would like to thank my husband, Akira Kobayashi, for his unconditional love and support during my PhD. His encouragement and kindness motivated me during the hard moments of research and living abroad. Despite experiencing his own difficulties and obstacles, he was always by my side and ready to step in when I needed him the most. Next, I would also like to extend my gratitude to my brother Marcelo. It feels very special to have a sibling who is not only one of my best friends, but an exceptional colleague in the same field of research. During my PhD, not only I could count with him as family member and friend, but also with assistance for my coding and manuscript writing.

During my years at Monash University, I was able to learn and make friends and colleagues who helped me during my PhD. I would like to thank Kiri Mason, for her continuous assistance with field trips and equipment set up. Her companionship was able to make the 4.5 hour commute to the study site enjoyable and fun. A special thanks goes out to Valentina Marchionni, who I consider a great friend and mentor. She was always available to help and advise me. Valentina is someone that I will always look up to and admire. I would also like to thank my friends and colleagues from the postgraduate research at Monash University. An honourable mention must go to the following: Alberto Peixoto, Chi Nguyen, Gonzalo Guzman, Jihane Elyahyioui, Filomena Innella, Sabah Sabaghy, Simone Gelsinary, Jayaram Pudashine, and Ali Azarnivand.

My gratitude also goes to my friends in the Brazilian community, especially Flavia Medeiros and Pedro Bau, who were our first friends in Melbourne and who continue to provide us with invaluable support and friendship. I am also very grateful to my friends at Bayside Bjj, of note Sam Jeffrey, Dylan Minster, Peter Polygenis, James Talbot, and Matthew Greskie, whose priceless friendship helped us to integrate in the Australian community and culture.

Lastly, I would like to acknowledge Monash University for the scholarships I was awarded (MGS and MITS), which supported my living expenses and tuition fees. I was also supported by the Graduate Research Completion Award.

Table of Contents

Ał	ostra	xt ii
De	eclara	iii
Ρι	ublica	itions
Ac	cknov	vledgements v
Та	ble o	f Contents vi
Li	st of	Figures viii
Li	st of	Tables xi
1	Intro	aduction 1
2	Lite 2.1 2.2 2.3 2.4 2.4	rature Review4Impact of afforestation on rural catchments4Commercial plantations in Australia6Water and carbon fluxes in productive ecosystems92.3.1 Drivers of water and carbon fluxes in plantations112.3.2 Water use efficiency of plantations132.3.3 Water use and carbon assimilation in young plantations14Modelling plantation water use202.4.1 Zhang curves202.4.2 Water use models in forest growth models212.4.3 Tree Hydrodynamic modeling22Scope of research29
3	Wat 3.1 3.2 3.3 3.4 3.5	Pr and carbon fluxes in a young Eucalyptus globulus plantation32Aim32Materials and Methods333.2.1Study site333.2.2Data description343.2.3Data analysis36Results383.3.1Energy fluxes383.2.2Water and carbon fluxes39Discussion433.4.1Water and carbon fluxes with stand age433.4.2From carbon source to sink453.4.3Water use efficiency47Conclusion49

4	Tree 4.1 4.2	hydrodynamic model development: FETCH3 Aim Model description 4.2.1 Model overview 4.2.2 Governing equations 4.2.3 Boot water uptake and transpiration	50 51 51 52 55
	4.3	Numerical scheme	57 58 63
	4.4 4.5 4.6	Penman-Monteith transpiration	66 68 68 68 68 69
5	FET	CH3 model applications	71
	5.1 5.2	Aim	/1 72 73 75
	5.3	5.2.2 Steady-state solution Model application: case study	75 76 77 78
	5.4 5.5	5.3.3 Results And water capacitance Modelling LAD and water capacitance Model application: calibration against measured sap flux data 5.5.1 Data description 5.5.2 Model setup 5.5.3 Automatic calibration using PSO 5.5.3	78 81 82 83 83 84 86
	5.6	5.5.4 Results	86 90
6	Con 6.1	clusion Summary of results 6.1.1 Experimental work	91 91 91
	6.2 6.3 6.4	6.1.2 FETCH3 FETCH3 Contributions of research Practical Implications Practical Implications Feter Limitations and future research Feter 6.4.1 Experimental work 6.4.2 FETCH3	92 92 93 94 94 95
_			~-

References

List of Figures

2.1	Map of the National Plantation Inventory (NPI) regions for 2019-20. Retrieved	
	from Downham and Gavran (2020).	7
2.2	Annualised groundwater recharge reductions for blue gum plantations due to	
	interception of recharge, defined as the water entering the water table from re-	
	maining annual rainfall, after evapotranspiration and run-off are accounted for,	
	according to the 2006 South Australian model. Negative interception reflects an	
	agreed allowance for recharge (120%) compared to dryland pasture reference	
	levels. Modified from Greenwood (2013).	8
2.3	Schematic view of the NEE partition between GPP and ER	10
2.4	Overview of the research activities	31
3.1	Location of the study site, and photos of the trees and the eddy covariance tower	
	in a) December 2017, when the system was installed, and b) July 2021, at the	
	end of the monitoring period.	33
3.2	Total daily solar radiation (${ m MJ}~{ m m}^{-2}~{ m d}^{-1}$) and rainfall (P, mm d $^{-1}$), and daily av-	
	erages of the soil water content (SWC, %), air temperature (T_{air} , °C), and Va-	
	pour Pressure Deficit (VPD, kPa). Vertical dashed lines represent the transition	
	between years.	34
3.3	Top: daily mean net radiation (Q), sensible heat (H), and latent heat fluxes (LE)	
	$({ m W~m^{-2}})$ throughout the monitored period. Bold lines represent weekly moving	
	averages. Bottom: weekly moving averages of Bowen ratio (β). Vertical dashed	
	lines represent transition between years	39
3.4	Daily total evapotranspiration (ET, ${ m mm}~{ m d}^{-1}$) and mean daily net ecosystem ex-	
	change (NEE, $\mathrm{gCm^{-2}d^{-1}}$) during 2018, 2019, and 2020. Thick lines represent	
	weekly moving averages. The horizontal dashed line represents carbon neutrality.	40
3.5	Diurnal courses (hourly means) of evapotranspiration (ET, ${ m mm}~{ m h}^{-1}$) and net eco-	
	system exchange (NEE, $\mu { m molCO_2 m^{-2} s^{-1}}$) across Summer (Dec - Feb), Autumn	
	(Mar - May), Winter (Jun - Aug), and Spring (Sep - Nov). The horizontal dashed	
	line represents carbon neutrality. Vertical bars represent the standard error	41

3.6	Daily accumulation of gross primary production (GPP), modeled net ecosystem	
	exchange (NEE), and ecosystem respiration (ER) ($gCm^{-2}d^{-2}$). Shaded areas	
	respiration measurements	12
27	Monthly accumulation of gross access to production (GEP $_{\rm eff}$ $_{\rm eff}$ $_{\rm eff}$ $_{\rm eff}$	42
5.7	and evapotranspiration (ET $mm month^{-1}$) and water use officioney (WLIE) ($gC \log^{-1}$	1
	for the study site	201 12
38	Cumulative Net Ecosystem Production (NEP $_{\rm eff}$ m ⁻²). Gross Ecosystem Pro-	72
0.0	duction (GEP $gC m^{-2}$) Ecosystem Respiration (FB $gC m^{-2}$) and Evapotran-	
	spiration (ET mm) for 2018 2019 and 2020. The horizontal line represents	
	carbon neutrality	44
39	Monthly EB/GEP ratio. The horizontal line at EB/GEP=1 represents the trans-	
0.0	ition between carbon source (ER/GEP > 1) and sink (ER/GEP < 1) behaviour	
	Vertical dashed lines represent transitions between years	46
4.1	Representation of the coupling process between soil, root xylem, and stem xy-	
	lem applied in the model, where A_s represents a reference ground area, dz an	
	infinitesimal depth over an area (m), z the vertical coordinate (m), V volume of	
	soil (m ³), ρ the density of water (kg m ⁻³), F_{in} (kg s ⁻¹) the water fluxes entering	
	and F_{out} (kg s ⁻¹) exiting the volume, A_r/A_s (m ² _{root} m ⁻² _{ground}) the root xylem cross	
	area index, A_x/A_s ($m_{xylem}^2 m_{ground}^{-2}$) the stem xylem cross area index, S (s ⁻¹) the	
	rate at which water is extracted from the soil and enter the root xylem, and S_x	
	$(m^2 s^{-1})$ is the flow of water leaving the stem per unit of vertical length due to	
		53
4.2	Representation of the model domain	57
5.1	Summary of model applications described in the chapter.	71
5.2	Left: water potentials (MPa) from the exact (lines) and numerical solutions (dots)	
	using the sink term in Eq. (5.10) for the first 12 hours. For better visualization	
	not all points are shown for the numerical solution. Right: difference between	
	the exact and numerical solution (Δ) at 3 m and 6 m. The temporal and spatial	
	resolutions are 0.05 h and 0.01 m, respectively.	74
5.3	Left: water potentials (MPa) from the exact (lines) and numerical solutions (dots)	
	using the sink term in Eq. (5.11) for the first 12 hours. For better visualization	
	not all points are shown for the numerical solution. Right: difference between	
	the exact and numerical solution (Δ) at 3 m and 6 m. The temporal and spatial	
	resolutions are 0.05 h and 0.01 m, respectively.	75

5.4	Left: water potentials, Φ , (MPa) at steady state obtained from the exact (black
	line) (Eq. 5.16), and numerical solutions (dots), using 0.05 m and 0.08 h as
	spatial and temporal resolution, respectively. For the numerical solution, not all
	points are shown for better visualization. The lines with light colors present the
	initial condition and the first 2 hours of simulation. Right: difference between the
	exact and numerical solution (Δ) at steady state condition along tree height

5.5	a) Comparison between measured $(\rm T_{obs})$ and modelled $(\rm T_{mod})$ daily sap flux	
	rates excluding fluxes during the night. b) Root and stem xylem water potential	
	(MPa) as a function of elevation (z) at midday. The vertical position of 5 m (above	
	z=0, which is defined as the bottom of the soil column) represents the interface	
	between between the roots and the stem.	80

83

between $r_{\rm obs}$ and $r_{\rm mod}$ daily sap nux rates for the entire simulation period ex-	
cluding fluxes during the night, and c) comparison between $\rm T_{obs}$ and $\rm T_{mod}$ for	
the entire simulation period.	89

77

List of Tables

2.1	List of literature on young and developing plantations and a summary of their findings.	17
2.1	List of literature on young and developing plantations and a summary of their findings.	18
2.1	List of literature on young and developing plantations and a summary of their findings.	19
2.2	Key studies and application of the different modelling approaches for forest growth and water use	26
2.2	Key studies and application of the different modelling approaches for forest growth and water use	27
2.2	Key studies and application of the different modelling approaches for forest growth and water use	28
3.1	Date, Age (years), average Diameter at Breast Height (DBH, mm), Tree height (m), Leaf Area Index (LAI, $m^2 m^{-2}$), and Tower height (m) during the study period. Tower height refers to the height of the EC tower after being extended.	36
3.2	Annual carbon fluxes (modeled and measured NEP, GEP, and ER), ($gC m^{-2} y^{-1}$), evapotranspiration (ET, mm y ⁻¹) and percentage of ET to Precipitation (P y ⁻¹) for 2018, 2019, and 2020. The values are presented without estimating the missing data in incomplete months, and using the previous year to estimate missing	
		43
5.1	List of parameters used in the comparison between the exact and numerical solutions (section 3.1)	76
5.2	List of soil parameters used in the model application	78
5.3	List of parameters used in the application of the model as in Verma et al. (2014)	79
5.4	List of parameters used in the application of the model considering a water ca-	
	pacitance and a leaf area density function	84
5.5	List of parameters values used in the application of FETCH3 for the Digby plant-	
	ation	85
5.6	List of calibrated values of parameters for each simulation	87

5.7	Calibration and Simulation period, with the respective RMSE $(\mathrm{mm}~0.5\mathrm{h}^{-1})$ from	
	the calibration with PSO and R^2 for the simulation period. \ldots \ldots \ldots \ldots	87

Chapter 1

Introduction

The area planted with forests has increased world-wide by 123 million hectares since 1990 and covered 294 million hectares in 2020. Approximately 45 percent of the planted forests are plantations, which are intensively managed forests, mainly composed of one or two tree species, native or exotic, of equal age, planted with regular spacing and mainly established for productive purposes (Downham and Gavran, 2020; FAO and UNEP, 2020). Plantations have expanded in order to supply the demand for timber production and as a tool for carbon storage (Benyon et al., 2009; Landsberg and Waring, 1997). Despite their economic importance, large-scale plantations have been criticised because of the higher transpiration and interception rates associated with trees, resulting in a greater water use and soil nutrient depletion (David et al., 1997; Jackson, 2005; Tfwala et al., 2019). Especially in semi-arid regions, limited precipitation calls for regulations of water allocated to commercial plantation activity, creating tensions between land owners, plantation companies, and water planners (Greenwood, 2013; White et al., 2016).

In Australia, where commercial plantations cover approximately 2 million hectares, fast-growing species, such as the *Eucalyptus globulus* (blue gum), are planted for hardwood production (Downham and Gavran, 2020). Blue gums are one of the world's most planted and economically important hardwood species (Tomé et al., 2021). The environmental impact of afforestation with fast-growing species is not clear yet due to limited studies measuring tree water use during the entire management cycle (Benyon et al., 2006; Cabral et al., 2011; Cleverly et al., 2020; Forrester et al., 2006; Hinko-Najera et al., 2017). There are even fewer studies dedicated to provide knowledge regarding growth, water use and water use efficiency of young plantation stands (Dye and Versfeld, 2007; Dye, 2000).

In this context, knowledge on plantation water and carbon fluxes can assist in determining the dynamics of plantation carbon gains against the amount of water used by trees (White et al., 2021). As such, there is a need for a better understating of tree water use, in particular intradaily and intra-seasonal temporal scales, in order to understand and quantify how trees respond

CHAPTER 1. INTRODUCTION

to differences in environmental conditions (Cleverly et al., 2020; Migliavacca et al., 2021). The estimation of water and carbon trade-offs can be critical for agricultural management decisions and to predict how climate events, such as heatwaves or intense rainfall, can influence the physiology, productivity, and water use of *Eucalyptus* forests (Baldocchi et al., 2018). This information is useful for both plantation companies, which can increase productive rates and profit, and water agencies, which can organize governmental license procedures (Benyon et al., 2009; Greenwood, 2013).

Therefore, this project aims to improve the understanding and provide realistic estimations of plantation water use and carbon assimilation, bringing additional knowledge through experimental measurements and modelling of tree water use dynamics. Specifically, this project focuses on the relationship between afforestation, environmental variables, and water use in young and fast growing plantations, since it is still undetermined and sparse in literature (Chan et al., 2018; Ma et al., 2019).

Chapter 2 of this thesis contains a literature review on the dynamics of plantation water and carbon fluxes. The chapter starts reviewing the impacts of plantation activity on the water balance of rural catchments, followed by the description of the development of the plantation industry in Australia. Next, the literature review discusses the drivers of water and carbon fluxes and the water use efficiency in plantation ecosystems, lastly focusing on young and recently established plantations. A review of modelling approaches to assess plantation water use is also included, with the tree hydrodynamic modelling being highlighted as an alternative to represent intra-daily dynamics of transpiration and to realistic represent plant traits.

Chapter 3 presents the results of measurements of energy and carbon fluxes in a young *Eucalyptus globulus* plantation during its first 4 years after establishment. The fluxes were measured using the eddy covariance method with the equipment raised above the canopy as the trees grew; additional measurements of tree height and diameter at breast height were taken to monitor tree growth. The different dynamics in water and carbon fluxes for young stands captured described in this Chapter highlighted the need to evaluate stand age and species-specific characteristics.

Chapter 4 introduces the The Finite-difference Ecosystem-scale Tree Crown Hydrodynamics (FETCH3), a ready-to-use open-access model that simulates the water fluxes across the soilplant-atmosphere continuum. The chapter describes the derivation of the governing equations and provides the detailed numerical scheme for the solution of the three coupled partial differential equations defining the water fluxes across the system. The numerical solution includes an improved coupling between the soil, roots, and stem xylem. Lastly, the different formulations for the transpiration, root water uptake, hydraulic conductance, and capacitance functions are presented.

Chapter 5 evaluates FETCH3 against different applications. FETCH3 was tested against ana-

CHAPTER 1. INTRODUCTION

lytical solutions (unsteady and steady state formulations) and against a published case study, when the modeled transpiration was compared to observed sap flux data over a period of 5 months. Lastly, the Chapter describes the calibration process against measured sap flow data through the application of the Particle Swarm Optimization (PSO) algorithm.

Chapter 6 summarizes this thesis. The chapter outlines the results of the study, both in terms of experimental measurements and FETCH3 modelling applications, highlighting the contributions of the research and providing several suggestions for future research efforts.

Chapter 2

Literature Review

2.1 Impact of afforestation on rural catchments

The conversion of land from native or agricultural vegetation to commercial plantations is occurring in several countries, and over extensive areas of the southern hemisphere (Farley et al., 2005; Saadaoui et al., 2017). Since the 1950's, the establishment of commercial plantations has expanded, especially the genus *Eucalyptus*, which is the most common plantation worldwide (Tomé et al., 2021). Plantations areas expanded in several countries, including Brazil (Almeida et al., 2007), India (Calder, 2007), China (Chan et al., 2018; Ma et al., 2018), Chile (Acuña et al., 2018), South Africa (Dye, 2013, 2000), and Australia (Almeida et al., 2010; Downham and Gavran, 2020; Forrester et al., 2010a).

Interest has been growing for high productive tree species, given the potential for pulp and paper industries development (Calder, 2007). Another contributor for plantation expansion is the market incentives to reduce CO₂ atmospheric concentrations through carbon-sequestering projects, which include afforestation and reforestation (Fatichi et al., 2019). Additionally, commercial plantations generate a source of income for rural areas (Dye, 2000). However, expansion of commercial plantations has raised controversy because of the high water use associated with short-rotation plantations, especially in semi-arid areas (Greenwood, 2013).

Land-use change due to afforestation has the potential to affect ecosystem processes, such as transpiration, interception, and evaporation, and possibly alter stream flows and ground-water recharge (Brown et al., 2005, 2015; Dye, 2013; Jackson, 2005). These changes can subsequently alter catchment water budgets, supply, and quality (Dresel et al., 2018). To clarify the magnitude and the links between afforestation and water yield, paired catchment studies are commonly used to investigate the impacts of land-use change due to afforestation or deforestation on the catchment water yield.

Paired catchment studies estimate alterations on stream flow due to increases in evapotran-

CHAPTER 2. LITERATURE REVIEW

spiration and interception rates by the vegetation cover (Brown et al., 2005; Dean et al., 2016; Dye, 2013). They are based on two catchments with similar characteristics, such as slope, soils, climate and vegetation, and commonly a short distance from each other. After a period when both sites are monitored, known as calibration period, one catchment land cover is modified. Hence, this type of analysis is useful to understand how streamflow, as a consequence the water balance, is altered in comparison to different treatments and vegetation cover (Brown et al., 2005).

Several paired catchment studies have examined streamflow or evapotranspiration changes caused by afforestation. However, the relationship between land use and streamflow is still not completely understood, with studies documenting considerable streamflow reduction with afforestation (Adelana et al., 2014; Bonnesoeur et al., 2019; Brown et al., 2005; Vertessy et al., 2001; Zhang et al., 2011, 2012) and, contrarily, other studies not finding any significant changes after plantation establishment (Brown et al., 2015; Dean et al., 2016; Dresel et al., 2018; Hamilton et al., 2018). For example, Brown et al. (2015) found that, at regional scales, streamflow was mainly driven by climate variability, rather than land use and land use change. Likewise, Dean et al. (2016) found that the streamflow in a catchment forested with a Eucalyptus plantation did not change considerably when compared to a catchment covered by a pasture for grazing. Contrastingly, when studying data from 26 catchments around the world, Farley et al. (2005) found that afforestation could reduce the streamflow by 50% in catchments where the streamflow represents less than 30% of precipitation or dry streams where streamflow is 10% of precipitation. Similarly, Jackson (2005) analysed approximately 504 annual catchment observations and concluded that afforestation decreased streamflow after a few years of planting, and also caused soil nutrient scarcity.

The contrasting results from different climates and regions makes it challenging to define unified management practices and water regulation policies. Among the factors cited as being responsible for the lack of agreement within the results are the small percentage of land-use change (less than 20% of the catchment), and lack of study sites located in intermittent streams and low precipitation areas (< 800 mm/year) (Brown et al., 2005), and limited information on water yields at a seasonal scale (Dye, 2013, 2000). Additionally, paired catchment analyses usually investigate mean annual changes in the water yield, which result in less information about the effects of afforestation on monthly and seasonal time scales (Brown et al., 2015; Dresel et al., 2018). The changes occurring at the seasonal and intra-annual scale are important since rainfall frequency, weather conditions, management cycle, and practices are relevant site-specific parameters with high variability within a year (Dye, 2013). Changes in these parameters can alter the ecosystem dynamics and, consequently, streamflow (Adelana et al., 2014).

As the demand of water for productive plantations is growing over time, the monitoring of hydrological variables at relatively high temporal and spatial resolutions is increasing in importance (Dean et al., 2016; Dye, 2013). For example, during the first years after plantation establishment, studies found considerable variability in water yield. During this stage there is more recharge and soil exposure, given the shallow roots and shorter trees (Ferraz et al., 2019; Forrester et al., 2010a; Liu et al., 2017), which can result in more streamflow. Other studies have reported variable plantation water use due to different soil conditions, geological and topographical features (Dresel et al., 2018), and right after tree harvesting (Rodrigues et al., 2011). The length of the rotation has been also evaluated, and shorter rotation periods (i.e. 6-7 years in Brazilian *Eucalyptus*) may not allow soil water recharge and nutrient cycling (Christina et al., 2016), and probably result in high tree water use at any stage (Almeida et al., 2007; Hubbard et al., 2010).

The dearth of experimental observations on plantations considering site-specific studies, seasonal and inter- and intra-annual differences in catchment dynamics makes it difficult to estimate and predict the relationship between land use and water resources (Dean et al., 2016). Therefore, extensive research on plantation dynamics is required to fully understand the impact of afforestation on water resources (Adelana et al., 2014; Dye, 2013).

2.2 Commercial plantations in Australia

Australia is mostly a semi-arid country, with limited water availability. This limitation requires an investment into researching sustainable management practices to allow economical growth without harming surface and groundwater resources (Benyon et al., 2006; Greenwood, 2013). In some parts of Australia, native deep-rooted vegetation was cleared to open space for crop production and pastures (Benyon and Doody, 2004), resulting in an increase of water in the ecosystem (when compared to the native vegetation), leading to soil salinization due to shallower water tables (Daneshmand et al., 2020; Rengasamy, 2006). As a consequence, in addition to the industrial and economical incentives, commercial plantations were encouraged in an attempt to solve salinization problems. However, after a long drought period, known as the Millennium Drought, water availability issues were prioritized over the salinity problems (van Dijk et al., 2013).

Until the early 1990s, most of the *Eucalyptus* were harvested from native forest (Acuña et al., 2018; Harper et al., 2014). As demand for pulp increased along with the incentives to expand plantation areas, large areas of *E. globulus* and *E. nitens* were established between the 1990s and 2010 (Benyon and Doody, 2004). The latest plantation statistics show that the total commercial plantation area in Australia is 1,933,400 hectares, with Victoria having the largest area (418,500 hectares) (Figure 2.1). The *E. globulus*, known as the Tasmanian blue gum, is the most planted hardwood specie and represents 50% of all hardwood plantations, while the radiata pine is the most planted softwood tree (75% of all softwood plantations) (Downham and Gavran, 2020).



Figure 2.1: Map of the National Plantation Inventory (NPI) regions for 2019-20. Retrieved from Downham and Gavran (2020).

Despite the majority of commercial plantations being located in Victoria, the state of South Australia was the first to consider commercial plantation as a water interception activity. This led to the introduction of the first Australian plantation water policy in 2011 (Dye, 2013; Greenwood, 2013). This initiative was mainly motivated by the scarcity of water resources in the state, where most of the water comes from groundwater, with limited superficial water (Dye, 2013). Australia was the second country in the world to regulate water used by forests, only behind South Africa (Gush et al., 2002). Greenwood (2013) provides a comprehensive overview of the development of the Australian policy, including a description of the water accounting models used as a baseline for the South Australian policy. Considering the concern with groundwater recharge, defined as the water entering the water table from annual rainfall after evapotranspiration and run-off are accounted for. The legislation is applied on a interception rate of groundwater recharge, calculated as a annualised average over the management cycle. In 2006, the legislation was reviewed and started to include direct groundwater extraction, as there was evidence that trees could use groundwater (Benyon et al., 2006).

In the reviewed legislation, direct groundwater extraction happens when water tables are less than 6 m deep and after canopy closure is reached. The interception rates are assumed 78% per year, as an annualised average applied considering an 11 years management cycle. Interception rates are negative during the first year, reflecting an agreed allowance for recharge

(120%) compared to dryland pasture reference levels, 20% on the second year, 60% on the third year, and 100% starting on the fourth year, when canopy closure is assumed (Figure 2.2). This approach limits the estimation of the plantation impacts on the regional water resources, since applying an annualised average during the entire management cycle may not reflect a realistic water use by the trees and result in underallocation or overallocation of resources. In addition, the South Australian forestry policy was based on a limited dataset, with a few monitoring sites and limited monitored period (Benyon et al., 2009), which introduced uncertainties in the results used to provide the policy guidelines (Greenwood, 2013).



Figure 2.2: Annualised groundwater recharge reductions for blue gum plantations due to interception of recharge, defined as the water entering the water table from remaining annual rainfall, after evapotranspiration and run-off are accounted for, according to the 2006 South Australian model. Negative interception reflects an agreed allowance for recharge (120%) compared to dryland pasture reference levels. Modified from Greenwood (2013).

Considering that planning the allocation of water licenses for plantation activity depends on modelling, the management of commercial plantations and lessen water conflicts between landowners and the government require efficient modelling of the hydrological effects of forestry on the water resources (Dye and Versfeld, 2007; Greenwood, 2013; Gush et al., 2002). Model accuracy can be critical to not only protect the regional water resources, but also assure that the landowners are paying a fair and well based value for water and that licenses are distributed in an efficient way (Benyon et al., 2009; Benyon and Doody, 2004). As a consequence, the study of plantation water use and related ecosystem scale variables (e.g., evapotranspiration, interception, streamflow, water use efficiency) is required to better understand the variability and magnitude of plantation establishment impacts on the water resources (Dean et al., 2016; Scott, 2005), which will, in turn, benefit the water license allocation.

The current plantation management practices aim to achieve maximum productivity through successive plantings and harvests. Since plantation productivity is known to decrease with stand age (Bergeron et al., 2008; Clark et al., 2004; Ryan et al., 2004, 2010), rotations prioritize

maintaining fast growing trees such as *Eucalyptus globulus* and *Eucalyptus grandis*. However, high productivity can come at the expense of higher water use. When enough water is available to maintain high transpiration rates, it leads to considerable stem growth over the year, which makes *Eucalyptus* efficient trees (Cristiano et al., 2020; Dye, 2000).

Most plantation species, including blue gum trees, can adapt to the wide ranges of environmental conditions found in Australia (Drake et al., 2009), such as different geology (Dean et al., 2016), precipitation regimes (Bleby et al., 2012), temperature (Griebel et al., 2020), and salinity (Barrett et al., 2005; Beverly et al., 2005). Specifically, plantation evapotranspiration, interception, and growth can be highly variable over time and space (Almeida et al., 2007; Forrester et al., 2006). The non-linear relationship between stem growth and water use indicates that for a complete and accurate study of plantation characteristics, knowledge of the whole rotation period (or simultaneous measurements for different age classes) is necessary (Chan et al., 2018; Ma et al., 2019; Skubel et al., 2015). Therefore, to fully understand how plantations affect water resources in the long-term, information on water use during the entire management cycle, including the period before canopy closure, needs to be taken into account (Migliavacca et al., 2021).

2.3 Water and carbon fluxes in productive ecosystems

One of the main consequences of the changes in land use, such as replacing pastures with forests, is the increase in evapotranspiration from the landscape, as trees tend to transpire more water (Almeida et al., 2007; Stape et al., 2004). This can change the components of the catchment water balance and, at the same time, it alters the amount of carbon assimilated by the trees, which affects the productivity of the ecosystem. This productivity is defined as the amount of carbon assimilated against the amount of water used by the ecosystem. The impacts of afforestation or deforestation can be quantified by the ecosystem water and carbon fluxes, and the variables controlling them, such as precipitation, solar radiation, vapour pressure deficit (VPD), and air temperature (Pastorello et al., 2020).

Water, energy, and CO₂ fluxes can be measured simultaneously with the eddy covariance (EC) technique (Baldocchi et al., 2018; Baldocchi, 2003; Cleverly et al., 2020; de Araújo et al., 2010; Granier et al., 2008; Mendes et al., 2020), which can provide high temporal resolution measurements of energy, water, and CO₂ exchange at the ecosystem scale through the use of fast response instrumentation (Burba and Anderson, 2007). The technique is based on the turbulent exchange of energy and mass between the surface and the atmosphere (Baldocchi, 2008). Even though performed at a point, EC measurements can be considered to represent spatial averages for a larger area, referred to as "footprint". This area is a function of measurement height, wind speed and direction, surface roughness and atmosphere stability, which

CHAPTER 2. LITERATURE REVIEW

also means that the measurement height must be representative of the entire footprint surface (Burba and Anderson, 2007).

Many studies have measured CO_2 and water fluxes using the EC technique across different biomes around the globe. A global network, the FLUXNET project (http://fluxnet.ornl.gov/) is integrated by EC sites monitoring the land-atmospheric gas exchanges (Pastorello et al., 2020). EC sites in Australia are part of the OzFlux (http://www.ozflux.org.au/) within the Terrestrial Ecosystem Research Network (TERN) (Beringer et al., 2016). EC measurements have enabled detailed analysis of processes, relating plant physiological processes, water fluxes and atmospheric drivers. Additionally, the EC technique provides measurements of different components to generate a complete water balance (Baldocchi, 2008; Baldocchi et al., 2018).

At the ecosystem scale, EC measurements of latent heat flux (LE) can be used to derive evapotranspiration. Additionally, one of the main outputs from an EC system is the carbon exchange between the ecosystem and the atmosphere, which is denominated the Net Ecosystem Exchange (NEE). NEE is the net flux between the CO₂ from photosynthetic uptake (Gross Primary Production - GPP), and ecosystem respiration (ER), which is composed of the respiration by plants, from decomposition by soil microbial components, and root respiration (Anthoni et al., 2002; Granier et al., 2008; Luo and Zhou, 2006) (Figure 2.3). NEE is the measured variable, with GPP and ER being estimated by the application of mathematical models, which are usually based on light and temperature thresholds (Barr et al., 2013; Foken et al., 2012; Lasslop et al., 2010).



Figure 2.3: Schematic view of the NEE partition between GPP and ER.

The micrometeorological convention for fluxes is relative to the atmosphere, i.e. positive fluxes represent a gain to the atmosphere and negative fluxes a loss. The ecological convention is re-

lative to the ecosystem, i.e. positive (negative) fluxes represents ecosystem carbon gain (loss). The ecological convention is usually applied for annual budgets, with the terms Net Ecosystem Production (NEP), defined as NEP = -NEE, and Gross Ecosystem Production (GEP) defined as GEP = -GPP (Chapin et al., 2006, 2012).

NEE only includes carbon transferred from and to the atmosphere, thus not including other transfer pathways, such as aquatic exchange of dissolved organic and inorganic carbon. However, since other carbon transfers can be considered small in the majority of ecosystems and scenarios, NEE is a useful approximation of the total ecosystem carbon balance (Chapin et al., 2012; Luo and Zhou, 2006). Regarding plantation management, GPP can be a measurement of plantation productivity and wood production, since it gives the gross carbon uptake of the ecosystem, making it a strategic variable (Burba and Anderson, 2007). The magnitude of NEE, GPP, and ER responses can vary for different sites, species, and stand age (Baldocchi et al., 2021; Cleverly et al., 2020).

Analysing the variables measured with the EC technique over different sites and land uses can assist in the understanding of the key hydrological components of commercial plantations, and the impacts it may cause over a long term (Baldocchi et al., 2018). Year to year changes in structural and functional characteristics of an ecosystem may explain a significant portion of the evapotranspiration and carbon uptake observed. Therefore, EC measurements may be accompanied by measurements of LAI, diameter at breast height (DBH), and stand height, for example. Nevertheless, the EC measurements allow the development and application of models to quantify water use, forest productivity, and investigate tree traits (Almeida et al., 2007, 2010; Christina et al., 2016).

2.3.1 Drivers of water and carbon fluxes in plantations

The amount of water used and carbon assimilated by trees can provide insights about the plantation ecosystem dynamics (Baldocchi, 2008; Baldocchi et al., 2018; Pastorello et al., 2020). Water use and carbon assimilation are controlled by physiological variables such as the leaf area index (Irvine et al., 2004), plant functional type (Cabral et al., 2011; Campoe et al., 2020), and environmental variables, including solar radiation (van Gorsel et al., 2013), air temperature (Brümmer et al., 2012; Wardlaw, 2022), and water availability (Calder, 2007). Given the complexity and diversity of these drivers, measurements of carbon and water fluxes can vary considerably in space (site-specific to regional scales) and time (intra- and inter-annual periods) (Cleverly et al., 2020; Eamus et al., 2013).

In semiarid ecosystems, such as the majority of Australia, carbon and water fluxes are commonly correlated with precipitation (Baldocchi, 2008; Mendes et al., 2020). In conjunction, variables modulated by precipitation, such as temperature, VPD, and soil moisture, also impact the carbon and water fluxes (Jiang et al., 2020). As a result, plantations in semiarid regions can behave as a sink or source of carbon depending on precipitation, flipping from being a sink to a source, and vice-versa (Bracho et al., 2012; Gonçalves et al., 2017; Rodrigues et al., 2011). Additionally, the occurrence of disturbances (i.e., droughts, insect attacks, heat waves, and land use change) can shift the whole ecosystem to an overall carbon source (Cristiano et al., 2020; da Silva et al., 2015; Wardlaw, 2022). This is explained by the decreases in GPP, which might be accompanied by increases in ER from different inputs, such as litter, harvesting, and fertilization. This might surpass GPP and shift the NEP (Almagro et al., 2009; Brümmer et al., 2012; Reichstein et al., 2005a). Temperature can be also a main constraint on the NEP, since it is the main driver of the ER. Increases in temperature may decrease local soil moisture and increase evaporation, affecting both NEP and ET (Marcolla et al., 2011; Pereira et al., 1986, 2007; Pita et al., 2013). High temperatures are usually found during drier years, which are associated with more radiation, and increases in photosynthesis until high VPD and low soil moisture limit carbon uptake due to stomatal closure, which reduces photosynthesis (Pita et al., 2013; Renchon et al., 2018). In productive ecosystems, the amount of rain during spring, a period of high productivity for the plants, can be also an important factor controlling NEP (Chan et al., 2018; Ma et al., 2019; Ryan et al., 2010; Sun et al., 2010).

Despite a strong correlation, water availability not always leads to a carbon sink. Wetter periods followed by decreases in solar radiation may reduce photosynthesis and, as a results, the NEE can become positive (Eamus et al., 2013). For example, in a temperate *Eucalyptus* forest in NSW, NEE was mostly correlated with incoming solar radiation, with precipitation negatively correlated with NEE on a scale of 4-8 days due to enhanced cloud cover, which reduced radiation (van Gorsel et al., 2013). Water availability not only influences the NEP and ET, but can modify above and below ground carbon allocation, favoring roots or stem development (Dye, 2000). Mature trees usually present an already developed root system, resulting in more growth during dry seasons (Ryan et al., 2010; Stape et al., 2004). Contrarily, younger stands may allocate more carbon into developing their root system in order to get access to water in deeper soil layers and survive drier periods (Moroni et al., 2003).

Changes in stand structure and ecosystem traits can also drive the dynamics of water and carbon fluxes in plantations (Bracho et al., 2012; Chan et al., 2018; Nosetto et al., 2020). Variations in LAI, for example, affect the photosynthetic capacity of the plant, which in turn affects the carbon uptake and ET (Baptista et al., 2018; Battaglia et al., 1998; Binkley et al., 2010). Other functional traits, such as the size and shape of the leaves, alter the dynamics of light capture and, thus, affect the carbon assimilation (Drake et al., 2018; Whitehead and Beadle, 2004). Therefore, forest growth causes variability in the NEE and ET (Christina et al., 2016; Stape et al., 2004). Fast growing species, such as plantation trees, are associated with increasing LAI and height as the stand develops, promoting a fast increments in carbon assimilation (Forrester et al., 2010a,b). The water uptake by the expanding root system and greater leaf area may also increase water use with tree growth (Benyon and Doody, 2015; Benyon et al., 2006). For plantation trees, productivity is an important variable, given the importance of maximizing the amount of carbon sink against the amount of water used (Almeida et al., 2010).

2.3.2 Water use efficiency of plantations

Continuous measurements of NEE, and the subsequent partition of NEE to GPP and ER, allow estimations of the ecosystem water use efficiency (WUE). WUE is the ratio between the net production of carbon stocks and the amount of water consumed by the trees to produce these stocks. At the ecosystem level, WUE is defined as the ratio of gross ecosystem production (GEP) to evapotranspiration (ET) (WUE = GEP/ET) (Dye, 2000). The WUE represents an important ecophysiological variable reflecting the relationship between ecosystem water use and productivity (Baldocchi et al., 2021).

The WUE also assesses the ecosystem response to climate variability (Cabral et al., 2020). Seasonal and inter annual variability in environmental variables, including soil water supply, might impact the carbon balance by modifying the photosynthesis rate, which affects the stomatal conductance and the evapotranspiration (Jassal et al., 2009; Tfwala et al., 2019). For example, Tong et al. (2014) concluded that, for a mixed plantation in China, WUE decreased considerably during a vigorous growing season, due to strong solar radiation and higher VPD, which in this case increased evapotranspiration at a higher rate than GPP. In contrast, for a *Eucalyptus globulus* site in Brazil, Reis et al. (2019) discussed the highly effective stomatal regulation of blue gum trees, resulting in a maximum WUE even during the warm and drier season.

The stomatal conductance reflects how plants respond to changes in environmental variables, such as humidity, soil moisture, and light, thereby influencing the amount of water used and carbon assimilated by trees (Dye, 2000; Ma et al., 2018). Through stomatal responses, the WUE indicates plant water use strategies under different scenarios of water availability and distinct environments (Hubbard et al., 2010; Sun et al., 2010). For example, reductions in stomatal conductance can work as a defense mechanism to avoid water stress. As a result, water losses are avoided but at the same time CO_2 supply is limited. Therefore, plant growth, biomass allocation and evapotranspiration rates are strongly influenced by the WUE (Chapin et al., 2006, 2012).

The WUE is also modulated by species, plant functional type, stand age, and disturbances. For example, during the early years of a blue gum plantation studied by White et al. (2014), WUE decreased as the soil water deficit increased, with more carbon being allocated to roots, and less to wood. Contrarily, when studying mature stands, Griebel et al. (2020) registered an increase in WUE in response to water stress; this was explained by stomatal closure and deeper roots, as supported by other studies (Li et al., 2012; Pereira et al., 1986; Tong et al., 2014). However, for all these studies, WUE ultimately decreased during extreme drought stress.

Management practices, such as fertilization and irrigation, can influence WUE by increasing GPP due to above ground stem increment. For example, when studying the carbon allocations in a *Eucalyptus* plantation, Ryan et al. (2010) found that the annual fraction of GPP used for below ground (roots and stem) did not change in irrigated stands at the same time when the fraction to above ground carbon increased. In a different study, Ryan et al. (1996) identified that the combination of fertilizers and irrigation decreased the below ground allocation. Recent studies indicate that managing plantations for maximising their above ground growth can also result in a maximum WUE, given that increases in water use are offset by a greater increase in above ground carbon allocation (Stape et al., 2004; White et al., 2014). When comparing a native forest to commercial *Eucalyptus globulus* stands, White et al. (2021) found that while blue gum stands transpire more, they grow at a much faster rate when compared to native stands, thus having higher WUE. Similarly, Stape et al. (2004) investigated the WUE for irrigated stands and found a higher WUE during wet and drier years, when compared to control stands, with a 52% increase in wood production versus a 32% increase in water use.

Stand age and development stage are also factors that influence on the plantation WUE. Studies have found increasing WUE as stands grow and develop as a result of larger increases in GPP than in ET, as a consequence of an expanding LAI, sapwood area, and root system (Chan et al., 2018; Irvine et al., 2004; Jassal et al., 2009). When trees are very young and canopies are sparse, soil evaporation can be a major water flux and contribute significantly to ET. Additionally, reduced stand LAI may favour the growth of understory vegetation, since sparser canopies intercept less light and water than closed canopies. Understory vegetation might increase the ecosystem GPP, but it also has the ability to use water with low stomatal control, usually resulting in more water being transpired, generating more ET than increments in GPP (Forrester et al., 2010b; Irvine et al., 2002; Ntshidi et al., 2021; Rodrigues et al., 2011). After reaching canopy closure, WUE may decline following decreases in GPP, due to an overall decline in LAI and growth rates (Almeida et al., 2007). For Forrester et al. (2010a), WUE and growth peaked around 5 years for blue gum stands, and then declined with stand age.

The different responses in WUE when comparing climatic variability and differences in management highlight the need to evaluate WUE in order to optimize productivity while minimizing the water use, taking into account tree physiological dynamics and species characteristics. Optimum management practices need a good understanding of the effects of climate on the growth and WUE of plantations (Forrester et al., 2010b; Hubbard et al., 2004).

2.3.3 Water use and carbon assimilation in young plantations

Plantation trees experience changes in their structure and physiology during the entire rotation. Young trees have been found to display these changes in a more rapid way than mature trees and well established forests (Chan et al., 2018; Li et al., 2012). Therefore, as trees age and develop, there is also a change in the dynamics of water use and carbon assimilation (Chan et al., 2018; Clark et al., 2004; Coursolle et al., 2012).

A number of studies were dedicated to evaluate the water, carbon and energy balance of mature forests (Battaglia and Sands, 1997; Law et al., 2001; Nosetto et al., 2005) or forested stands that underwent productivity declines as they reached older ages (Almeida et al., 2007; Hubbard et al., 2010; Law et al., 2001; Ryan et al., 2004). However, the monitoring of water and carbon fluxes in young and newly established tree stands is recognized to be sparse in the literature, with a limited number of studies monitoring fluxes during the first years after plantation establishment (Anthoni et al., 2002; Bergeron et al., 2008; Chan et al., 2018; He et al., 2012; White et al., 2021). This means that the dynamics of carbon assimilation and water uptake in young stands are not well-known. Table 2.1 summarizes the findings of several studies on carbon assimilation and water use that are available in the literature.

A common aspect highlighted by some of the studies listed in the Table 2.1 was the predominance of the sensible heat flux over the latent heat flux during the monitoring period (Chan et al., 2018; Ma et al., 2018). This was explained by the sparse canopy of the sites, since the plantations had not reached canopy closure yet. Latent heat flux dominated during the growing season, when enough soil moisture was present. Alternatively, other studies did not register much variability in the partition of the sensible and latent heat flux over time, but registered changes in the understory vegetation of the plantation (Forrester et al., 2010a; Irvine et al., 2002; Ntshidi et al., 2021). The reduced LAI of younger trees facilitates the growth of understory vegetation, which usually have substantial fluxes of carbon and water. As trees develop and the LAI increases, the understory contribution reduces considerably and the trees become the main contributor to the ecosystem fluxes. For a pine plantation in early developmental stage, Bracho et al. (2012) found that carbon assimilation was majorly driven by increases in LAI.

Younger trees are usually a source of carbon during the first few years after establishment. The age of transition from a carbon source to a sink is quite variable even among stands from the same species due to variable climate and site-specific aspects (i.e., soil type, disturbance history, and land use) (Bjarnadottir et al., 2009; Chan et al., 2018; Coursolle et al., 2012; Peichl et al., 2010; Sun et al., 2008). During the first few years after plantation establishment the eco-system usually releases more carbon through ER than what is assimilated by photosynthesis (GPP) (Anthoni et al., 2002; Bracho et al., 2012). This was observed in the site described in Granier et al. (2008), who concluded that more than half of the assimilated carbon in a plantation in France was lost by the respiratory processes. Similarly, for a site located in Iceland, Bjarnadottir et al. (2009) concluded that the annual variations in NEE were apparently more linked to differences in ER due to decomposition of organic matter.

The prevalence of ER over GPP soon after plantation establishment was highlighted in Chan et al. (2018), who measured NEE for a pine plantation forest at the beginning of the rotation. In

CHAPTER 2. LITERATURE REVIEW

their study, the site was a source of carbon during the first 4 years. High temperatures during the dryer periods drove increases in ER, which was larger than carbon assimilation, turning the site to an overall source. Contrastingly, the sites of Bergeron et al. (2008) and Coursolle et al. (2012) registered lower values of NEE since plantation establishment, due to a low residual soil carbon pool, which resulted in considerably lower ER. The low residual soil carbon enabled the sites to become sinks in a shorter time interval, highlighting how site-specific aspects can change the dynamics of carbon fluxes. For example, for the same species and with both sites located in Canada, the site described in Coursolle et al. (2012) became a sink after only 2 years, versus 4 years for Chan et al. (2018).

The dynamics of NEE can change dramatically over the years, due to the influence of tree age, structure and the effect of climate on growth (Cabral et al., 2011; Campoe et al., 2020). In addition to management practices, the climate of the region strongly influences the potential carbon assimilated during the first years of development. Since young trees have a shallower root system, they depend on soil moisture from the shallow layers of soil, which is mainly driven by precipitation (Zha et al., 2009). As a result, young stands are more vulnerable to water stress (Anthoni et al., 2002). In particular, Zha et al. (2009) identified larger variability for variables monitored for the younger stands, meaning that young trees can display more dramatic responses due to changes in temperature and precipitation, when compared to more mature trees.

The reviewed studies emphasized the high spatial and temporal variability of the carbon fluxes and water use among different species. In particular, the studies highlighted the need for contribution in the monitoring of carbon and water fluxes over diurnal, seasonal and annual scales for the entire cycle, since only a few studies have examined how age and growth influence the ecosystem-scale energy balance, transpiration rates, and net carbon uptake.

Table 2.1: List of literature on young and developing plantations and a summary of their findings.				
Reference	Species	Location	Age	Findings
Anthoni et al. (2002); Law et al. (2001)	Pinus ponderosa	Central Oregon, USA	15 years old	The ratio ER/GEP varied seasonally, and ranged from 0.4 to 1.45
Bjarnadottir et al. (2009)	Siberian Iarch (<i>Larix sibirica</i> <i>Ledeb</i>)	Eastern Iceland	12 - 14 years	Sink after 12 years. Respiration and decomposition were the main driver for NEE variability
Bracho et al. (2012)	Slash pine (<i>Pinus</i> <i>elliottii</i>)	Gainesville, Florida, USA	1 - 9 years	Sink after 4 years. Drought had a much stronger impact on GEP than on ER, resulting in a clear reduction in NEP
Cabral et al. (2011)	E. grandis x E. urophylla	Sao Paulo, Brazil	2 - 4 years	Source or sink behaviour (NEP) was mainly influenced by changes in GEP due to rainfall variability
Chan et al. (2018)	White pine (<i>Pinus</i> strobus L.)	Southern Ontario, Canada	1-14 years	Sink after 5 years, subsequent source years related to dry years and droughts
Coursolle et al. (2012)	White pine (<i>Pinus strobus L.</i>)	Ontario, Canada	Young 1- 15 years	Source up to 17 years. LAI was the main factor affecting GEP, ET, and NEP of young stands
Forrester et al. (2010a)	Eucalyptus globulus	Ballarat, Victoria, Australia	2, 4, 5, 6, 7 and 8 years	Tranpiration and LAI declined for older stands (5-7 years), consequently, WUE also declined
Granier et al. (2008)	European beech (<i>Fagus sylvatica</i> <i>L</i> .)	North-eastern France	29 - 40 (+/- 5) years	NEP correlated with tree growth. Drought and management explained most of the variability in fluxes and growth

17

Reference	Species	Location	Age	Findings
Harper et al. (2014)	Eucalyptus globulus, Eucalyptus occidentalis, and Pinus radiata	Corrigin, Western Australia	1 - 6 years	Marked differences in tree response to slope across the site. Increasingly WUE as tree aged
Irvine et al. (2002)	Pinus ponderosa	Oregon, USA.	14 years	CO ₂ fluxes in young stands correlated with LAI and root development
Krishnan et al. (2009)	Douglas-fir	East coast of Vancouver Island, BC, Canada	3-7 years	Stand-age effects on carbon fluxes were much higher than that of interannual variability
Ntshidi et al. (2021)	Orchard	Koue Bokkeveld, South Africa	5- 7 years old	Dense understory contributed to almost 40% of ET of young orchards
Ma et al. (2018)	Pinus tabuliformis	Nothern China (Beijing, China)	*	Seasonality in ET was controlled by biological factors and water availability
Marcolla et al. (2011)	Festuca rubra (L.)	Viote del Monte Bondone, Italian Alpes	cut annually	In the warmest and driest years the ecosystem was a significant source of CO ₂
Nosetto et al. (2005)	Eucalyptus grandis	Western coast of the mid-Uruguay river in Argentina	1 - 9 years	ET changes with age showed that tree water use rose at a steady state until 4–5 years of age, having a slow increase afterwards
Reis et al. (2019)	Eucalyptus grandis	Mato Grosso do Sul State, Brazil	1-2 years	Young eucalypt exhibited higher photosynthetic capacity, particularly during the dry season, which is likely associated with growth maintenance

Table 2.1: List of literature on young and developing plantations and a summary of their findings.

Reference	Species	Location	Age	Findings
			before (8 - 12 years old	The young eucalypt coppice behave as a
Rodrigues	Eucalyptus	Espirra Estate,	trees) and after a	carbon
et al. (2011)	globulus	Portugal	felling (1 - 3 years old	source in the first seven months of the new
			trees)	rotation. Increasinly WUE with tree growth
Skubel et al. (2015)	Pinus strobus L.	Ontario, Canada	11 years	ET was most conservative in the youngest
				forest, which led to an increase in WUE when
				compared to mature stands
Sun et al. (2010)	Loblolly pine	North Carolina, USA	3-5 years	ET/P was 0.66 for 4 years old stands
				compared to 0.88 for 16 years old stands
Thornton et al. (2002)	Pinus ponderosa	California, EUA	1-10 years	Ecosystem behaved as a carbon source
				followed by strong and gradually diminishing
				sink after harvesting
Vertessy et al. (2001)	Eucalyptus regnans	Melbourne, Australia	15 years	Total annual ET declines from 1371 mm at
				age 15 years to 911 mm at age 240 years,
				following decreases in LAI
Vickers et al. (2012)	Pinus ponderosa	Central Oregon, USA	16 years	Annual NEE 3 times smaller when comparing
				young to mature stands. This is correlated
				with the larger LAI for mature stands
Zha et al. (2009)	Jack pine (<i>Pinus</i> <i>banksiana</i>)	Saskatchewan, Canada	2-90 years	Following clear cutting, site site behaved as
				slight sink at year 10 and strong sink at year
				30. NEE followed increases in LAI

Table 2.1: List of literature on young and developing plantations and a summary of their findings.

2.4 Modelling plantation water use

Modelling plantation water use is important to guide policy and management decisions, also allowing the exploration of different afforestation scenarios (Almeida et al., 2007; Bonan et al., 2014). Models that can accurately simulate the complexity of plantation trees, with adequate data requirements, are recommended for this task (Greenwood, 2013).

A large number of studies have been conducted to model the impact of land use, including afforestation, on the water resources and have shown such changes to be important (Adelana et al., 2014; Almeida et al., 2010; Azarnivand et al., 2020; Dresel et al., 2018; Miehle et al., 2006). There are different approaches to model vegetation water use and its impact on water resources. This section will discuss a variety of approaches, from simpler models, such as the Zhang curves, to process-based hydrodynamic models. Table 2.2 presents some key studies on the different model approaches discussed in this review.

2.4.1 Zhang curves

One of the first efforts in assessing the hydrological role of vegetation resulted in the so-called Zhang curves (Zhang et al., 1999, 2001), which were derived from data obtained in catchment water balance studies and paired-catchment studies. The Zhang curves are based on a relationship proposed to predict the effect of vegetation changes on evapotranspiration. This relationship yields a dimensionless function and it can be used to calculate the mean annual actual evapotranspiration when annual rainfall and potential evapotranspiration are known. The Zhang curves were the basis for estimating the impact of forest and land-use change on streamflow in a number of studies, including applications in the Australian Government, policy organisations, and research publications (Bren et al., 2006; Brown et al., 2006; Keenan et al., 2004; Zhang et al., 2001, 2003, 2007).

The Zhang curves were developed using data from catchments where slopes were gentle and the soil was relatively thick. Detailed information regarding vegetation types was not available, so vegetation was classified as herbaceous plant, herbaceous plants and trees, and forest. The catchment size was variable, including areas from 1 km² to 600,000 km² (Zhang et al., 1999, 2001).

The Zhang curves have the advantage of being a practical tool, usually requiring data already available and only using two parameters, which are fixed. They have potential uses in catchment-scale studies of land-use change, such as afforestation and deforestation. However, one of the main limitation of this method is that it is only reliable for catchments with average annual precipitation greater than 1000 mm. In a technical report, Greenwood (2007) describes the discrepancies between measured South Australian data and theoretical data from the Zhang

curves (Zhang et al., 1999, 2003). The theoretical Zhang curves did not show strong agreement with measured data from any Australian catchment with mean annual rainfall less than approximately 1000 mm/year, which includes the majority of Australia. Under lower annual rainfall, it underestimates evapotranspiration in forest, mixed vegetation, and pasture, resulting in overestimated predictions of runoff (Dresel et al., 2018). Additionally, since the method is based on mean annual data, it should not be used to explore inter-annual variability, as stated in the original work of Zhang et al. (1999). For values of the parameters, the Zhang curves also do not satisfy the water balance, as noted by Daly et al. (2019). Therefore, the application of the Zhang curves can underestimate the impacts of afforestation and its implications for water resource management. As consequence, resources can be over-allocated, with effects for all downstream users and the environment (Greenwood, 2007).

2.4.2 Water use models in forest growth models

Process-based forest growth models have become widely used as a means of predicting the growth of plantations and native forests in Australia and South Africa (Dye, 2000; Dye et al., 2004). These models estimate forest physiological and environmental factors to calculate their influence on stand growth. Usually, these models estimate the GPP from photosynthetic active radiation and use simple empirical relationships to predict forest growth and productivity. The empirical relationships are derived from measurements and experimental data collected on forests and plantations during a considerably long period. Process-based models vary in complexity and scale of application (Battaglia and Sands, 1997; Miehle et al., 2006; Sands and Landsberg, 2002), but are usually applied at seasonal and annual resolutions (Feikema et al., 2010).

The 3PG (Physiological Principles in Predicting Growth) forest growth model was developed by Landsberg and Waring (1997) and requires relatively few input data. It has been variously described as process-based, mechanistic, hybrid, and semi-empirical (Sands and Landsberg, 2002). Much of the focus of the development of 3PG, and indeed other forest growth models in the past, has been on their use for predicting the growth of commercial plantations (Almeida and Sands, 2016; Almeida et al., 2007), and relatively little emphasis has been given to validating or improving predictions of plantation water use. The 3PG model has been parameterized for different plantation species, including *Eucalyptus globulus* (Sands and Landsberg, 2002) and *Eucalyptus grandis* (Almeida et al., 2004), but it has shown poor results before canopy closure and during early canopy development (Dye, 2001; Dye et al., 2004).

An updated version of the model, 3PG+ was designed to be coupled to the Catchment Analysis Tool (CAT) (Beverly, 2007; Beverly et al., 2005). CAT is a modelling framework based on the one-dimensional crop model PERFECT (Productivity, Erosion and Runoff Functions to Evaluate Conservation Techniques) (Littleboy et al., 1989). CAT includes a water balance model, which

CHAPTER 2. LITERATURE REVIEW

allowed 3PG+ to estimate a daily multi-layered soil water balance, runoff from rainfall, and soil evaporation. The coupling of 3PG+ to CAT enabled the simulation of plantation water use at a daily time step and, at the same time, the simulation of dynamic forest growth. When applying 3PG+ in CAT, Feikema et al. (2010) concluded that the model can satisfactorily simulate stand transpiration at 2-8 weeks timescale.

Another extensively used model is the Cabala model (Battaglia et al., 2004), developed by CSIRO. Cabala simulates light, nitrogen and water capture by trees and, through the use of these resources, derives estimations of tree growth (foliage, roots, and stem). Similarly to the 3-PG model, the soil is represented as a single bucket, whose size is determined by plant available water and root zone depth. Water enters the soil through rainfall and is removed by soil evaporation and transpiration (Benyon et al., 2009). The Cabala model was described and parameterized to simulate *Eucalyptus globulus* trees in Battaglia et al. (2004). In this study, the Cabala model performed reasonably well when simulating growth of younger stands.

Benyon et al. (2009) applied and compared 3PG+ and Cabala to simulate a *Eucalyptus globulus* plantation in the Green Triangle, a region located in Victoria, Australia. Both models estimated vegetation water use based on inputs of rainfall, estimated ET, simulated LAI and root depth, and soil water capacity. Simulations were in one dimension and on a daily or monthly time-step. The 3PG+ model performed better than the Cabala when simulating annual ET in plantations without access to groundwater, but both models performed poorly when groundwater was accessible to the trees. The Cabala model performed better when it came to monthly ET. In 3PG+ and Cabala, LAI is a simulated variable rather than an input variables, and it is well known that it has a strong influence on the ET. For both models, simulations of LAI constantly returned overestimated values, which affected predictions of ET. Additionally, root depth, also a simulated variable, may have affected ET values.

Process-based models enabled progress in the simulation of water use and growth by plantations. However, these models lack a more detailed representation of the transpiration process and constraints (Almeida et al., 2010; Feikema et al., 2010; Miehle et al., 2006), as was concluded by Benyon et al. (2009). The results from Benyon et al. (2009) highlight that forest growth models do not include the representation of important processes related to the movement of water within trees. For example, forest growth models do not include physiological constraints to root water uptake, which can explain how transpiration is reduced under dry conditions leading to cavitation or is regulated under well-watered conditions, such as when roots reach groundwater.

2.4.3 Tree Hydrodynamic modeling

While the Zhang curves and the forest growth models represent simpler approaches to investigate vegetation water use, they do not provide knowledge of intra-daily dynamics of stand

CHAPTER 2. LITERATURE REVIEW

transpiration. A detailed representation of the transpiration process over intra-daily resolutions can identify and characterize different vegetation traits and adaptive mechanisms (Matheny et al., 2017). These mechanisms can provide, for example, important information on how trees cope with water deficit and recovery after droughts; this can in turn be critical to outline more sustainable management practices to improve tree productivity and resilience (Fatichi et al., 2016; Mencuccini et al., 2019; Verma et al., 2014).

The complex relationship between vegetation water use and environmental drivers, as discussed in Section 2.3.1, cannot be captured by simple models since it involves intricate biological and physical processes (Mencuccini et al., 2019). Transpiration is a complex process, considering that plants regulate how much water they transpire through controls at the leaf, stem, and root levels (Matheny et al., 2017). Models that can simulate the water transport below and above ground can provide important information on tree water use strategies and incorporate mechanistic representation of the internal water storage, water flow through the vegetation, xylem hydraulic properties, or stem and canopy structure (Bohrer et al., 2005; Mirfenderesgi et al., 2016).

The application of plant hydraulic models has been growing in recognition over the years for linking root water uptake, transpiration rates and carbon assimilation. Likewise, the modeling of processes such as stomatal conductance and conductivity can help to further understand the partitioning of energy fluxes between latent and sensible heat flux (Kumagai, 2001; Mirfenderesgi et al., 2018; Vrugt et al., 2001). Several quantitative approaches and applications have been developed to describe water flow in plants from the tissue to the biosphere scale. Some studies have focused on reviewing the different models and equations describing the water flow in plants, and they include an historical description (Mencuccini et al., 2019), the evolution and comparison of approaches at different spatial and temporal scales (Manzoni et al., 2013b), and a discussion regarding the relevant processes and interaction between water and plants (Fatichi et al., 2016; Matheny et al., 2017), including quantification of vegetation traits.

Most models that explicitly resolve the movement of water within the plant system rely on the cohesion-tension theory, which explains how water can be transferred upward from the soil to the atmosphere across a tree height of several meters, in the absence of osmotic pressure differences (Couvreur et al., 2018). An uninterrupted water column can extend from the roots to the leaves under tension and, as the stomata open, water is transferred to the atmosphere pulling water from the soil, through the roots and stem (Steudle, 2001). Accordingly, the system composed by the Soil, Plant, and Atmosphere is interpreted as a Continuum (SPAC) with water flowing through its different compartments following a path of decreasing water potentials (Nobel, 2009).

In this context, the first models proposed to describe transpiration fluxes used an electrical analogy, with water flowing from one compartment to the other following water potential gradients associated across plant conductive tissue with resistances to the flow (Cowan, 1965;
de Jong van Lier et al., 2008; Jones, 2009; Sperry et al., 2003; van den Honert, 1948). Recent advances in these models account for the water storage within the plant using capacitors, and link the water and CO₂ fluxes through the stomatal conductance (Bartlett et al., 2014; Cruiziat et al., 2002; Daly et al., 2004a,b; Hartzell et al., 2018; Manoli et al., 2014; Manzoni et al., 2013b). Electric-circuit models commonly assume that the water flows along the SPAC occurs as a succession of steady states, whereby the water potential in the different compartments of the system adjusts instantaneously to environmental changes. Many electric-circuit models also treat the soil as a finite capacity and often consider a single compartment for each plant component (e.g., root and stem xylem) (Daly et al., 2004a; Hartzell et al., 2017). A finer resolution of resistances and capacitances might be used if a more detailed representation is desirable, but adding more layers may yield ordinary differential equations that are more difficult to solve (Chuang et al., 2006; Fatichi et al., 2016). A few electric-circuit models include formulations that account for root water compensation and other traits, although such inclusion requires the introduction of empirical parameters in the root water uptake formulation (Couvreur et al., 2012; Kennedy et al., 2019; Meunier et al., 2018).

A continuous representation of the SPAC can be achieved in models that describe the water flow in the soil and plant xylem as flow in porous media (Fruh and Kurth, 1999). Porous-media models combine the continuity equation with Darcy's law to define partial differential equations for the unsteady dynamics of the water potential across the SPAC and account for the transient response of water potential along the tree system. Some applications of these models focus on the water fluxes within the above-ground stem (Bohrer et al., 2005; Chuang et al., 2006; Kumagai, 2001), others are centred on the simulation of below-ground fluxes and the interaction between soil and roots (Amenu and Kumar, 2007; Mendel et al., 2002; Somma et al., 1998; Teodosio et al., 2017), with more recent applications looking at the whole SPAC system (Huang et al., 2017; Janott et al., 2011; Mirfenderesgi et al., 2016; Quijano and Kumar, 2015; Verma et al., 2014). Porous-media models are able to simulate a variety of processes, such as root water compensation and hydraulic redistribution (Verma et al., 2014), which are embedded in the root water uptake formulation. A canopy representation can also be accomplished by accounting for a leaf area distribution and light distribution functions throughout the stem (Christoffersen et al., 2016), and dynamic formulations for the stem capacitance and conductances can be considered (Mirfenderesgi et al., 2018).

Porous-media models that simulate the entire tree structure, with a detailed 3D representation of branches and root systems, are computationally demanding and require specific and complex parameterizations. As a result, application of these models is impracticable to simulate water flow in more than a single tree (Bohrer et al., 2005; Janott et al., 2011). One-dimensional models that lump within-tree spatial hydraulic variability in their parameters are a more practical option to represent water movement in individual trees and within stands (Amenu and Kumar, 2007; Mirfenderesgi et al., 2016, 2018; Quijano and Kumar, 2015).

Another axis of complexity that differentiates transpiration models is the level of vertical detail of the canopy representation. Single-leaf models represent the simplest approach and resolve evaporative demand from the canopy as a single surface. More advanced approaches represent the canopy as two layers, of light and shade leaves, or as multiple layers, each of a different type/size cohort of trees within the canopy (e.g., Medvigy et al., 2009). Advances in the canopy representation include the development of vertically detailed canopy representations (e.g., Bonan et al., 2018; Chen et al., 2016; Drewry et al., 2010), which led to a strong call to advance global land surface models by including a multi-layered canopy representation (Bonan et al., 2021). The complexity of the vertical representation of the canopy for the purpose of light attenuation and atmospheric demand for water could be decoupled from the complexity of the vertical representation of the hydraulic conductive pathway. For example, some models represent a vertically detailed canopy but represent the hydraulic pathway at its most simplistic form as a set of three (soil, xylem, and leaf) reservoirs (Trugman et al., 2016; Xu et al., 2016). While this approach is numerically more efficient, it may lose some of the stomatal control dynamics that are due to different rates of water storage losses at different elevations through the canopy. Specifically, it was shown that the higher leaves would experience water limitations due to storage loss sooner in the day than the lower leaves (Bohrer et al., 2005). Conversely, the continuous vertically detailed system of partial differential equations solved by porous-media models makes them a better choice to simulate plant hydraulic behavior, species-specific hydraulic traits, and their interactions with environmental drivers across different species and ecosystem types (Matheny et al., 2017), providing a more detailed representation of the tree domain and canopy structure effects than electric-circuit models or single-layer porous-media models.

Reference	Application					
Empirical models - Zhang curves						
Zhang et al.	These studies presented the Zhang curves, a model and tool to be used for assessing the effect of vegetation					
(1999, 2001)	changes on catchment average water balance					
Bren et al.	Derived estimates of change in water yield from grassland sites converted to Radiata pine by using the model of					
(2006)	Zhang et al. (2001) to estimate the difference in water use between native forest and grassland					
Brown et al.	Zhang curves were used to estimate the impact of land use change from pasture to plantation in Tasmanian					
(2006)	catchments					
Forest Growth Model - 3-PG						
Landsberg and	The 3-PG model is presented as a model to simulate stand growth and biomass increment, using a monthly time					
Waring (1997)	step					
	3-PG was applied to predict growth and water use of Pinus radiata trees. The study highlighted the need and					
Dye (2001)	difficulties experienced in finding useful data sets that include descriptions of forests					
Sands and	3-PG model was parameterized for <i>E. globulus</i> trees. 3-PG provided a good simulation of future growth of blue					
Landsberg	gums stands when the model was initialised with observed biomass data at some age around or following canopy					
(2002)	closure					
Almeida et al.	3-PG model was parameterized for <i>E. grandis</i> trees. When quality data was available, 3-PG could simulate different					
(2004)	Eucalyptus clones					
Dye et al.	3-PG was applied to simulate stand growth over 12 months of stands of <i>E. grandis</i> x <i>E. camaldulensis</i> hybrid					
(2004)	clones, representing early, mid and late rotation age					
Feikema et al. (2010)	This study introduced 3-PG+, which integrated 3-PG to CAT. The integration of 3PG+ into CAT resulted in an					
	improvement in accuracy and applicability, and provided for the spatial application of 3PG+ across diverse and					
	mixed land use catchments					
	Forest Growth Model - Cabala					

 Table 2.2: Key studies and application of the different modelling approaches for forest growth and water use

Reference	Application					
Battaglia et al. (2004)	This study introduced the Cabala model. Cabala performed well in simulating the growth of young stands					
Benyon et al. (2009)	This study applied and compared Cabala and 3PG+. The 3PG+ model performed better than the Cabala when simulating annual ET in plantations without access to groundwater, but both models performed poorly when ground-water was accessible to the trees. The Cabala model performed better when it came to monthly ET					
Hydrodynamic models - Electric circuit						
Daly et al. (2004a,b)	Applied a SPAC model of soil water balance, transpiration, and photosynthesis, comparing both Leuning (1990) and Jarvis (1989) approaches for stomata behavior					
Bartlett et al. (2014)	A CAM photosynthesis model was used to simulate the leaf carbon uptake and coupled to a simple SPAC approach to simulate soil water balance and different photosynthetic pathways (C3, C4, CAM)					
Hartzell et al. (2017, 2018)	Model similar to Daly et al. (2004a,b), but includes a scheme for plant water storage					
	Hydrodynamic models - Porous media					
Kumagai (2001)	Simulated 1-D water flow through a tree stem using an experimental dataset to validate the model and parameters					
Mendel et al. (2002)	Modeled root water uptake and 2-D water transport through the soil and root system as coupled processes.					
Bohrer et al. (2005)	Introduced FETCH, a model that simulates water flow through a realist 1-D canopy structure, including branches.					
Chuang et al. (2006)	Proposed to solve the PDE of porous media models as ODE, which could be used to derive sap flow observations into transpiration estimations					
Amenu and Kumar (2007)	Modeled the root system to investigate hydraulic redistribution and root water compensation					

 Table 2.2: Key studies and application of the different modelling approaches for forest growth and water use

Reference	Application					
Janott et al.	Combined 1D approaches to simulate water flow in the aboveground plant based on the porous media equation					
(2011)	with root water uptake and soil water flow models also based on the porous media equation					
Manoli et al.	The model coupled Richard's equation for soil moisture and plant transpiration. A 3D root water uptake model was					
(2014)	applied to investigate overlapping root systems					
Verma et al.	Simulated 1-D water flow, coupling below and above ground dynamics; solved the system of equations through a					
(2014)	software					
Mirfenderesgi	Introduced EETCH2, successor of EETCH. The model simulated water flow in a 1 Divertical tree domain, including					
et al. (2016,						
2018)	son, roots, and stern xylern					
Quijano and	Coupled below ground water transport to an eachydrological model (MI Can) to simulate above ground dynamics					
Kumar (2015)	Coupled below ground water transport to an econydrological model (MECan) to simulate above ground dynamics					

2.5 Scope of research

This project aims to understand the water use and carbon trade-off in commercial plantations, specifically focusing on quantifying and understanding the dynamics of plantation water use and carbon assimilation during the first years after plantation establishment. This work involves the collection of experimental data and the development of a tree hydrodynamic model. Figure 2.4 presents an overview of the research objectives and how they relate to each other. The research objectives are:

1: *Quantify the transpiration rates in a young* E. globulus *plantation during the first few years after its establishment.*

Measurements of actual transpiration from growing plantations over the first few years after being planted will improve the knowledge of the water cycle and provide reliable data for a more precise water balance. Water accounting models for plantation regulation, which are currently applied in South Australia and South Africa, are based only on data from mature plantations; therefore, they do not account for the dynamics of a growing plantation.

Specifically, it is necessary to understand how much water young *E. globulus* plantations use. Additionally, this data provide useful high-resolution data of tree response to different environmental conditions (e.g., heatwaves, dry conditions, and wet conditions) in different years.

2: Analyse and quantify the relationship between water use and growth rates (i.e., Water Use Efficiency – WUE) in a growing plantation before reaching maturity.

Afforestation with *Eucalyptus globulus* is a strong economic activity in Australia. Quantifying the variation in the amount of carbon that the trees are assimilating against the water they are using can help authorities to determine more efficient license procedures based on experimental data from growing plantations.

The WUE of a young plantation can provide information on the dynamics of carbon assimilation and water use with plantation age. The strong seasonal variability in both transpiration and carbon uptake increases the need for a more detailed and precise framework for water allocation for plantations. Taking into account the benefits of a sustainable management of *E. globulus* plantations, this aim is relevant to show the possibility for plantations to support increases in land productivity without harming the water resources available.

3: Develop a tree hydrodynamic model to support plantation management.

The amount of water transpired by trees can have fluctuations according to the amount of water taken up by roots, which can alter the amount of carbon fixed by trees. In this context,

the modelling of plant hydraulics can support the understanding of the water fluxes within the soil-plant-atmosphere continuum, and generate reliable projections of tree transpiration for different scenarios. In the model, this can be performed by accounting for the movement of water through the xylem, and stem and canopy structure.

This objective entails the development of the tree hydrodynamic model FETCH3, a new version of FETCH2 (Mirfenderesgi et al., 2016, 2018). FETCH3 simulates tree transpiration, coupling the trees to the soil to achieve a fully coupled soil-atmosphere-continuous model. A software package in Python will be developed and made available open-access and open source.



Figure 2.4: Overview of the research activities.

Chapter 3

Water and carbon fluxes in a young *Eucalyptus globulus* plantation

This chapter contains large parts of the article "Trading a little water for substantial carbon gains during the first years of a *Eucalyptus globulus* plantation" published in Agricultural and Forest Meteorology.

3.1 Aim

Determining water and carbon fluxes at the plantation scale during the first years of tree development after establishment is important to quantify plantation productivity (Baldocchi et al., 2021; Bergeron et al., 2008). The water use efficiency (WUE), which indicates how much water the trees use to grow, can provide important information for managing commercial plantations, possibly assisting land owners and government planners in allocating water for plantation activity in a sustainable manner (Pereira et al., 1986; White et al., 2016).

The aim of this study is to quantify the water and carbon fluxes in a *Eucalyptus globulus* plantation during its initial 4 years after establishment. The main objectives are (i) to quantify changes in carbon assimilation and evapotranspiration rates with stand development, and (ii) to determine the water use efficiency of a commercial plantation during the early stage of tree development. The outcome of this study provides an estimation of the trade offs between carbon and water in commercial plantations during the first years after planting.

3.2 Materials and Methods

3.2.1 Study site

The study area is located in a *E. globulus* plantation near Digby, in Southwest Victoria, Australia, within the Glenelg River Basin (Figure 3.1). The plantation was established in August 2017, with a mean tree density of approximately 1000 trees-ha⁻¹. The depth to the water table is between 5 to 10 m (Dresel et al., 2018), with trees being likely too young to reach groundwater resources (Benyon et al., 2006). Tree height at planting was between 150 mm and 300 mm, and trees were on average 11 meters tall at the end of the monitoring period (July 2021). Fertilization and weed control were applied on planting mounds before the planting, after establishment, and after 1 year to help promote growth, as it is common practice for a commercial rainfed plantation in Australia (Benyon et al., 2006).



Figure 3.1: Location of the study site, and photos of the trees and the eddy covariance tower in a) December 2017, when the system was installed, and b) July 2021, at the end of the monitoring period.

The site has a long-term average annual precipitation (1950 - 2020) of 711 mm (Bureau of Meteorology station - Merino #90057, located 18 km from the site). Precipitation is dominantly in the winter (Jun - Aug) and annual precipitation in the region is highly variable. The estimated average annual potential evapotranspiration is 1020 mm (2017 - 2019) and the Koppen-Geiger climate classification is Cfb (temperate no dry season warm summer) (Dresel et al., 2018). Environmental data collected during this study (section 2.2) showed strong seasonal variation, as expected of a temperate latitude site (Figure 3.2). Daily incident solar radiation was on average approximately threefold higher in summer (34 MJ m⁻² d⁻¹) than winter (7 MJ m⁻² d⁻¹). The mean daily air temperature was 18.5 °C during the hottest month (January) and 8.9 °C

CHAPTER 3. WATER AND CARBON FLUXES IN A YOUNG *EUCALYPTUS GLOBULUS* PLANTATION

during the coldest month (July). The combination of higher rainfall and lower temperatures, hence a lower vapour pressure deficit (VPD), resulted in a generally high soil water content during this period (exceeding 40%), with low values (below 5%) observed during summer. Environmental conditions were relatively consistent across the years of the study, with 2018 being the warmest year (average of 13.42 °C - mostly due to high temperatures in the early part of the year), while daily averaged temperatures were consistent in 2019 and 2020 (12.58 °C and 12.55 °C, respectively). The annual precipitation was 749, 725 and 778 mm for 2018, 2019, and 2020, respectively (all above the long term average); however, seasonal dynamics varied, with precipitation being largest during winter in 2018 and 2019, and during the spring in 2020. Precipitation was also lower during the summer of 2018 relative to subsequent years, resulting in lower soil moisture, which reached its minimum value in February (0.02 m³m⁻³).



Figure 3.2: Total daily solar radiation ($MJ m^{-2} d^{-1}$) and rainfall (P, mm d⁻¹), and daily averages of the soil water content (SWC, %), air temperature (T_{air} , °C), and Vapour Pressure Deficit (VPD, kPa). Vertical dashed lines represent the transition between years.

3.2.2 Data description

Energy, water, and CO₂ fluxes were continuously measured above the tree canopy by an eddy covariance (EC) system. The EC system was installed on an extendable tower, that was raised to different heights as the trees grew. The tower height was initially 5 meters (December 2017)

CHAPTER 3. WATER AND CARBON FLUXES IN A YOUNG *EUCALYPTUS GLOBULUS* PLANTATION

and was raised to its maximum height of 14.5 meters in January 2021. Table 3.1 reports the dates when the tower was raised with the corresponding canopy heights. The sensors were positioned to maintain a distance of at least 3 m above the canopy top (Burba and Anderson, 2007). The extension of the homogeneous plantation cover over the flat surface is a guarantee for good fetch (Rodrigues et al., 2011). Increasing the tower height as the canopy developed avoids the large fetch uncertainties that would have been created if the tower had a fixed position.

Turbulent flux measurements were provided by an integrated sonic anemometer/open path infra-red gas analyser (IRGA - model IRGASON, Campbell Scientific Instruments, USA). Raw data were collected at 10 Hz frequency and accumulated to a half-hourly temporal resolution. Radiant flux measurements were provided by a four-component (incoming and outgoing short- and long-wave) radiometer (CNR4 - Kipp and Zonen, The Netherlands). The soil heat flux (G) at the soil surface was calculated using heat flux plates (model HFP01-L - Hukseflux, The Netherlands) installed at 8 cm from the soil surface and corrected using soil temperature measurements at 4 cm from the surface. The soil water content (SWC, sensors model CS650 - Campbell Scientific Intruments, USA) at 30 cm depth was measured simultaneously with the fluxes (Figure 3.2).

Additional measurements included air temperature (T_{air}) and relative humidity (RH) (HC2S3, Campbell Scientific Instruments, USA). Precipitation (P) was measured using a tipping-bucket rain gauge (CS701, Campbell Scientific Instruments, USA) installed at 1.5 m height in an open area near the tower. Rainfall measurements were cross checked and gap filled with data from the nearest Bureau of Meteorology station (Merino #90057).

The measurement period was from 14th December 2017 to 7th July 2021. There are major gaps in the data during January and part of February in 2019, and from late 2020 (starting from October for radiation and from November for the remaining variables) to late January 2021. These gaps were predominantly due to equipment failures and logistical challenges due to the COVID-19 pandemic, which interfered with the access to the site during 2020 and 2021.

Measurements of soil respiration were collected from November 2018 to November 2019 to provide an independent check on the ecosystem respiration derived from the CO₂ flux. Soil respiration was measured using a portable CO₂ gas analyser (EGM-4 - PP Systems) combined with a closed-dynamic chamber (SRC-1 - PP Systems). The measurements were collected in 9 locations within 2 plots located inside the footprint of the EC system. Both plots contained 36 trees, which were equally spaced approximately 3.5 to 4 m between rows and 2 m within rows, covering an area of approximately 250 m². Soil respiration was measured close to the tree mounds and between rows to account for the spatial variability of the fluxes within the plantation. Measurements were collected between 10 am and 1 pm, which yields a good estimate of daily averages (Almagro et al., 2009, 2013; Davidson et al., 1998; Suseela et al., 2012).

Table 3.1: Date, Age (years), average Diameter at Breast Height (DBH, mm), Tree height (m), Leaf Area Index (LAI, $m^2 m^{-2}$), and Tower height (m) during the study period. Tower height refers to the height of the EC tower after being extended.

Date	Age	DBH	Height	LAI	Tower height
	(years)	(mm)	(m)	$(m^2 m^{-2})$	(m)
Dec-17	0.5		0.5		5
Sep-18	1.2	24.07	2.54	0.25	8
Nov-18	1.4	39.35	3.12	0.46	
Mar-19	1.7	48.55	4.01	0.88	
Oct-19	2.4	80.74		2.17	10
Jan-21	3.7	123.76		3.72	14.5
Jul-21	4.0	131.43	11	3.81	

Tree height and diameter at breast height (DBH) were measured to account for tree growth. From September 2018 to March 2019, tree height and DBH were manually measured in the two plots using a ruler and a measuring tape. From September 2019, DBH was measured using automatic dendrometers (DBL60 - ICT International) installed on 10 trees, and tree height was measured again in July 2021 using a clinometer. LAI was estimated from DBH and tree age using allometric equations developed by Veiga (2008) for *E. globulus* in southeast Australia (Feikema et al., 2010; Forrester et al., 2010a).

Table 3.1 reports how the DBH, tree height, and LAI increased with tree age, showing that tree growth occurred at a faster rate between September and November.

3.2.3 Data analysis

The site was equipped with a CR6 datalogger running the Easy Flux program (Campbell Scientific Inc, USA). The Easy Flux algorithm allows the logger to correct fluxes from raw high frequency data (10 Hz) into half-hourly means. It also includes the corrections commonly applied in post-processing software packages. The main corrections include detection of spikes and high frequency filtering, lag H_2O/CO_2 against sonic wind measurements (Foken et al., 2012), double rotation method (Lee et al., 2005), corrections for density fluctuation (WPL correction) (Webb et al., 1980), frequency response correction and sensor separation (Aubinet et al., 2012; van Dijk, 2002). Additionally, the datalogger program grades the relative quality of the fluxes following Foken et al. (2012) into categories from 1 to 9, where 1 is the best data and 9 the poorest data. Further quality assurance/quality control was applied to filter and eliminate remaining spikes and invalid data caused by heavy rainfall or system failures.

The accuracy of the dataset was assessed by calculating the energy balance closure of the site (Q - G = LE + H), with Q being the net radiation, G the soil heat flux, LE the latent heat flux, and H the sensible heat flux. Following Leuning et al. (2012), the energy balance was calculated considering the daily sum of fluxes (not shown) to avoid phase lags caused

by incorrect estimates of energy storage in the soil and biomass. The slope of the regression between the right- and left-hand sides of the energy balance equation was 0.82 and the R^2 was 0.94.

The Net Ecosystem Exchange (NEE) under low turbulence conditions was identified and filtered by calculating the year-specific u^* threshold for the site. The u^* threshold was calculated to filter the carbon flux measured under low turbulence conditions, and a value for each year was calculated following the change-point detection method using 1,000 iterations (Barr et al., 2013); this was applied for day- and night-time carbon fluxes following McHugh et al. (2017). NEE measurements presenting a value of u^* lower than the calculated values of 0.3 (2018) and 0.39 m s⁻¹ (2019, 2020, 2021) were excluded to avoid bias in carbon flux measurements.

Gaps originated by screening and exclusion of fluxes were filled by applying a marginal distribution sampling (MDS) algorithm (Falge et al., 2001; Reichstein et al., 2005b), which considers not only the covariance between fluxes and meteorological data but also temporal auto-correlation of fluxes (Aubinet et al., 2012; Mahabbati et al., 2021). The gap filling method was performed using ReddyProc, an online tool and R package developed by the Max Planck Institute, which has been extensively used in the literature (Wutzler et al., 2018). In this method, the missing values are replaced by the average value under similar meteorological conditions.

The resulting data set was subsequently used for the partitioning of NEE into Gross Primary Productivity (GPP) and Ecosystem Respiration (ER) using the day-time based flux-partitioning algorithm introduced by Lasslop et al. (2010). In this approach, respiration is estimated from fitting a rectangular hyperbolic light–response curve. This curve is used to fit daytime NEE measurements and, as a result, respiration is estimated from the intercept of the ordinate, avoiding the use of potentially problematic nighttime data (Falge et al., 2001).

Gap filled fluxes were used to calculate annual budgets. Considering that some months were missing in the data set (January - February 2019; October/November 2020 - January 2021), the annual budgets from years with missing months were calculated using the data from previous years. Full months of missing fluxes cannot be gap filled by the most common gap filling methods; thus, this approach was used to provide reasonable estimates of yearly budgets. For January - February 2019, values from the same months in 2018 were used. For the gap in 2020, values from October or November 2019 were used. Since young trees are constantly developing and growing, the previous year fluxes were used in order to better represent the stage of tree development for that missing period.

The micrometeorological convention for fluxes is relative to the atmosphere, i.e. positive fluxes represent a gain to the atmosphere and negative fluxes a loss. The ecological convention is relative to the ecosystem, i.e. positive (negative) fluxes represents ecosystem carbon gain (loss). When presenting results regarding the fluxes between the plantation and the atmosphere, we use the micrometereological convention. The ecological convention is used for the annual

budgets, with the terms Net Ecosystem Production (NEP), defined as NEP = -NEE, and Gross Ecosystem Production (GEP) defined as GEP = -GPP. Although NEP should also account for effects of dissolved inorganic carbon export/import (Chapin et al., 2006), we assume this to be negligible for our study.

The half-hourly gap-filled latent heat flux (LE) (W m⁻²) was accumulated to daily time steps and converted to ET (mm d⁻¹ = kgH₂O m⁻²d⁻¹), and GPP was converted from μ molCO₂ m⁻²s⁻¹ to GEP (gC m⁻²s⁻¹), and accumulated monthly. Water Use Efficiency (WUE) was derived as WUE = GEP/ET (gC kg⁻¹H₂O), using monthly accumulations of GEP and ET (Dye, 2000). Similarly, the ER/GEP monthly ratios were calculated using monthly accumulations of ER and GEP.

3.3 Results

3.3.1 Energy fluxes

The daily energy fluxes showed similar patterns during the monitored period (Figure 3.3). Sensible heat flux (H) dominated the energy partition between December and April, with the latent heat flux (LE) being larger than H during the rest of the year. In December each year, the increasing radiation and temperature favoured higher sensible heat fluxes, with a reduction in LE due to lower precipitation and soil water availability (see Figure 3.2)

The daily mean net radiation (Q) and H peaked in summer, with a daily mean maximum of 232.82, 238.03, 217.78 W m⁻² during 2018, 2019, and 2020, respectively, for Q, and 152.81, 150.09, 156.04 W m⁻², respectively, for H. Daily mean LE peaked in spring (September - November), with a maximum of 118.66, 133.62, and 129.70 W m⁻² during 2018 - 2020. During winter, when Q was at its minimum, H decreased considerably reaching negative values. This happened when the air temperature was higher than the soil temperature (usually with cold weather and wetter soils) creating a positive gradient of temperature from the air to the soil (Bonan, 2016).

The yearly mean Bowen ratios (β = H/LE) were 1.18, 0.88, and 0.88 for 2018, 2019, and 2020, respectively. The Bowen ratio was higher during summer and lower during winter, when it was close to zero and slightly negative in some periods. During summer, β reached a mean value of 3.04 in 2018, decreasing to 2.87 in summer 2019, and to 2.13 in summer 2020. The larger value in 2018 may reflect the physiological structure of the plantation, although as noted previously, the season was also characterised by lower rainfall and soil moisture.



Figure 3.3: Top: daily mean net radiation (Q), sensible heat (H), and latent heat fluxes (LE) (W m⁻²) throughout the monitored period. Bold lines represent weekly moving averages. Bottom: weekly moving averages of Bowen ratio (β). Vertical dashed lines represent transition between years.

3.3.2 Water and carbon fluxes

Mean daily ET patterns were similar between the years, with yearly mean daily values equaling about 1.50 mm in the 3 full years of measurements, with a daily maximum of 4.27, 5.1, and 5.1 mm d⁻¹ for 2018, 2019, and 2020, respectively (Figure 3.4). Differences could be seen in late January and February 2020, when an increase in rainfall during the summer (32 mm in January and February 2018 against 75.4 mm for the same period in 2020) generated more ET in comparison to 2018.

Differently from ET, NEE showed a large intra- and inter-annual variability across all three monitored years. The plantation changed from behaving as a carbon source during most of 2018 and 2019 to being a carbon sink for most of 2020. The minimum daily totals of NEE were -7.41 (2018), -8.04 (2019), and -6.90 (2020) $gCm^{-2}d^{-1}$. In 2018, the ecosystem acted as a carbon source early in the year (summer and autumn), became a sink in winter and early spring, and rapidly switched to act as a source with an average NEE of 3.42 $gCm^{-2}d^{-1}$ during the last 40 days of the year. The strong seasonal cycle of NEE became less marked as trees grew in 2019 and 2020. Even though 2018 and 2019 had minimum values of NEE lower than 2020, the number of days the plantation behaved as a sink was 131 in 2018, 152 in 2019, and 250 in 2020. Taking into account the gaps in 2019 and 2020, this difference would almost certainly be even larger.



Figure 3.4: Daily total evapotranspiration (ET, $mm d^{-1}$) and mean daily net ecosystem exchange (NEE, $gCm^{-2}d^{-1}$) during 2018, 2019, and 2020. Thick lines represent weekly moving averages. The horizontal dashed line represents carbon neutrality.

Figure 3.5 shows the diurnal cycles of ET and NEE in different seasons during the three years from 2018 to 2020. The ET daily cycles showed similar values between years. Major differences were observed during summer, when the daily ET peak was lower in 2019 and higher in 2020. However, data are missing in January and part of February in 2019, certainly aggravating the differences between the years. ET peaks of 0.27, 0.25, and 0.22 mm h^{-1} were reached in spring in the three different years. Both maximum hourly CO₂ assimilation and ET were observed during spring, highlighting a period of high productivity for the trees.

Carbon fluxes partition

The daily accumulated GPP varied from -12.74 to -0.54 in 2018, -12.86 to -0.67 in 2019, and -9.3 to -1.21 $gCm^{-2}d^{-1}$ in 2020; the ER from 12.04 to 0.06 in 2018, 15.25 to 0.55 in 2019, and 9.52 to 0.53 $gCm^{-2}d^{-1}$ in 2020. Although the peak magnitudes were greater in 2018 and 2019, GPP and NEE showed a larger sink with stand growth, especially when comparing values registered during January and February over the years (Figure 3.6). Despite the data gap in 2020, the first part of 2021 shows that larger values of GPP are achieved in the first part of the year compared to 2018 and 2019. The modeled NEE is calculated as the sum of the modeled GPP and ER; therefore, it is not equal to the measured NEE and it is only shown in Figure 3.6 for the visualization and validation of the partition process (Lasslop et al., 2010).



Figure 3.5: Diurnal courses (hourly means) of evapotranspiration (ET, mm h⁻¹) and net ecosystem exchange (NEE, $\mu molCO_2m^{-2}s^{-1}$) across Summer (Dec - Feb), Autumn (Mar - May), Winter (Jun - Aug), and Spring (Sep - Nov). The horizontal dashed line represents carbon neutrality. Vertical bars represent the standard error.

Soil respiration accounts for a considerable portion of ER (Law et al., 1999), and it is the largest carbon source to the atmosphere, and the second largest carbon flux after GPP (Yuste et al., 2005). Soil respiration measurements were averaged for both plots and ranged from 1.39 to 11.22 $gCm^{-2}d^{-1}$ over the measurement period. As shown in Figure 3.6, the majority of soil respiration points are smaller and close to the ER value observed for that day, thereby supporting the results of the partition of CO_2 fluxes.

Water use efficiency

After converting GPP to GEP, WUE was calculated using monthly accumulations of GEP and ET (Figure 3.7). During the monitored period, the WUE was minimum in February 2018 (1.48 $gC kg^{-1}H_2O$), and maximum in June 2020 (5.63 $gC kg^{-1}H_2O$). The yearly maximum WUE was 4.15 in 2018, 4.13 in 2019, and 5.63 $gC kg^{-1}H_2O$ in 2020. WUE increased continuously during the monitoring period following tree growth, averaging 2.86 $gC kg^{-1}H_2O$ in 2018, 3.48 $gC kg^{-1}H_2O$ in 2019, and 4.3 $gC kg^{-1}H_2O$ in 2020. Even during winter, when ET and GEP



Figure 3.6: Daily accumulation of gross primary production (GPP), modeled net ecosystem exchange (NEE), and ecosystem respiration (ER) $(gCm^{-2}d^{-1})$. Shaded areas represent the standard deviation of the variables. Dots represent discrete soil respiration measurements.

were minimum, WUE increased from 3.5 in June 2018, to 3.99 in June 2019, reaching 5.63 $\rm gC~kg^{-1}H_2O$ in June 2020.



Figure 3.7: Monthly accumulation of gross ecosystem production (GEP, $\rm gC\ m^{-2}month^{-1}$) and evapotranspiration (ET, $\rm mm\ month^{-1}$), and water use efficiency (WUE) ($\rm gC\ kg^{-1}_{H_2O}$) for the study site.

Table 3.2: Annual carbon fluxes (modeled and measured NEP, GEP, and ER), ($gC m^{-2} y^{-1}$), evapotranspiration (ET, mm y⁻¹) and percentage of ET to Precipitation (P y⁻¹) for 2018, 2019, and 2020. The values are presented without estimating the missing data in incomplete months, and using the previous year to estimate missing data.

Without estimating missing months							
	NEP _{measured}	NEP _{modeled}	GEP	ER	ET		
	$(gC m^{-2}y^{-1})$	$(gC m^{-2}y^{-1})$	$(gC m^{-2}y^{-1})$	$(gC m^{-2}y^{-1})$	$(mm y^{-1})$		
2018	-42.54	-38.23	1538.78	1577.02	488.30		
2019	131.09	132.94	1623.82	1490.87	449.92		
2020	398.89	290.94	1312.52	1021.58	301.35		
Estimating missing months							
2018	-54.6	-49.37	1585.04	1634.42	517.47 (70% of P)		
2019	90.30	96.30	1709.50	1613.20	516.12 (71% of P)		
2020	529.81	556.35	2093.61	1537.25	575.74 (74% of P)		

Annual water and carbon budgets

Table 3.2 shows annual budgets with and without estimation of the missing months. The yearly ET slightly increased between 2018 and 2020, representing a 4% increase in ET/P ratios. On the other hand, the annual carbon fluxes (NEP and GEP) increased yearly with tree growth, with a shift from the ecosystem behaving as a carbon source (2018) to a carbon sink (2020) (Figure 3.8). Similarly to 2018, 2019 was a cumulative carbon source until late November when the plantation became a net sink. From 2018 to 2020, GEP increased by more than 500 gC m⁻². On the other hand, ER slightly decreased over the years, registering a difference of 97.17 gC m⁻² from 2018 to 2020. Increases in GEP with respective decreases in ER resulted in a turning point for the site, when the ecosystem started to behave as a continuous sink.

An almost constant yearly ET increased the WUE throughout the study period, given the increase in growth, and hence GEP, over successive growing seasons. For example, between 2018 and 2020, annual GEP increased by 508 $\rm gC~m^{-2}$ (32% increase from 2018), while ET varied by 58.27 mm (11% increase from 2018), which represented a change from 70% to 74% of P over this period.

3.4 Discussion

3.4.1 Water and carbon fluxes with stand age

In growing ecosystems, the combined effect of environmental variability and growth makes it challenging to determine the drivers of ET, GPP, and NEE (Chan et al., 2018; Coursolle et al., 2012; Kwon et al., 2018; Ma et al., 2019). Environmental conditions and seasonal climate patterns were mostly similar over the years of the study, suggesting that the differences in

CHAPTER 3. WATER AND CARBON FLUXES IN A YOUNG *EUCALYPTUS GLOBULUS* PLANTATION



Figure 3.8: Cumulative Net Ecosystem Production (NEP, $gC m^{-2}$), Gross Ecosystem Production (GEP, $gC m^{-2}$), Ecosystem Respiration (ER, $gC m^{-2}$), and Evapotranspiration (ET, mm) for 2018, 2019, and 2020. The horizontal line represents carbon neutrality.

water and carbon fluxes observed during the period 2018 - 2021 can be attributed to structural changes related to tree growth, and not to the influence of meteorological variability.

As the plantation underwent structural changes between 2018 and 2021, increases in annual GEP and NEP were stronger than the increases in ET (Figure 3.8). This was likely caused by a gradual reduction in the understory weeds and grasses combined with tree LAI increases as the stand developed. LAI expansion increases shaded areas and interception, causing a decline in radiation and water for the understory vegetation (Irvine et al., 2002; Kwon et al., 2018; Vickers et al., 2012). At the study site, the understory was mostly composed of grasses, which have high transpiration rates when water and energy are available, but perish when environmental conditions become unfavorable due to higher temperatures and water stress (Pereira et al., 2007; Wiesner et al., 2019). In contrast, trees generally have more capacity to adjust transpiration, through the regulation of their stomatal conductance, which makes trees more hydrologically conservative (Jones, 2013; O'Grady et al., 1999; Pita et al., 2013).

Vegetation growth can result in transpiration increases, but the developing canopy will also offer more stomatal control over ET during drier periods than bare soil and grasses. This may have

CHAPTER 3. WATER AND CARBON FLUXES IN A YOUNG *EUCALYPTUS GLOBULUS* PLANTATION

influenced the annual ET values registered over the monitored period (Figure 3.4). Ecosystemscale measurements of ET comprise not only tree transpiration, but also soil evaporation, understory transpiration, and canopy-intercepted water. Thus, the stages of tree development, such as changes from sparser to closer canopies, can shift the partition of ET over time, with tree transpiration increasing with tree growth as soil evaporation and understory transpiration decrease. This was observed in Ntshidi et al. (2021), where transpiration by the dense understory of young orchard stands accounted for close to 40% of the total ET. Likewise, the ET of old and young ponderosa pine plantations in Irvine et al. (2002) did not differ significantly during their study period. This was explained by the considerable transpiration by understory shrubs, which accounted for 40% of the site's LAI. For the old stands, understory vegetation was sparse and did not contribute significantly to the LAI. When studying the same ponderosa pine plantations, Law et al. (2001) identified that understory transpiration and soil evaporation were inversely proportional to the LAI, and that differences in LAI strongly influenced maximum evapotranspiration between the young and old stands.

Therefore, as the LAI increases and the canopy starts closing, a larger share of ET can switch from understory transpiration and evaporation to tree transpiration until the plantation reaches its peak (Kwon et al., 2018; Vickers et al., 2012). A similar pattern was observed for an age series of *E. globulus* trees located in Victoria (approximately 200 km from our study site), where tree transpiration and LAI increased until reaching their peak, declining afterwards (Forrester et al., 2010b). For fast-growing *E. grandis* plantations in Brazil, Almeida et al. (2007) reported a reduction in tree transpiration, LAI, and annual biomass increment when the stand age exceeded 4 years.

3.4.2 From carbon source to sink

As the plantation grows older, the assimilation of carbon in tree biomass eventually exceeds the loss from respiration and, as a consequence, NEP becomes consistently larger (Figure 3.8) (Bergeron et al., 2008; Heijmans et al., 2004; Rodrigues et al., 2011). The prevalence of GEP over ER at the site increased over time with the ER/GEP ratio decreasing from 1.03 in 2018 to 0.73 in 2020. For the global flux network, Baldocchi (2008) found an ER/GEP ratio for undisturbed ecosystems of 0.77. For Australian ecosystems, Beringer et al. (2016) estimated 0.79. Therefore, the plantation fast growing capacity allowed the site to switch to an ER/GEP ratio comparable to an undisturbed site in less than three years after establishment (Figure 3.9). Apart from June 2021, the plantation registered monthly values of ER/GEP < 1 starting from September 2019. This is associated with a continuously increasing NEP, which experienced a consistently increasing trend from September 2019 to the end of 2020 (Figure 3.8).

Increases in annual NEP were reported by Vickers et al. (2012) when comparing mature to young ponderosa pine stands. During their 5 year study period the NEP did not differ for mature



Figure 3.9: Monthly ER/GEP ratio. The horizontal line at ER/GEP=1 represents the transition between carbon source (ER/GEP > 1) and sink (ER/GEP < 1) behaviour. Vertical dashed lines represent transitions between years.

stands, whereas young stands registered a yearly increase in NEP values. The increasing NEP was associated with a 64% expansion in LAI over the 5 years. Likewise, for a young pine plantation after clear-cutting, Bracho et al. (2012) identified that carbon assimilated by developing trees was driven by LAI increases, which controlled the increments in GEP and NEP.

Only a few studies have addressed the net and gross carbon assimilated by young *Eucalyptus* plantations. Considering the particularities of each study site, it is difficult to establish a direct comparison with the results from our study. Factors such as climate, management, and age can sharply change the carbon uptake patterns of tree stands (Baldocchi et al., 2018).

For Brazilian *E. grandis* stands, which are usually managed on rotations of 6 years, the NEP increased from 993 to 1400 $gC m^{-2}$ from the second to the third year of the management cycle (Cabral et al., 2011). Harper et al. (2014) found a mean GEP of 2720 $gC m^{-2}$ when looking at 2 to 5 year old blue gum stands in Western Australia. Rodrigues et al. (2011) studied a Portuguese blue gum plantation before (8 to 12 year old trees) and after a felling in 2007 (1 to 3 year old trees) from 2002 to 2009. For this site, both the NEP and GEP reached their maximum value (865.56 and 2206.04 $gC m^{-2}$, respectively) for 8 year old trees when precipitation was also highest, and a minimum value of 11 $gC m^{-2}$ (NEP) and 939.44 $gC m^{-2}$ (GEP) soon after the felling. As the trees regenerated, NEP and GEP strongly increased, reaching 209.01 $gC m^{-2}$ and 1294.05 $gC m^{-2}$, respectively, in 2009.

These studies show the fast-growing capacity of Eucalyptus stands, which influences NEP and

CHAPTER 3. WATER AND CARBON FLUXES IN A YOUNG *EUCALYPTUS GLOBULUS* PLANTATION

GEP at the ecosystem scale. Therefore, considerable increases in NEP and GEP following tree development occur for different climates and tree species. The results from the previous studies are comparable to the NEP of 556.35 $\rm gC\ m^{-2}y^{-1}$ and GEP of 2093.61 $\rm gC\ m^{-2}y^{-1}$ estimated in 2020 for the 3 years old trees at the study site (Table 3.2). At the study site, NEP started increasing consistently in September 2019 after about 2 years since the plantation establishment (Figure 3.8). This point can be highly variable according to differences in factors such as species, climate, understory composition, and management (Cleverly et al., 2020). For example, a Canadian pine forest became a consistent sink after 4 years (Bracho et al., 2012), and (Amiro et al., 2010) estimated a period of 10 to 20 years after a disturbance for forests across North America. Contrastingly, semi-arid ecosystems have different patterns of carbon assimilation and can behave as a source or a sink depending on wet and dry periods, which affect the carbon budget seasonally and inter-annually. This makes it difficult to estimate a period in which trees will only act as a carbon sink (Hinko-Najera et al., 2017; Tarin et al., 2020).

The present study estimated the changes in ecosystem scale carbon fluxes with stand growth. However, it is important to highlight that plantations experience other carbon inputs and losses originated from harvesting, logging, and fertilization. These components, which are part of the Net Biome Production (NBP), reflect the long-term load of CO_2 (Kutsch et al., 2010; Pregitzer and Euskirchen, 2004; Smith et al., 2008) and were not captured in this study.

3.4.3 Water use efficiency

The WUE continuously increased with stand age over the monitored period (Figure 3.7). This pattern could also be verified by other studies in growing ecosystems that investigated different species and climates (Chan et al., 2018; Irvine et al., 2004; Jassal et al., 2009; Skubel et al., 2015). For a white pine plantation in Canada (CA-TP1, Fluxnet Canada) planted in 2003, the WUE increased from 1.6 in 2003 to 4.29 $\rm gC~kg^{-1}H_2O$ at the end of the monitoring period in 2016, reaching a maximum of 5.11 $\rm gC~kg^{-1}H_2O$ in 2015 (Chan et al., 2018). Similarly, for the pine plantation described in Skubel et al. (2015), the WUE increased from 2.9 when stands were 6 years to 4.5 $\rm gC~kg^{-1}H_2O$ when they reached 11 years. In both studies, increasing productivity was a result of a larger increase in GEP than in ET, as a consequence of expanding LAI, sapwood area, and root system.

Factors such as water availability and VPD can alter how carbon is allocated in trees, favoring more stem biomass (wet periods) or root development (dry periods), thus altering the tree growth and GEP (Moroni et al., 2003; Ryan et al., 2010; Stape et al., 2004; Tfwala et al., 2019). At our study site, the WUE followed variations in the SWC. The high SWC in the first 30 cm was associated with increases in tree transpiration and carbon assimilation, since available soil water is a major factor limiting the growth in young plantations (Irvine et al., 2002; White et al.,

2016). WUE was higher when water was readily available and the ecosystem was not energy limited. During these periods, trees could transpire and assimilate CO_2 without suffering major stress or losing a great amount of water due to high temperatures.

The WUE was considerably lower during summer in 2018 (January - February). The reduced summer rainfall, and higher temperatures and VPD played important roles in this reduction, but tree development was also a crucial factor. During summer 2018, the trees were less than 1 year old; therefore, this year's summer GEP could be attributed to limited photosynthetic capacity and shallow roots, which restricted transpiration, and hence GEP, since trees could not extract water from deeper soil layers (Benyon et al., 2006; Jassal et al., 2009; Skubel et al., 2015). The sparse canopy and reduced LAI favoured more soil evaporation, which contributed to ET. Since understory grasses perish during summer, GEP contributions from the understory were minimal. In summer 2019 and 2020, the WUE consistently increased and was more than double that in 2018. For these years, a combination of higher SWC at the soil surface and increasing diameter and LAI with stand age resulted in a greater GEP and consequently higher WUE.

Few available studies investigated the variations in WUE in *Eucalyptus* stands. For mature blue gum trees (12 years) in Portugal, Pereira et al. (1986) calculated monthly variations in the WUE, registering an average of 4.2 $gC kg^{-1}H_2O$. Also in Portugal, Rodrigues et al. (2011) registered a maximum WUE of 4.84 $gC kg^{-1}H_2O$ for mature trees, and 1.62 $gC kg^{-1}H_2O$ immediately after tree felling. Two years after the felling, the WUE increased to 2.35 $gC kg^{-1}H_2O$. For a native *Eucalyptus* forest minimally managed since the 1970s, Griebel et al. (2020) found, from January 2013 to November 2015, an average baseline WUE of 3.02 $gC kg^{-1}H_2O$, with lower values during drier and hotter periods.

For our study site, the monthly averaged WUE surpassed the value of a native forest (Griebel et al., 2020) and a young commercial plantation after felling (Rodrigues et al., 2011) after approximately one year (2019) and, after that, even during summer, when the SWC reached its lowest yearly values, WUE values were above $3 \text{ gC kg}^{-1}\text{H}_2\text{O}$. For 2020 and 2021, most months were already in the range of those registered in mature closed-canopy blue gums (Pereira et al., 1986; Rodrigues et al., 2011).

The different responses in GEP, NEP, and WUE when comparing young and mature stands, as well as different species, climate, and treatment (i.e., natural forest and commercial plantation), highlight the need to evaluate and study stand age and species-specific characteristics. Therefore, management practices should be planned in order to maximise WUE, and thus productivity and water conservation, taking into account tree physiological dynamics and climate characteristics.

3.5 Conclusion

Fluxes of energy, water, and carbon were measured in a young *Eucalyptus globulus* plantation during the first four years after its establishment. During the study period, the NEP, GEP, and ecosystem WUE sharply increased with tree development and growth. As a result, the plantation switched from being a carbon source to a carbon sink after about 2 years, acting consistently as a sink thereafter. Tree growth and development strongly influenced the increases in NEP, GEP, and ecosystem WUE. Increases in GEP, which offset the ER resulting in a positive NEP, were caused by the expansion in LAI and, thus, photosynthesis capacity.

The results from our study show that local conditions are important in determining the water use of a plantation and that, in the early stages of growth, the increase in water consumption might be outweighed by the large gain in carbon assimilation. The water used by the plantation might, however, increase over time as the trees mature, as estimated by Dresel et al. (2018) at the same site.

The results emphasize that stands at different stages of development can show markedly different carbon fluxes and WUE. Understanding the trade-offs between water use and carbon assimilation in commercial plantations is important given the economic and ecological impacts caused by plantation establishment and expansion. Therefore, continued cultivation of large scale commercial plantations requires the application of efficient management practices.

Chapter 4

Tree hydrodynamic model development: FETCH3

This chapter contains large parts of the article and supplementary material of "Tree hydrodynamic modelling of the soil plant atmosphere continuum using FETCH3" published in Geoscientific Model Development.

4.1 Aim

The aim of this study is to present The Finite-difference Ecosystem-scale Tree Crown Hydrodynamics (FETCH3), an open source and open access tree hydrodynamic model for the simulation of the temporal and vertical dynamics of water storage and fluxes from the soil to the atmosphere, accounting for the vegetation response to environmental conditions and soil water availability. As a porous-media model, FETCH3 solves a system of three partial differential equations in a 1D domain to describe the water flow through the soil, root xylem, and stem xylem. The primary novelty of the model is a full coupling of the soil, roots, and stem xylem by clarifying the links between these 3 components of the system when re-scaling the processes into a single, continuous vertical dimension. The numerical formulation of FETCH3 was verified against exact solutions of simplified expressions of the equations, the model performance was evaluated against observational data collected during six months from a case study, and the inclusion of details of the canopy structure and stem xylem capacitance is discussed.

4.2 Model description

4.2.1 Model overview

FETCH3 builds upon FETCH2 (Mirfenderesgi et al., 2016, 2018), which is based on its precursor, the Finite Element Tree Crown Hydrodynamics (FETCH) model (Bohrer et al., 2005). FETCH simulates water flow along a tree's stem and branches accounting for the branch structure in three dimensions. Simulating the three-dimensional tree crown structure is computational demanding and can solely be applied to a single tree. As a result, FETCH2 was developed to offer a more mechanistic approach that could be scaled to entire ecosystems. To achieve this, FETCH2 simplifies branches along the vertical direction, leading to a 1D model; the equations in FETCH2 are solved using a finite difference scheme (Mirfenderesgi et al., 2016).

Similarly to FETCH and FETCH2, FETCH3 assumes that the water movement in the xylem resembles flow in porous media; as in FETCH2, a macroscopic approach is used to simulate the water fluxes across the soil, roots, and stems with the fluxes being described in one dimension along the vertical direction (Figure 4.1). As a development from FETCH2, FETCH3 presents a clearer link between the three different components of the system (i.e., soil, roots and stem), based on the conservation of water in each of the components. In its 1D domain, FETCH3 allows for the vertical variation of the soil, root xylem, and stem xylem hydraulic parameters, which are able to vary along the tree. As a result, when combined, the quantities in the equations for the roots and stem are scaled to a reference ground area, consistently with the Richardson-Richards equation for the soil. This guarantees the conservation of mass as water flows from one component to the other. The system of equations in FETCH3 is also solved differently from FETCH2. The equations in FETCH3 are discretised using the method by Celia et al. (1990) generating a system of algebraic equations combined into a single matrix, that is solved at the same time to guarantee the conservation of mass across the whole system comprising soil, roots and stem.

In FETCH3, water in a variably saturated soil is exchanged between the soil and the root system. The water flow in the soil is modelled using the Richardson-Richards equation with a term simulating the exchange of water between the soil and the roots. This term is a function of the difference in water potential between the soil and root layers; it thus results in a water sink during the day, when the water potential in the roots is low due to water loss by transpiration, but may act as a source of water to the soil during some nights, depending on the water content in different soil layers. The boundary conditions at the top and bottom of the soil column can be expressed as a flux or a value of soil water potential.

Water fluxes within roots are likewise modelled with a Richardson-Richards type equation with the same term (of the opposite sign) representing water exchange between roots and soil. Soil

and roots are coupled through this term, such that a sink of water in the soil is a source of water in the roots, and vice versa. The transfer of water between the soil and the roots is modulated by a conductance, representing the radial resistance between the bulk soil, roots surface, and root xylem, and a stress function, accounting for the reduction of the root water uptake associated with different soil moisture conditions possibly leading to water and oxygen stress. The 3D root architecture is scaled along the vertical dimension using a vertical mass distribution of the roots and an index that summarizes the extent of lateral root area per unit of ground area (Quijano and Kumar, 2015). Water fluxes through the soil are defined as the mass flow of water per unit of ground volume. Thus, when referring the water fluxes in the roots to the same water mass that was contributed by the soil, the water storage and water fluxes within the roots must be re-scaled to the ground volume and thus, when normalized by unit depth, to the ground area.

A similar approach is used to model the water flow in the above-ground xylem, which is also described with a Richardson-Richards type equation with a sink term associated with transpiration losses from the canopy to the air. This equation is commonly used to simulate water flow for a single tree (Chuang et al., 2006); however, in order to correctly couple the above-ground and the below-ground components of the system, both equations must refer the water flux to units of ground area. This ensures the water mass balance and the continuity of the fluxes from soil through the root system to the above-ground stem xylem and ultimately to the air. This conservation of flux throughout the system is important but not trivial, as the amount of roots that fits within a reference area of soil, for example, is different than the xylem area or leaf area which are located above the same area of soil. FETCH3 simulates variable plant water storage below- and above-ground by using a dynamic capacitance function that can vary in time (t) and space (z). Accounting for whole-plant water storage enables different model applications in which plant storage plays an important role, such as water use efficiency and plant hydraulic stress during dry periods (Huang et al., 2017; Li et al., 2021).

The complete system of equations simulates the water fluxes assuming a spatial distribution of trees, and their associated roots, stem xylem, and leaves, with an average cross-sectional area per unit of ground area. In this manner, FETCH3 presents a novel up-scaling technique required to properly calculate tree transpiration from small and large areas, such as a forest stand or plantations, assuming that all trees within the simulated area are similar on their dimensions and conductive parameters.

4.2.2 Governing equations

The water flow within the soil, and root and stem xylem is described as flow in porous media. Equations for the mass conservation are combined with the Darcy's equation extended to unsaturated porous media to derive an equation for the water potential in the soil, roots, and



Figure 4.1: Representation of the coupling process between soil, root xylem, and stem xylem applied in the model, where A_s represents a reference ground area, dz an infinitesimal depth over an area (m), z the vertical coordinate (m), V volume of soil (m³), ρ the density of water (kg m⁻³), F_{in} (kg s⁻¹) the water fluxes entering and F_{out} (kg s⁻¹) exiting the volume, A_r/A_s (m²_{root} m⁻²_{ground}) the root xylem cross area index, A_x/A_s (m²_{xylem}m⁻²_{ground}) the stem xylem cross area index, S (s⁻¹) the rate at which water is extracted from the soil and enter the root xylem, and S_x (m² s⁻¹) is the flow of water leaving the stem per unit of vertical length due to transpiration.

stem.

Soil

According to the schematic shown in Figure. 4.1, for mass conservation, considering a volume of soil with an infinitesimal depth over an area A_s , $dV_s = A_s dz$, changes of the mass of water, M_s (kg), over time within this volume are due to the difference between the water fluxes entering, F_{in} (kg s⁻¹), and exiting, F_{out} (kg s⁻¹), the volume, and the water exchanged with the roots. This can be expressed as

$$\frac{\partial M_s}{\partial t} = F_{in} - F_{out} - \rho S dV_s, \tag{4.1}$$

where $dM_s = \rho \theta_s dV_s$, ρ is the density of water (kg m⁻³), θ_s is the soil volumetric water content, t (s) is time, and S (s⁻¹) is the rate at which water is extracted from the soil per unit of mass of

water contained in dV_s . The term F_{out} can be written as

$$F_{out} = F_{in} + \frac{\partial F}{\partial z} dz, \qquad (4.2)$$

which, substituting Eq. (4.2) into Eq. (4.1), yields

$$\rho \frac{\partial \theta_s}{\partial t} A_s dz = -\frac{\partial F}{\partial z} dz - \rho S A_s dz.$$
(4.3)

The flux F can be written as

$$F = \rho \ v_s \ A_s, \tag{4.4}$$

where v_s is the Darcy's velocity (m s⁻¹), expressed as

$$v_s = -K_s \left(\frac{\partial \Phi_s}{\partial z} + \rho g\right),\tag{4.5}$$

with K_s being the hydraulic conductance of the soil (m² s⁻¹Pa⁻¹), *g* the gravitational constant, and Φ_s the soil water potential (Pa), which in FETCH3 refers to the matric component.

Eq. (4.3) can thus be simplified into

$$C_s \frac{\partial \Phi_s}{\partial t} = \frac{d\theta_s}{d\Phi_s} \frac{\partial \Phi_s}{\partial t} = \frac{\partial}{\partial z} \left[K_s \left(\frac{\partial \Phi_s}{\partial z} + \rho g \right) \right] - S.$$
(4.6)

where $C_s = \frac{\partial \theta}{\partial \Phi_s}$ (Pa⁻¹) is the soil water capacitance, and z (m) is distance along vertical direction, assuming positive represents upward flux. The relationships between K_s , Φ_s and θ_s are modelled according to van Genuchten (1980) (refer to Section 4.5).

Roots

Following the same procedure as the soil, the conservation of water mass in the roots, with $dM_r = \rho \theta_r A_r dz$, results in

$$\rho \frac{\partial (\theta_r A_r)}{\partial t} dz = \rho \frac{\partial}{\partial z} \left[K_r A_r \left(\frac{\partial \Phi_r}{\partial z} + \rho g \right) \right] dz + \rho S A_s dz.$$
(4.7)

Dividing Eq. (4.7) by $\rho A_s dz$ leads to

$$C_r \frac{\partial \Phi_r}{\partial t} = \frac{d}{d\Phi_r} \left(\frac{\theta_r A_r}{A_s}\right) \frac{\partial \Phi_r}{\partial t} = \frac{\partial}{\partial z} \left[K_r \frac{A_r}{A_s} \left(\frac{\partial \Phi_r}{\partial z} + \rho g\right)\right] + S,$$
(4.8)

where C_r (Pa⁻¹) is the root xylem water capacitance, Φ_r (Pa) is the root water potential, θ_r is

the root volumetric water content, K_r (m² s⁻¹ Pa⁻¹) is the effective axial hydraulic conductivity of the roots, and A_r/A_s (m²_{root} m⁻²_{ground}) is the root cross sectional area index, representing the total root cross-sectional area at a given elevation per unit of ground area.

Stem

The conservation of water mass in the stems, with $dM_x = \rho \theta_x A_x dz$, results in

$$\rho \frac{\partial(\theta_x A_x)}{\partial t} dz = \rho \frac{\partial}{\partial z} \left[K_x A_x \left(\frac{\partial \Phi_x}{\partial z} + \rho g \right) \right] dz - \rho S_x dz, \tag{4.9}$$

where Φ_x (Pa) is the stem xylem water potential, K_x (m² s⁻¹ Pa⁻¹) is the effective axial hydraulic conductivity of the stem xylem, and S_x (m² s⁻¹) is the flow of water leaving the stem per unit of vertical length due to transpiration. It is convenient to express S_x as a function of transpiration per unit of ground area, T (m s⁻¹), which is most commonly measured. Because transpiration along the stem depends on the leaf area, S_x can be expressed as lA_sT , with l(z) (m² m⁻² m⁻¹) being the leaf area per ground area per unit of stem height, and A_s the ground area.

Dividing Eq. (4.9) by $\rho A_s dz$, such that water fluxes across soil, roots, and stem are expressed in terms of ground area, one obtains

$$C_x \frac{\partial \Phi_x}{\partial t} = \frac{d}{d\Phi_x} \left(\frac{\theta_x A_x}{A_s}\right) \frac{\partial \Phi_x}{\partial t} = \frac{\partial}{\partial z} \left[K_x \frac{A_x}{A_s} \left(\frac{\partial \Phi_x}{\partial z} + \rho g\right) \right] - \frac{S_x}{A_s},$$
(4.10)

where C_x (Pa⁻¹) is the stem xylem water capacitance, θ_x (m³ m⁻³) is the stem xylem volumetric water content, and A_x/A_s (m²_{stem}m⁻²_{ground}) is the stem xylem cross-sectional area index. This index can be calculated from the tree sapwood area and stand density (typically reported for forest plots as number of trees per hectare), representing the total sapwood area per unit of ground area. The cross sectional area indicies applied to the roots and stem xylem guarantee the conservation of water as it flows across soil, roots, and stem.

4.2.3 Root water uptake and transpiration

Eqs. (4.6) and (4.8) are coupled through the exchange of water between the soil and roots. The term S is modelled as a function of the difference between the water potential in the soil and the roots. This approach, introduced by Gardner (1960), was applied in several studies (Amenu and Kumar, 2007; Herkelrath et al., 1977; Mendel et al., 2002). Accordingly, S (s⁻¹) is expressed as

$$S(z,t) = k_{s,rad} f(\theta_s(z,t)) \cdot \frac{A_{ls}}{A_s}(z) \cdot \frac{r(z)}{\int_{z_{r_i}}^{z_{r_j}} r(z)dz} \cdot (\Phi_s(z,t) - \Phi_r(z,t)),$$
(4.11)

where $k_{s,rad}$ (m³s⁻¹m⁻²_{root}Pa⁻¹) is the soil-to-root radial conductance per unit of root surface area, $f(\theta_s)$ is a dimensionless reduction function due to soil moisture, r(z) the root mass distribution, with z_{r_i} and z_{r_j} (m) representing the elevation of the bottom and top of the roots, and A_{ls}/A_s (m²_{root}m⁻²_{ground}) is an index defining the lateral root surface area per unit of ground, representing the root surface area taking up water from the soil. The vertical profile of root mass distribution represents the percentage of roots contained in different soil layer. The product of these two terms in Eq. 4.11 provides the portion of roots contributing to the exchange of water between soil and roots; this changes with depth depending on how the roots are vertically distributed.

The water lost to the atmosphere is calculated using a transpiration function that depends on meteorological variables and limits the amount of water leaving the stomata as a function of the stem water potential. FETCH3 allows for the implementation of different transpiration functions, and a complete description of the transpiration formulation applied in this study is in Section 4.4. Accordingly, S_x/A_s (s⁻¹) reads

$$\frac{S_x}{A_s}(z,t) = T \cdot l(z), \tag{4.12}$$

where $T \pmod{s^{-1}}$ is the transpiration rate defined per unit of ground area, which is distributed along the canopy height via the leaf area density distribution $(l(z), m^2 m^{-2} m^{-1})$, which is the leaf area per unit of ground area per unit of height, which integrates vertically to the leaf area index (LAI). This effectively assumes that transpiration is proportional to leaf area throughout the depth of the canopy. We have found that in canopies where most leaves are concentrated near the upper layers, the results are not very sensitive to this simplification. More complex representations of the vertical distribution of transpiration through the canopy depth have been developed. Such vertically detailed canopy transpiration models assume, for example, that transpiration is vertically distributed proportionally to vertical light extinction through the depth of the canopy (Bohrer et al., 2009; Shaw and Schumann, 1992), or that transpiration rate is vertically distributed as a function that combines light attenuation and the vertical profiles of other physical radiative forcing, such as turbulence, wind speed, temperature, and humidity (Bonan et al., 2018; Drewry et al., 2010). Such transpiration models can be easily implemented in FETCH3 by replacing Eq. (4.12) with a more elaborate vertical redistribution scheme, provided that the vertical descriptions of the required parameters for leaf area density, light attenuation and other physical forcing are available for the simulated forest plot.

4.3 Numerical scheme

The water flow across the soil-plant-atmosphere continuum is lumped along the vertical direction. The domain of the model can be idealized as the combination of the vertical extent of the soil and the tree (root and stem) xylem, with exchange of water between the soil and the roots (Figure 4.2).



Figure 4.2: Representation of the model domain

Each of Eqs. (4.6), (4.8), and (4.10) can be written as

$$C\frac{\partial\Phi}{\partial t} = -\frac{\partial F}{\partial z} \pm S^*, \tag{4.13}$$

where C is a capacitance, Φ is the water potential, F is the flux, and S^* is a source or sink term accounting for either the exchange of water between soil and roots or the water loss due to transpiration for the stem xylem.

Eq. (4.13) is discretized using constant intervals, Δz , with values of Φ calculated at the nodes and fluxes, *F*, calculated between nodes. Eq. (4.13) is approximated using a fully implicit backward Euler method, and its discretized form for a generic node *i* reads

$$C_i^{n+1} \frac{\Phi_i^{n+1} - \Phi_i^n}{\Delta t} = -\frac{F_{i+1/2}^{n+1} - F_{i-1/2}^{n+1}}{\Delta z} \pm S_i^{*,n+1},$$
(4.14)

where n and n+1 indicate values of the variables at two consecutive times, with $\Delta t = t^{n+1} - t^n$. The fluxes can be then expressed as

$$F_{i+1/2}^{n+1} = -K_{i+1/2}^{n+1} \frac{\Phi_{i+1}^{n+1} - \Phi_i^{n+1}}{\Delta z} - K_{i+1/2}^{n+1} \rho g$$
(4.15)

$$F_{i-1/2}^{n+1} = -K_{i-1/2}^{n+1} \frac{\Phi_i^{n+1} - \Phi_{i-1}^{n+1}}{\Delta z} - K_{i-1/2}^{n+1} \rho g,$$
(4.16)

where K is the hydraulic conductance (including the ratio of the areas appearing in Eqs. 4.8 and 4.10).

In the following, a full detailed description of the numerical approximation of the system of equations and the implementation of initial and boundary conditions are presented.

4.3.1 Discretization

Eqs. (4.6), (4.8), and (4.10) are approximated as in Eq. (4.13), and are then combined and solved at the same time with the standard fully implicit Picard method following the scheme in Celia et al. (1990).

With m denoting the number of the Picard iteration, and the subscript i denoting a generic node, Eq. (4.14) reads

$$C_{i}^{n+1,m} \frac{(\Phi_{i}^{n+1,m+1} - \Phi_{i}^{n})}{\Delta t} = +K_{i+1/2}^{n+1,m} \frac{(\Phi_{i+1}^{n+1,m+1} - \Phi_{i}^{n+1,m+1})}{(\Delta z)^{2}} - K_{i-1/2}^{n+1,m} \frac{(\Phi_{i}^{n+1,m+1} - \Phi_{i-1}^{n+1,m+1})}{(\Delta z)^{2}} + \frac{K_{i+1/2}^{n+1,m}}{\Delta z} \rho g - \frac{K_{i-1/2}^{n+1,m}}{\Delta z} \rho g \pm S_{i}^{*,n+1}, \quad (4.17)$$

with the hydraulic conductance calculated in the middle of two neighbouring nodes as

$$K_{i-1/2}^{n+1,m} = \frac{1}{2} (K_{i-1}^{n+1,m} + K_i^{n+1,m})$$
(4.18)

$$K_{i+1/2}^{n+1,m} = \frac{1}{2} (K_i^{n+1,m} + K_{i+1}^{n+1,m}).$$
(4.19)

The terms in Eq. (4.17) can be rearranged as

$$\frac{C_{i}^{n+1,m}}{\Delta t} \left(\Phi_{i}^{n+1,m+1} - \Phi_{i}^{n+1,m} \right) + \frac{C_{i}^{n+1,m}}{\Delta t} \left(\Phi_{i}^{n+1,m} - \Phi_{i}^{n} \right) = + \frac{K_{i+1/2}^{n+1,m}}{\Delta z^{2}} \left(\Phi_{i+1}^{n+1,m+1} - \Phi_{i+1}^{n+1,m} \right) \\
+ \frac{K_{i+1/2}^{n+1,m}}{\Delta z^{2}} \left(\Phi_{i+1}^{n+1,m} \right) - \frac{K_{i+1/2}^{n+1,m}}{\Delta z^{2}} \left(\Phi_{i}^{n+1,m+1} - \Phi_{i}^{n+1,m} \right) - \frac{K_{i+1/2}^{n+1,m}}{\Delta z^{2}} \left(\Phi_{i}^{n+1,m} \right) + \frac{K_{i+1/2}^{n+1,m}}{\Delta z} \rho g \\
- \frac{K_{i-1/2}^{n+1,m}}{\Delta z^{2}} \left(\Phi_{i}^{n+1,m+1} - \Phi_{i}^{n+1,m} \right) - \frac{K_{i-1/2}^{n+1,m}}{\Delta z^{2}} \left(\Phi_{i-1}^{n+1,m} \right) + \frac{K_{i-1/2}^{n+1,m}}{\Delta z^{2}} \left(\Phi_{i-1}^{n+1,m+1} - \Phi_{i-1}^{n+1,m} \right) \\
+ \frac{K_{i-1/2}^{n+1,m}}{\Delta z^{2}} \left(\Phi_{i-1}^{n+1,m} \right) - \frac{K_{i-1/2}^{n+1,m}}{\Delta z^{2}} \left(\Phi_{i-1}^{n+1,m} \right) - \frac{K_{i-1/2}^{n+1,m}}{\Delta z} \rho g \pm S_{i}^{*,n+1}, \quad (4.20)$$

which, defining the increment $\delta \Phi^{n+1,m} = \Phi^{n+1,m+1} - \Phi^{n+1,m}$, yields

$$\frac{C_{i}^{n+1,m}}{\Delta t} \left(\delta \Phi_{i}^{n+1,m} \right) + \frac{C_{i}^{n+1,m}}{\Delta t} \left(\Phi_{i}^{n+1,m} - \Phi_{i}^{n} \right) = + \frac{K_{i+1/2}^{n+1,m}}{\Delta z^{2}} \left(\delta \Phi_{i+1}^{n+1,m} \right) + \frac{K_{i+1/2}^{n+1,m}}{\Delta z^{2}} \left(\Phi_{i+1}^{n+1,m} \right) \\
- \frac{K_{i+1/2}^{n+1,m}}{\Delta z^{2}} \left(\delta \Phi_{i}^{n+1,m} \right) - \frac{K_{i+1/2}^{n+1,m}}{\Delta z^{2}} \left(\Phi_{i}^{n+1,m} \right) + \frac{K_{i+1/2}^{n+1,m}}{\Delta z} \rho g - \frac{K_{i-1/2}^{n+1,m}}{\Delta z^{2}} \left(\delta \Phi_{i}^{n+1,m} \right) \\
- \frac{K_{i-1/2}^{n+1,m}}{\Delta z^{2}} \left(\Phi_{i}^{n+1,m} \right) + \frac{K_{i-1/2}^{n+1,m}}{\Delta z^{2}} \left(\delta \Phi_{i-1}^{n+1,m} \right) + \frac{K_{i-1/2}^{n+1,m}}{\Delta z^{2}} \left(\Phi_{i-1}^{n+1,m} \right) - \frac{K_{i-1/2}^{n+1,m}}{\Delta z} \rho g \pm S_{i}^{*,n+1}.$$
(4.21)

Reorganizing Eq.(4.21), one obtains

$$\left(\frac{-K_{i-1/2}^{n+1,m}}{\Delta z^2}\right)\delta\Phi_{i-1}^{n+1,m} + \left(\frac{C_i^{n+1,m}}{\Delta t} + \frac{K_{i+1/2}^{n+1,m}}{\Delta z^2} + \frac{K_{i-1/2}^{n+1,m}}{\Delta z^2}\right)\delta\Phi_i^{n+1,m} + \left(\frac{-K_{i+1/2}^{n+1,m}}{\Delta z^2}\right)\delta\Phi_{i+1}^{n+1,m} = \frac{1}{\Delta z^2}\left[K_{i+1/2}^{n+1,m}\left(\Phi_{i+1}^{n+1,m} - \Phi_i^{n+1,m}\right) - K_{i-1/2}^{n+1,m}\left(\Phi_i^{n+1,m} - \Phi_{i-1}^{n+1,m}\right)\right] + \frac{K_{i+1/2}^{n+1,m}}{\Delta z}\rho g - \frac{K_{i-1/2}^{n+1,m}}{\Delta z}\rho g - \frac{C_i^{n+1,m}}{\Delta t}\left(\Phi_i^{n+1,m} - \Phi_i^{n}\right) \pm S_i^{*,n+1}.$$
(4.22)

For the soil, $S_i^{*,n+1}$ is a sink of water (i.e., the sign '-' applies to Eq. (4.22))

$$S_i^{*,n+1} = k_{e,rad_i}(z,t) \cdot f(\theta_i^{n+1,m+1})(\Phi_{s_i}^{n+1,m+1} - \Phi_{r_j}^{n+1,m+1}),$$
(4.23)

where the subscripts s and r refer to soil and roots, and j is a node of the roots corresponding to the same elevation z as i for the soil (Figure 4.2). k_{e,rad_i} represents the effective radial conductance.

Introducing the increments $\delta \Phi^{n+1,m}$ for the water potentials, the expression for $S^{*,n+1}$ in the soil
(with the sign '-' in Eq. 4.21) and root xylem (with the sign '+' in Eq. 4.21) is

$$S_i^{*,n+1}(z,t) = k_{e,rad_i} f(\theta_i^{n+1,m+1}) (\delta \Phi_{s_i}^{n+1,m} - \delta \Phi_{r_j}^{n+1,m} - \Phi_{s_i}^{n+1,m} + \Phi_{r_j}^{n+1,m}).$$
(4.24)

Reorganizing the equation in the same form of Eq. (4.22), and substituting the expression for the sink term (Eq. 4.24), the numerical approximation for the soil nodes becomes:

$$-\frac{K_{s_{i}-1/2}^{n+1,m}}{\Delta z^{2}}\delta\Phi_{s_{i}-1}^{n+1,m} + \left[\frac{C_{s_{i}}^{n+1,m}}{\Delta t} + \frac{K_{s_{i}+1/2}^{n+1,m}}{\Delta z^{2}} + \frac{K_{s_{i}-1/2}^{n+1,m}}{\Delta z^{2}} + k_{e,rad_{i}}f(\theta_{i}^{n+1,m+1})\right]\delta\Phi_{s_{i}}^{n+1,m} - \frac{K_{s_{i}+1/2}^{n+1,m}}{\Delta z^{2}}\delta\Phi_{s_{i}+1}^{n+1,m} - k_{e,rad_{i}}f(\theta_{i}^{n+1,m+1})\delta\Phi_{r_{j}} = \frac{1}{\Delta z^{2}}\left[K_{s_{i}+1/2}^{n+1,m}\left(\Phi_{s_{i}+1}^{n+1,m} - \Phi_{s_{i}}^{n+1,m}\right) - K_{s_{i}-1/2}^{n+1,m}\left(\Phi_{s_{i}}^{n+1,m} - \Phi_{s_{i}-1}^{n+1,m}\right)\right] + \frac{K_{s_{i}+1/2}^{n+1,m}}{\Delta z}\rho g - \frac{K_{s_{i}-1/2}^{n+1,m}}{\Delta z}\rho g - \frac{C_{s_{i}}^{n+1,m}}{\Delta t}\left(\Phi_{s_{i}}^{n+1,m} - \Phi_{s_{i}}^{n}\right) - k_{e,rad_{i}}f(\theta_{i}^{n+1,m+1})\left(\Phi_{s_{i}}^{n+1,m} - \Phi_{r_{j}}^{n+1,m}\right).$$
 (4.25)

Likewise, the numerical approximation for the nodes of the root xylem becomes

$$-\frac{K_{r_{i}-1/2}^{n+1,m}}{\Delta z^{2}}\delta\Phi_{r_{i}-1}^{n+1,m} + \left[\frac{C_{r_{i}}^{n+1,m}}{\Delta t} + \frac{K_{r_{i}+1/2}^{n+1,m}}{\Delta z^{2}} + \frac{K_{r_{i}-1/2}^{n+1,m}}{\Delta z^{2}} + k_{e,rad_{i}}f(\theta_{i}^{n+1,m+1})\right]\delta\Phi_{r_{i}}^{n+1,m} - \frac{K_{r_{i}+1/2}^{n+1,m}}{\Delta z^{2}}\delta\Phi_{r_{i}+1}^{n+1,m} - k_{e,rad_{i}}f(\theta_{i}^{n+1,m+1})\delta\Phi_{s_{j}} = \frac{1}{\Delta z^{2}}\left[K_{r_{i}+1/2}^{n+1,m}\left(\Phi_{r_{i}+1}^{n+1,m} - \Phi_{r_{i}}^{n+1,m}\right) - K_{r_{i}-1/2}^{n+1,m}\left(\Phi_{r_{i}}^{n+1,m} - \Phi_{r_{i}-1}^{n+1,m}\right)\right] + \frac{K_{r_{i}+1/2}^{n+1,m}}{\Delta z}\rho g - \frac{K_{r_{i}-1/2}^{n+1,m}}{\Delta z}\rho g - \frac{C_{r_{i}}^{n+1,m}}{\Delta t}\left(\Phi_{r_{i}}^{n+1,m} - \Phi_{r_{i}}^{n}\right) + k_{e,rad_{i}}f(\theta_{i}^{n+1,m+1})\left(\Phi_{s_{i}}^{n+1,m} - \Phi_{r_{j}}^{n+1,m}\right).$$
 (4.26)

A similar approximation can be written for the stem xylem, where the sink term is simpler than for the soil and root xylem, because it does not include interactions between nodes at different elevations.

Considering, for example, that both soil and roots have the same depth (i.e, roots are present from the bottom to the top of the soil column), the system of partial differential equations can be approximated by a system of algebraic equations for each iteration within each time step. In matrix format, this system of equations can be written as

$\int b_1$	c_1	0	0	 $K_{rad,1}$	0	0							١	١	$\left(\delta_1^{n+1,m}\right)$	١	$\left(d_{1}^{n+1} \right)$	۱
a_2	b_2	c_2	0	 0	$K_{rad,2}$	0									$\delta_2^{n+1,m}$		d_2^{n+1}	
0	a_3	b_3	c_3	 0	0	$K_{rad,3}$									$\delta_3^{n+1,m}$		d_3^{n+1}	
$K_{rad,1}$	0			0	b_r	c_r	0								$\delta_r^{n+1,m}$		d_r^{n+1}	
0	$K_{rad,2}$	0		0	a_{r+1}	b_{r+1}	c_{r+1}	0						·	$\delta_{r+1}^{n+1,m}$	=	d_{r+1}^{n+1}	,
0	0	$K_{rad,3}$	0	 0	0	a_{r+2}	b_{r+2}	c_{r+2}	0						$\delta_{r+2}^{n+1,m}$		d_{r+2}^{n+1}	
										 a_{k-2}	b_{k-2}	c_{k-2}	0		$\delta_{k-2}^{n+1,m}$		d_{k-2}^{n+1}	
										 0	a_{k-1}	b_{k-1}	c_{k-1}		$\delta_{k-1}^{n+1,m}$		d_{k-1}^{n+1}	
\										 0	0	a_k	b_k	/	$\left\{ \delta_{k}^{n+1,m} \right\}$	/	$\begin{pmatrix} d_k^{n+1} \end{pmatrix}$	/

where the subscripts r and k denote the bottom of the roots and top of the canopy, respectively, and the node 1 represents the bottom of the soil (Figure 4.2). The system of algebraic equations is solved to find the values of $\delta \Phi_i^{n+1}$; this is then added to $\Phi_i^{n,m}$ to calculate $\Phi_i^{n+1,m+1}$, which is then used to calculate the new values of the coefficients for the calculation of a new value of $\delta \Phi_i$. These iterations proceed until the difference of two successive calculated water potentials in each node approaches a predefined tolerance, δ , such that

$$\left|\Phi^{n+1,m+1} - \Phi^{n+1,m}\right| \le \delta.$$
(4.27)

The matrix coefficients for a generic node within the soil read

$$a_i = -\frac{K_{i-1/2}^{n+1,m}}{\Delta z^2}$$
(4.28)

$$b_{i} = \left(\frac{C_{i}^{n+1,m}}{\Delta t} + \frac{K_{i+1/2}^{n+1,m}}{\Delta z^{2}} + \frac{K_{i-1/2}^{n+1,m}}{\Delta z^{2}} + k_{e,rad_{i}}f(\theta_{i}^{n+1,m+1})\right)$$
(4.29)

$$c_i = -\frac{K_{i+1/2}^{n+1,m}}{\Delta z^2} \tag{4.30}$$

$$K_{rad,i} = -k_{e,rad_i} f(\theta_i^{n+1,m+1})$$
(4.31)

$$d_{i} = \frac{1}{\Delta z^{2}} \left[K_{i+1/2}^{n+1,m} (\Phi_{i+1}^{n+1,m} - \Phi_{i}^{n+1,m}) - K_{i-1/2}^{n+1,m} (\Phi_{i}^{n+1,m} - \Phi_{i-1}^{n+1,m}) \right]$$
(4.32)

$$+\rho g \frac{(K_{i+1/2}^{n+1,m} - K_{i-1/2}^{n+1,m})}{\Delta z} - \frac{C_i^{n+1,m}}{\Delta t} (\Phi_i^{n+1,m} - \Phi_i^n) - k_{e,rad_i} f(\theta_i^{n+1,m+1}) \left(\Phi_{s_i}^{n+1,m} - \Phi_{r_j}^{n+1,m}\right).$$
(4.33)

The coefficients for a generic node associated with the roots are

$$a_i = -\frac{K_{i-1/2}^{n+1,m}}{\Delta z^2}$$
(4.34)

$$b_{i} = \left(\frac{C_{i}^{n+1,m}}{\Delta t} + \frac{K_{i+1/2}^{n+1,m}}{\Delta z^{2}} + \frac{K_{i-1/2}^{n+1,m}}{\Delta z^{2}} + k_{e,rad_{i}}f(\theta_{i}^{n+1,m+1})\right)$$
(4.35)

$$c_i = -\frac{K_{i+1/2}^{n+1,m}}{\Delta z^2}$$
(4.36)

$$K_{rad,i} = -k_{e,rad_i} f(\theta_i^{n+1,m+1})$$
 (4.37)

$$d_{i} = \frac{1}{\Delta z^{2}} \left[K_{i+1/2}^{n+1,m} (\Phi_{i+1}^{n+1,m} - \Phi_{i}^{n+1,m}) - K_{i-1/2}^{n+1,m} (\Phi_{i}^{n+1,m} - \Phi_{i-1}^{n+1,m}) \right]$$
(4.38)

$$+\rho g \frac{(K_{i+1/2}^{n+1,m} - K_{i-1/2}^{n+1,m})}{\Delta z} - \frac{C_i^{n+1,m}}{\Delta t} (\Phi_i^{n+1,m} - \Phi_i^n) + k_{e,rad_i} f(\theta_i^{n+1,m+1}) \left(\Phi_{s_i}^{n+1,m} - \Phi_{r_j}^{n+1,m}\right).$$
(4.39)

The coefficients for a generic node of the stem read

$$a_i = -\frac{K_{i-1/2}^{n+1,m}}{\Delta z^2} \tag{4.40}$$

$$b_{i} = \left(\frac{C_{i}^{n+1,m}}{\Delta t} + \frac{K_{i+1/2}^{n+1,m}}{\Delta z^{2}} + \frac{K_{i-1/2}^{n+1,m}}{\Delta z^{2}}\right)$$
(4.41)

$$= -\frac{K_{i+1/2}^{n+1,m}}{\Delta z^2}$$
(4.42)

$$d_{i} = \frac{1}{\Delta z^{2}} \left[K_{i+1/2}^{n+1,m} (\Phi_{i+1}^{n+1,m} - \Phi_{i}^{n+1,m}) - K_{i-1/2}^{n+1,m} (\Phi_{i}^{n+1,m} - \Phi_{i-1}^{n+1,m}) \right]$$
(4.43)

$$+\rho g \frac{(K_{i+1/2}^{n+1,m} - K_{i-1/2}^{n+1,m})}{\Delta z} - \frac{C_i^{n+1,m}}{\Delta t} (\Phi_i^{n+1,m} - \Phi_i^n) - S_{x,i}^{n+1}/A_{s,i},$$
(4.44)

with details on the expression of S_x provided in Section 4.4.

4.3.2 Initial and boundary conditions

The initial condition needs to be a known function $\Phi_0 = \Phi(z, 0)$ across the soil, roots and stem. A common choice is to assume that soil and both root and stem xylem are in hydrostatic conditions (i.e., $\Phi_0 = -\rho gz$), without water fluxes occurring across the system.

 c_i

Boundary conditions are required at the bottom of the roots, at the top of the stem, and at the bottom and top of the soil. No-flux boundary conditions are the default conditions for the bottom of the roots and top of the stem. No flux boundary conditions can be imposed at the bottom of the soil and a specific flux can be imposed at the top of the soil as infiltration. Conditions at the

bottom of the soil can be also provided as a given value of water potential or as free drainage.

No flux boundary conditions can be specified by imposing the flux at the corresponding nodes to be zero. For example, in the case of the bottom of the soil, for i = 1, a no flux condition is obtained by imposing $F_{1-1/2}^{n+1} = 0$ in Eq. (4.14).

Accordingly, the terms in the matrix summarizing the system of equation for node 1 are

$$\begin{split} b_1 &= \frac{C_1^{n+1,m}}{\Delta t} + \frac{K_{1+1/2}^{n+1,m}}{\Delta z^2} + k_{e,rad_1} f(\theta_i^{n+1,m+1}), \\ &c_1 &= -\frac{K_{1+1/2}^{n+1,m}}{\Delta z^2}, \\ K_{rad,1} &= -k_{e,rad_1} f(\theta_i^{n+1,m+1}) \\ d_1 &= \frac{1}{\Delta z^2} \left[K_{1+1/2}^{n+1,m} (\Phi_2^{n+1,m} - \Phi_1^{n+1,m}) \right] + \rho g \frac{K_{1+1/2}^{n+1,m}}{\Delta z} \\ &- \frac{C_1^{n+1,m}}{\Delta t} (\Phi_1^{n+1,m} - \Phi_1^n) - k_{e,rad_1} f(\theta_i^{n+1,m+1}) \left(\Phi_1^{n+1,m} - \Phi_{r_j}^{n+1,m} \right). \end{split}$$

The conditions at the bottom of the roots and top of the stem can be obtained in a similar way. In the case of a specified water potential boundary condition (Dirichlet), the potential at the bottom of the soil is a series of known values, such that $\phi(1,t) = \Phi_{b_i}$. This leads to the following expression for $\Phi_2^{n+1,m+1}$:

$$\begin{pmatrix} -K_{2-1/2}^{n+1,m} \\ \overline{\Delta z^2} \end{pmatrix} \delta \Phi_{b_i} + \begin{pmatrix} C_2^{n+1,m} \\ \overline{\Delta t} + \frac{K_{2+1/2}^{n+1,m}}{\Delta z^2} + \frac{K_{2-1/2}^{n+1,m}}{\Delta z^2} + k_{e,rad_2} f(\theta_2^{n+1,m+1}) \end{pmatrix} \delta \Phi_2^{n+1,m} + \\ \begin{pmatrix} -K_{2+1/2}^{n+1,m} \\ \overline{\Delta z^2} \end{pmatrix} \delta \Phi_3^{n+1,m} - k_{e,rad_2} f(\theta_2^{n+1,m+1}) \delta \Phi_{s_i} = \\ \frac{1}{\Delta z^2} \left[K_{2+1/2}^{n+1,m} \left(\Phi_3^{n+1,m} - \Phi_2^{n+1,m} \right) - K_{2-1/2}^{n+1,m} \left(\Phi_2^{n+1,m} - \Phi_{b_i}^{n+1,m} \right) \right] + \frac{K_{2+1/2}^{n+1,m}}{\Delta z} \rho g \\ - \frac{K_{2-1/2}^{n+1,m}}{\Delta z} \rho g - \frac{C_2^{n+1,m}}{\Delta t} \left(\Phi_2^{n+1,m} - \Phi_2^n \right) - k_{e,rad_2} f(\theta_2^{n+1,m+1}) (\Phi_{s_i}^{n+1,m} - \Phi_{r_j}^{n+1,m})$$
(4.45)

Reorganizing in the form of the matrix yields

$$a_2 = \frac{-(K_{2-1/2}^{n+1,m})}{\Delta z^2} \tag{4.46}$$

$$b_2 = \left(\frac{C_2^{n+1,m}}{\Delta t} + \frac{K_{2+1/2}^{n+1,m}}{\Delta z^2} + \frac{K_{2-1/2}^{n+1,m}}{\Delta z^2} + k_{e,rad_2}f(\theta_2^{n+1,m+1})\right)$$
(4.47)

$$c_2 = \frac{-(K_{2+1/2}^{n+1,m})}{\Delta z^2} \tag{4.48}$$

$$K_{rad,2} = -k_{e,rad_2} f(\theta_2^{n+1,m+1})$$
(4.49)

$$d_2 = \frac{1}{\Delta z^2} \left[K_{2+1/2}^{n+1,m} (\Phi_3^{n+1,m} - \Phi_2^{n+1,m}) - K_{2-1/2}^{n+1,m} (\Phi_2^{n+1,m} - \Phi_{b_i}^{n+1,m}) \right]$$
(4.50)

$$+\rho g \frac{(K_{2+1/2}^{n+1,m} - K_{2-1/2}^{n+1,m})}{\Delta z} - \frac{C_2^{n+1,m}}{\Delta t} (\Phi_2^{n+1,m} - \Phi_2^n) - k_{e,rad_2} f(\theta_2^{n+1,m+1}) \left(\Phi_{s_i}^{n+1,m} - \Phi_{r_j}^{n+1,m}\right)$$
(4.51)

In the case of soil profile with deep groundwater levels, the water flux at the bottom of the soil is only due to gravity; therefore, the pressure gradient is equal to zero:

$$F_{1-1/2}^{n+1} = -K_{1-1/2}^{n+1} \left(0 + \rho g\right), \tag{4.52}$$

$$C_1^{n+1} \frac{\Phi_1^{n+1} - \Phi_1^n}{\Delta t} = -\frac{F_{1+1/2}^{n+1} - (-K_{1-1/2}^{n+1}\rho g)}{\Delta z} - S(\Phi_1^n).$$
(4.53)

Reorganizing in matrix format, one obtains

$$b_{1} = \frac{C_{1}^{n+1,m}}{\Delta t} + \frac{K_{1+1/2}^{n+1,m}}{\Delta z^{2}} + k_{e,rad_{1}}f(\theta_{1}^{n+1,m+1}),$$

$$c_{1} = -\frac{K_{1+1/2}^{n+1,m}}{\Delta z^{2}},$$

$$K_{rad,1} = -k_{e,rad_{1}}f(\theta_{1}^{n+1,m+1})$$

$$d_{1} = \frac{1}{\Delta z^{2}} \left[K_{1+1/2}^{n+1,m}(\Phi_{2}^{n+1,m} - \Phi_{1}^{n+1,m}) \right] + \rho g \frac{K_{1+1/2}^{n+1,m}}{\Delta z} - \frac{C_{1}^{n+1,m}}{\Delta t}(\Phi_{1}^{n+1,m} - \Phi_{1}^{n})$$

$$-\frac{K_{1-1/2}^{n+1,m}\rho g}{\Delta z} - k_{e,rad_{1}}f(\theta_{1}^{n+1,m+1}) \left(\Phi_{1}^{n+1,m} - \Phi_{r_{j}}^{n+1,m}\right).$$

For the top of the soil column, considering inputs from infiltration, the flux at the node i = s is known as $F_{s+1/2}^{n+1} = q_{inf}$, resulting in

$$C_s \frac{\Phi_s^{n+1} - \Phi_s^n}{\Delta t} = -\frac{q_{inf} - F_{s-1/2}^{n+1}}{\Delta z} - S,$$
(4.54)

$$q_{inf}^{n+1,m+1} = max\left(-q, -\left(\theta_{sat} - \theta_s^{n+1,m+1}\right) \cdot \left(\frac{\Delta z}{\Delta t}\right)\right),\tag{4.55}$$

where q is the precipitation rate.

Reorganizing in matrix format leads to

$$\begin{split} a_s &= -\frac{K_{s-1/2}^{n+1,m}}{\Delta z^2}, \\ b_s &= \frac{C_s^{n+1,m}}{\Delta t} + \frac{K_{s-1/2}^{n+1,m}}{\Delta z^2} + k_{e,rad_s} f(\theta_s^{n+1,m+1}), \\ K_{rad_s} &= -k_{e,rad_s} f(\theta_s^{n+1,m+1}), \\ d_s &= \frac{1}{\Delta z^2} \left[-K_{s-1/2}^{n+1,m} (\Phi_s^{n+1,m} - \Phi_{s-1}^{n+1,m}) \right] - \rho g \frac{K_{s-1/2}^{n+1,m}}{\Delta z} - \frac{C_s^{n+1,m}}{\Delta t} (\Phi_s^{n+1,m} - \Phi_s^n) \\ &- k_{e,rad_s} f(\theta_s^{n+1,m+1}) \left(\Phi_s^{n+1,m} - \Phi_{r_j}^{n+1,m} \right) + \frac{q_{inf}}{\Delta z}. \end{split}$$

4.4 Penman-Monteith transpiration

Transpiration is calculated in a subroutine and can thus be defined according to the formulation assigned by the user. Here, we present how transpiration was calculated in Section 4.2.3, following the formulation implemented in (Verma et al., 2014).

The water that the trees lose to the atmosphere via transpiration S_x/A_s is calculated as

$$\frac{S_x}{A_s} = T \cdot l(z), \tag{4.56}$$

where *T* is the transpiration rate per unit of ground area (m s⁻¹). Transpiration is distributed throughout the canopy height using the leaf area density (LAD) (l(z), m² m⁻² m⁻¹), which is the leaf area index (LAI) distributed along the canopy height *z*.

Transpiration is calculated through the Penman-Monteith formulation (Allen et al., 1998; Verma et al., 2014):

$$T = \left[\frac{Q_n \Delta + C_p D g_a}{\lambda [\Delta g_c + \gamma (g_c + g_a)]}\right] g_c,$$
(4.57)

where Q_n (W m⁻²) is the net radiation, Δ (kg m⁻¹s⁻²K⁻¹) is the slope of the saturation vapor

pressure curve for a given temperature, C_p (kg m⁻¹s⁻²K⁻¹) is the specific heat of air, D is the vapour pressure deficit (VPD) (Pa), λ (kg m⁻¹s⁻²) is the latent heat of vaporisation, g_a (m s⁻¹) is the aerodynamic conductance, γ (kg m s⁻²K⁻¹) is the psychrometric constant, and g_c (m s⁻¹) is the canopy conductance. In this study, Q_n is assumed to be 60% of incoming solar radiation; this is different from Verma et al. (2014), where this term was calculated as 70% of solar radiation . Δ is calculated as

$$\Delta = \left[\frac{4098}{(T_a - 35.85)^2}\right] e_{sat},\tag{4.58}$$

where e_{sat} (Pa) is the saturation vapour pressure for a given temperature of the atmosphere (T_a) , given as

$$e_{sat} = 611 \exp\left[\frac{17.27(T_a - 273.15)}{T_a - 35.85}\right].$$
(4.59)

The canopy conductance, g_c , is calculated as

$$g_c = \left[\frac{g_s g_b}{g_s + g_b}\right] LAI,\tag{4.60}$$

where LAI is the leaf area index, g_b (m s⁻¹) is the leaf boundary layer conductance per m² of leaf area and g_s (m s⁻¹) is the stomatal conductance, which depends on both plant physiology and environmental factors. In the present model, g_s is modeled as (Jarvis, 1976),

$$g_s = g_{smax} \cdot f(S_{in}) \cdot f(T_a) \cdot f(D) \cdot f(\Phi_x), \tag{4.61}$$

where g_{smax} (m s⁻¹) is the maximum stomatal conductance, and $f(S_{in})$, $f(T_a)$, f(D), and $f(\Phi_x)$ are empirical functions [-], representing the behavior of g_s according to variations in solar radiation, temperature, VPD, and leaf water potential, respectively. These functions vary between 0 and 1, and are calculated as

$$f(S_{in}) = 1 - \exp(-k_r S_{in}), \tag{4.62}$$

$$f(T_a) = 1 - k_t (T_a - T_{opt})^2,$$
(4.63)

$$f(D) = \frac{1}{(1+Dk_d)},$$
(4.64)

$$f(\Phi_x) = \left[1 + \left(\frac{\Phi_{xleaf}}{\Phi_{x50}}\right)^{n_l}\right]^{-1},\tag{4.65}$$

where k_r , k_t , k_d and n_l are empirical constants, T_{opt} is the air temperature at which $f(T_a)$ is 1, and Φ_{x50} is the leaf water potential at 50% loss of conductivity. $f(\Phi_x)$ represents the inverse polynomial expression commonly used to express the effect of water potential on xylem conductance (Manzoni et al., 2013a).

Since nighttime transpiration is not always zero under low or absent solar radiation (ie., during heat waves), a different approach was used to model night time transpiration, as in

$$E_n = E_{max} \cdot f(T_a) \cdot f(D) \cdot f(\Phi_x), \tag{4.66}$$

where E_{max} (m s⁻¹) is the maximum night time transpiration, which was estimated in Verma et al. (2014).

4.5 Soil water retention relationships

The relationships between K_s , Φ_s and θ_s are modeled according to van Genuchten (1980) as

$$K_s = \frac{k_{s,sat}}{\rho g} \Theta^{1/2} \left[1 - (1 - \Theta^{1/n})^m \right]^2,$$
(4.67)

$$C_s = \frac{d\theta_s}{d\Phi_s} = \frac{d\theta_s}{\rho g dh_s} = \frac{-\alpha m (\theta_{sat} - \theta_{res})}{\rho g (1-m)} \Theta^{1/m} (1 - \Theta^{1/m})^m,$$
(4.68)

where,

$$\Theta(z,t) = \frac{\theta - \theta_{res}}{\theta_{sat} - \theta_{res}},$$
(4.69)

$$\theta = \theta_{res} + \frac{(\theta_{sat} - \theta_{res})}{[1 + (\alpha|h|)^n]^m},$$
(4.70)

$$m = 1 - \frac{1}{n},$$
 (4.71)

$$0 < m < 1,$$
 (4.72)

where $K_s \ (m^2 s^{-1} Pa^{-1})$ is the effective soil hydraulic conductivity, $k_{s,sat} \ (m s^{-1})$ is the soil hydraulic conductivity under saturated soil conditions, θ_{sat} and θ_{res} indicate the saturated and residual values of the soil volumetric water content, and $C_s \ (Pa^{-1})$ is the soil capacitance.

4.6 Hydraulic conductances

4.6.1 Root radial conductance

The exchange of water between soil and roots depends of the amount of roots contained in different soil layers. The fraction of roots at different soil depths is modeled following Vrugt et al. (2001), with the function

$$r(z) = \left(1 - \frac{z}{z_{r_j} - z_{r_i}}\right) \exp\left(\frac{-q_z}{z_{r_j} - z_{r_i}} z\right) \quad z_{r_i} \le z \le z_{r_j},\tag{4.73}$$

where z is the vertical coordinate (positive upwards), z_{r_j} is the elevation of the top of the roots, and z_{r_i} is the elevation of the bottom of the roots, with $z_{r_j} - z_{r_i}$ the root depth.

Using Eq.(4.73), the root water uptake is written as

$$S(z,t) = k_{s,rad} f(\theta_s(z,t)) \cdot \frac{A_{ls}}{A_s}(z) \cdot \frac{r(z)}{\int_{z_{r_i}}^{z_{r_j}} r(z)dz} \cdot (\Phi_s(z,t) - \Phi_r(z,t)),$$
(4.74)

where $f(\theta)$ [-] is a water stress reduction function, Φ_s , and Φ_r , are water potentials (Pa) for the soil and roots, respectively, $k_{s,rad}$ (m³s⁻¹m⁻²_{root}Pa⁻¹) is the soil-to-root radial conductance per unit of root surface area, and A_{ls}/A_s (m²_{root}m⁻²_{ground}) is an index defining the lateral root surface area per unit of ground, representing the root surface area taking up water from the soil. The $f(\theta)$ function is written as,

$$\begin{pmatrix}
0 & \theta \le \theta_1, \\
(\theta - \theta_1)
\end{pmatrix}$$
(4.75)

$$f(\theta) = \begin{cases} \frac{(\theta - \theta_1)}{(\theta_2 - \theta_1)} & \theta_1 < \theta \le \theta_2, \end{cases}$$
(4.76)

$$\begin{pmatrix} 1 & \theta > \theta_2, \\ (4.77) \end{pmatrix}$$

where θ_1 and θ_2 are the soil water content below which root water uptake is ceased and the soil water content below which root water uptake start decreasing, respectively (Feddes et al., 1976). The current version of FETCH3 does not consider water stress by low oxygen conditions (anoxia).

4.6.2 Axial conductances

The effective axial conductance of the roots, K_r (m²s⁻¹Pa⁻¹), can be defined as

$$K_r = \frac{k_r}{\rho g} \frac{A_r}{A_s},\tag{4.78}$$

with,

$$k_r(z,t) = k_{sax} \cdot \left(1 - \frac{1}{1 + \exp(a_p(\Phi_r - b_p))}\right),$$
(4.79)

where k_{sax} (m s⁻¹) is the specific axial conductivity for the root system, and a_p (Pa⁻¹) and b_p (Pa) are parameters for the root xylem cavitation curve, which describes the vulnerability of xylem to cavitation.

The effective hydraulic conductance of the axial stem xylem, K_x (m² s⁻¹ Pa⁻¹), is defined similarly as Eq. (4.78), as

$$K_x = \frac{k_x}{\rho g} \frac{A_x}{A_s},\tag{4.80}$$

with

$$k_x(\Phi_x(z,t)) = k_{max} \cdot \left(1 - \frac{1}{1 + \exp(a_p(\Phi_x - b_p))}\right),$$
(4.81)

where k_{max} (m s⁻¹) is the maximum conductivity of saturated stem xylem, and a_p (Pa⁻¹) and b_p (Pa) are the shape parameters of the cavitation curve.

Chapter 5

FETCH3 model applications

This chapter contains large parts of the article "Tree hydrodynamic modelling of the soil plant atmosphere continuum using FETCH3" published in Geoscientific Model Development.

5.1 Aim

Three applications were used to i) test the correctness of the numerical scheme against analytical solutions, ii) compare results to a published case study, iii) show the implementation of a leaf area density profile and a xylem capacitance function dependent on the xylem water potential, and iv) calibrate the model against measured sap flow data. Figure 5.1 summarizes the model applications described in this chapter.



Figure 5.1: Summary of model applications described in the chapter.

5.2 Testing against analytical solutions

The numerical scheme was tested against three simplified cases that permit the derivation of solutions in closed form. Because of the nonlinear nature of the Richardson-Richards equation, only a few exact solutions are available, particularly when including sink and source terms (Broadbridge et al., 2017). An exact solution of the combined soil-to-air system (Eqs. 4.6 - 4.10) is thus too challenging to be derived. Therefore, the numerical scheme was tested against one of the equations. Eq. (4.10) was selected for this exercise and it was re-written as

$$\frac{\partial}{\partial t} \left(A_x \ \theta_x \right) = \frac{\partial}{\partial z} \left(A_x \ K_x \frac{\partial \Phi_x}{\partial z} \right) + \frac{\partial}{\partial z} \left(A_x \ K_x \rho g \right) - S_x, \tag{5.1}$$

where $S_x = l_a T$, with $T = T(\Phi_x, z, t)$ (m s⁻¹) the transpiration rate, and $l_a = lA_s$ (m) the leaf area per unit of height; z (m) is bound between 0 at the bottom of the tree and L at the top of the tree.

For analytical tractability of Eq. (5.1), simplified formulations of the hydraulic conductivity and water capacitance are used. The hydraulic conductivity is assumed to decrease with the water potential following the vulnerability curve (Bohrer et al., 2005; Chuang et al., 2006)

$$K_x = K_m \ e^{\alpha_0 \Phi_x},\tag{5.2}$$

with α_0 (Pa⁻¹) an empirical constant and K_m (m² s⁻¹ Pa⁻¹) the maximum hydraulic conductivity. Eq. (5.2) implies that $d\Phi_x = dK_x/(\alpha_0 K_x)$. The xylem water content is assumed to depend on Φ according to

$$\theta_x = \theta_{res,x} + (\theta_{sat,x} - \theta_{res,x}) e^{\alpha_0 \Phi_x}, \tag{5.3}$$

with $\theta_{res,x}$ (-) and $\theta_{sat,x}$ (-) being the residual and maximum water content of the stem xylem. With the further assumption that $A_x = A_0 \exp(-\beta z)$, with β (m⁻¹) an empirical allometric parameter, Eq. (5.1), can be re-written as

$$\gamma_0 \frac{\partial K_x}{\partial t} = \frac{1}{\alpha_0} \frac{\partial^2 K_x}{\partial z^2} + \left(\rho g - \frac{\beta}{\alpha_0}\right) \frac{\partial K_x}{\partial z} - \rho g \beta K_x - l_a T e^{\beta z},\tag{5.4}$$

where $\gamma_0 = (\theta_{sat,x} - \theta_{res,x})/K_m$, (s Pa m⁻²). Eq. (5.4) is similar to the equation described in Chuang et al. (2006), and more details regarding its derivation can be found in this study.

Assuming that at time t = 0 the water potential is $\Phi_x(z, 0)$, the initial condition for Eq. (5.4) reads

$$K_x(z,0) = K_m e^{\alpha_0 \Phi_x(z,0)}.$$
(5.5)

The boundary condition at the bottom of the tree is defined by a time series of water potentials (i.e., $\Phi_0 = \Phi_x(0, t)$), which results in

$$K_x(0,t) = K_m e^{\alpha_0 \Phi_0(t)}.$$
(5.6)

The flux of water at the top of the tree (z = L) is zero, leading to the boundary condition

$$\left(\frac{1}{\alpha_0}\frac{\partial K_x}{\partial z} + K_x \rho g\right)_{z=L} = 0.$$
(5.7)

Solutions of Eq. (5.4) with initial and boundary conditions in Eqs. (5.5) - (5.7) can be obtained for different expressions of T(z,t) for some cases as presented in the following sections.

5.2.1 Simplified unsteady case

An exact solution of Eq. (5.4) can be obtained by assuming $\beta = 0$ and considering that the gradient of water potentials is the main contributor to the water fluxes (i.e., neglecting the term $\partial_z (A_x \ K_x \rho g)$ in Eq. 5.1)). With these assumptions, Eq. (5.4) becomes a linear diffusion equation with a sink term that can be re-written in compact form as $f_x(z,t) = l_a T$ and the boundary condition at the top of the tree reading $(\partial_z K_x / \alpha_0)_{z=L} = 0$.

A general solution of this equation can be written as (Polyanin, 2001)

$$K_{x}(z,t) = \int_{0}^{L} K_{x}(\xi,t)G(z,\xi,t)d\xi + \frac{1}{\gamma_{0}\alpha_{0}}\int_{0}^{t} K_{x}(0,\tau) \left[\frac{\partial}{\partial\xi}G(x,\xi,t-\tau)\right]_{\xi=0}d\tau + \int_{0}^{t}\int_{0}^{L} f_{x}(\xi,\tau)G(z,\xi,t-\tau)d\xi d\tau, \quad (5.8)$$

where

$$G(x,\xi,t) = \frac{2}{L} \sum_{n=0}^{\infty} \sin\left[\frac{\pi(2n+1)x}{2L}\right] \sin\left[\frac{\pi(2n+1)\xi}{2L}\right] \exp\left[-\frac{\pi^2(2n+1)^2t}{4L^2\gamma_0\alpha_0}\right].$$
 (5.9)

For a case where transpiration depends only on time, the sink is expressed as

$$f_x = T_m (1 - \cos(2\pi t/24)), \tag{5.10}$$

where T_m is the maximum transpiration rate, t is considered to be in hours, and it is assumed that $l_a = 1 \text{ (m}^{-1}\text{)}$.

A fixed potential, equal to 0 MPa, was considered at the bottom of the stem and along the vertical direction as initial condition. This solution was tested for a 6 m high tree with the

parameters listed in Table 5.1. Comparisons between the exact and numerical solutions using the sink term in Eq. (5.10) are shown in Figure (5.2). The errors associated with the numerical solution are small, reaching a maximum of approximately $0.25 \cdot 10^{-3}$ MPa at the top of the tree. The error followed the pattern of transpiration, reaching its peak during day time and corresponding to a maximum error of 0.09% of the exact solution. The mass balance error equalled 0.05% of the total water entering the tree during the simulated 2 days. Similarly, the lowest error could be observed at night, when transpiration approaches zero. The numerical solution presents errors that change periodically. After the influence of the initial condition disappears, the errors remain stable in time.



Figure 5.2: Left: water potentials (MPa) from the exact (lines) and numerical solutions (dots) using the sink term in Eq. (5.10) for the first 12 hours. For better visualization not all points are shown for the numerical solution. Right: difference between the exact and numerical solution (Δ) at 3 m and 6 m. The temporal and spatial resolutions are 0.05 h and 0.01 m, respectively.

For a case where transpiration depends on both time (t) and the vertical position (z), the sink is written as

$$f_x = T_m \ z \ (1 - \cos(2\pi t/24)), \tag{5.11}$$

where $l_a = z \,(m^{-1})$.

Comparisons between the analytical and numerical solutions using Eq. (5.11) are shown in Figure 5.3, where 0 MPa was assumed at the bottom of the tree and as initial condition along the vertical direction. The error for this case is higher than for the previous case, with a maximum value that is about 0.2% of the exact solution, with a mass balance error equal to 0.05% of the total water entering the tree during the simulated 2 days. These errors would reduce using smaller values of Δz .



Figure 5.3: Left: water potentials (MPa) from the exact (lines) and numerical solutions (dots) using the sink term in Eq. (5.11) for the first 12 hours. For better visualization not all points are shown for the numerical solution. Right: difference between the exact and numerical solution (Δ) at 3 m and 6 m. The temporal and spatial resolutions are 0.05 h and 0.01 m, respectively.

5.2.2 Steady-state solution

A solution of Eq. (5.4) at steady state can be obtained accounting for effects due to gravity and using a distribution of leaf area per unit of stem height. It is assumed that the leaf area per unit of height is compatible with Eq. (5.4) and satisfies $l_a(0) = 0$; a possible expression for $l_a(z)$ is

$$l_a(z) = \frac{l_m \beta_1}{\beta_2 - \beta_1} \left(\frac{\beta_2}{\beta_1}\right)^{\frac{\beta_2}{(\beta_2 - \beta_1)}} \left(e^{-\beta_1 z} - e^{-\beta_2 z}\right),\tag{5.12}$$

with $\beta_2 > \beta_1$. It is also assumed that the transpiration rate depends on the water potential and the elevation as

$$T = T_m e^{\alpha_0 \Phi_x} e^{\beta_1 - \beta}.$$
(5.13)

Accordingly, Eq. (5.4) at steady state reads

$$\frac{1}{\alpha_0}\frac{\partial^2 K_x}{\partial z^2} + \left(\rho g - \frac{\beta}{\alpha_0}\right)\frac{\partial K_x}{\partial z} - \rho g\beta K_x + \zeta(e^{-\eta - 1})K_x = 0,$$
(5.14)

where $\eta = \beta_2 - \beta_1 > 0$ and

$$\zeta = \frac{l_m T_m \beta_1}{A_0 K_m (\beta_2 - \beta_1)} \left(\frac{\beta_2}{\beta_1}\right)^{\beta_2 / (\beta_2 - \beta_1)}.$$
(5.15)

If it is assumed that the water potential initially has a generic profile and at the bottom of the tree remains constant in time, the water potential will stabilize in time to a steady profile with the

Table 5.1: List of parameters used in the comparison between the exact and numerical solutions (section 3.1)

Parameters	Value	Units	Description
β	0	m^{-1}	Allometric parameter, vertical reduction rate of cross-sectional area
β_1	0.2	m^{-1}	Empirical shape parameter for vertical leaf area distribution per unit xylem lenght
β_2	1	m^{-1}	Empirical shape parameter for vertical leaf area distribution per unit xylem lenght
l_m	0.2	m	Mean leaf area per unit of length
A_0	0.0045	m^2	Basal cross-sectional area of the stem xylem
T_m	$3.47\cdot 10^{-8}$	${\rm m~s^{-1}}$	Maximum transpiration rate
α_0	$5\cdot 10^{-7}$	Pa^{-1}	Empirical constant
$\theta_{res,x}$	0.1	-	Residual water content of the stem xylem
$\theta_{sat,x}$	0.6	-	Saturated water content of the stem xylem
K_m	$1.02\cdot 10^{-9}$	$\mathrm{m}^2~\mathrm{s}^{-1}\mathrm{Pa}^{-1}$	Maximum hydraulic conductivity

flux of water from the bottom of the tree equalling the flux of water being lost via transpiration.

The solution of Eq.(5.14) can thus be written as

$$K(z) = C_1 y^{(\alpha_0 - \beta)/\eta} J_v(y) + C_2 y^{(\alpha_0 - \beta)/\eta} Y_v(y),$$
(5.16)

where J_v (.) and Y_v (.) are the Bessel functions of the first and second kind (Abramowitz and Stegun, 1964) of order

$$v = \frac{[4\alpha_0\zeta + (\alpha_0 + \beta)^2]^{1/2}}{\eta},$$
(5.17)

and C_1 and C_2 are constants to be determined numerically by imposing the boundary conditions, as in Eqs. (5.6) and (5.7) with Φ_0 constant.

The agreement between the exact and numerical solutions is shown in Figure (5.4), for a case considering a bottom boundary condition of $\Phi_0=0$ MPa, a no-flux boundary condition at the top, and a hydrostatic initial condition. Steady state was reached after a short interval of about 3 hours of model time set. For a 6 meter high tree, the error of the numerical solution increases with elevation reaching approximately $0.4 \cdot 10^{-3}$ MPa at the tree top, being 0.4% of the exact value. According to the steady-state condition, the differences in storage between the last two consecutive model time steps approached zero and were equal to $-2.77 \cdot 10^{-18}$ m³, with transpiration equalling 99.97% of the total flux entering the tree. A larger error was reached in comparison to the unsteady state solution cases due to the more complex formulation used for the steady-case scenario.

5.3 Model application: case study

FETCH3 was tested against a case study described in Verma et al. (2014). For reproducibility purposes, FETCH3 used the same model setup, environmental variables, and parameters as Verma et al. (2014), where the software COMSOL Multiphysics (Ver. 4.1) was selected to solve



Figure 5.4: Left: water potentials, Φ , (MPa) at steady state obtained from the exact (black line) (Eq. 5.16), and numerical solutions (dots), using 0.05 m and 0.08 h as spatial and temporal resolution, respectively. For the numerical solution, not all points are shown for better visualization. The lines with light colors present the initial condition and the first 2 hours of simulation. Right: difference between the exact and numerical solution (Δ) at steady state condition along tree height.

the system of equations using finite elements. Details of the dataset are reported in Zeppel et al. (2008), Yunusa et al. (2012), and Verma et al. (2014), with a brief summary presented here.

5.3.1 Site description

The study site is located at latitude 33°39' 41" S and longitude 150°46' 57" E in New South Wales, Australia. According to the long term statistics (1993–2013 - Royal Australian Air Force base in Richmond, Australian Bureau of Meteorology, station 067105) the average daily minimum and maximum temperatures are 10 °C and 24 °C, with annual rainfall approximately 730 mm.

Rainfall, solar radiation, air temperature, and humidity were collected every 30 minutes from January 1st to June 4th in 2007. Sap flux data were collected for the same period, using the heat ratio technique at a half-hour resolution. The vegetation is dominated by *Eucalyptus parramattensis* C.A. Hall (Parramatta red gum) and *Angophora bakeri* E.C. Hall (narrow-leaved apple). The trees were 14 m tall on average, with a LAI between 1.3 and 1.9.

The soil is duplex, with a first layer up to a depth of 0.8 m being predominantly sand , with clay underneath. The soil parameters used in the model are listed in Table 5.2.

Parameters	Units	Sand	Clay	Description
k _{s,sat}	${\rm m~s^{-1}}$	$3.45 \cdot 10^{-5}$	$1.94 \cdot 10^{-7}$	Saturated hydraulic conductivity
θ_{sat}	_	0.47	0.55	Saturated volumetric soil moisture content
$ heta_{res}$	_	0.045	0.068	Residual volumetric soil water content
α	m^{-1}	14.5	0.8	van Genuchten parameter
n	_	2.4	1.5	van Genuchten parameter
$ heta_1$	_	0.05	0.08	Root water uptake reduction function parameter
$ heta_2$	-	0.09	0.12	Root water uptake reduction function parameter

Table 5.2: List of soil parameters used in the model application

5.3.2 Model setup

The system of equations was solved for a soil depth of 5 m, and trees with a height of 14 m and root depth of 3.2 m. The boundary condition at the soil bottom was a constant water potential equal to -0.06 MPa, corresponding to a water content of $0.28 \text{ m}^3 \text{m}^{-3}$. At the surface, measured rainfall was used as a flux boundary condition to compute soil water infiltration (refer to Section 4.3.2). The boundary conditions for the trees are a zero-flux condition at the bottom of the roots and, above-ground, transpiration is applied as a boundary condition at the top of the canopy. Daytime transpiration is modelled through the Penman Monteith equation (Allen et al., 1998) combined with a stomata conductance function (Jarvis, 1976), whereas night time transpiration value modulated by temperature, VPD, and water potential at night (refer to Section 4.4). In order to follow the same setup as in Verma et al. (2014), transpiration is not distributed along the stem, but is imposed as a flux concentrated at the top of the tree, and the water capacitance of the xylem in the roots and stem is assumed constant (Verma et al., 2014).

In the sand layer, soil initial conditions are assumed to be a constant water potential equal to -0.004 MPa, corresponding to a water content of $0.08 \text{ m}^3 \text{m}^{-3}$. In the clay layer, water potential below a depth of 3 m was constant and equal to -0.06 MPa. Between these two depths, water potential was interpolated linearly. For the tree, water potential linearly decreased from -0.06 MPa at the bottom of the roots to -0.22 MPa at the top of the canopy. The spatial resolution used was 0.1 m, and the time step 20 s. The list of parameters used in the model, including root water uptake and transpiration parameters, is in Table 5.3.

5.3.3 Results

The model predictions for sap-flux during the day compared well with observation during the entire measurement period (Figure 5.5a), reaching a R^2 value of 0.74. The total mass balance error in the soil represented -0.30% of total infiltration, and it was calculated as the change in soil water storage minus the difference between the flux entering (bottom boundary condition and infiltration) and exiting the soil (root water uptake). In the tree (root and stem xylem), the

Parameters	Units	Value	Description
A_x/A_s	-	$8.6 \cdot 10^{-4}$	Stem xylem cross sectional area index surface ratio
A_r/A_s	-	1	Root xylem cross sectional area index
A_{ls}/A_s	-	1	Lateral root surface area index
LAI	-	1.5	Leaf area index
$k_{s,rad}$	s^{-1}	$7.2 \cdot 10^{-10}$	Total soil-to-root radial conductance
C_x	Pa^{-1}	$1.1\cdot10^{-11}$	Stem xylem water capacitance
C_r	Pa^{-1}	$1.1 \cdot 10^{-11}$	Root xylem water capacitance
h	m	14	Tree height
C_p	${ m J}~{ m m}^{-3}~{ m K}^{-1}$	1200	Heat capacity of air
T_{opt}	К	289.15	Jarvis temperature parameter
λ	${ m J}~{ m m}^{-3}$	$2.51\cdot 10^9$	Latent heat of vaporization
γ	${\rm Pa}~{\rm K}^{-1}$	66.7	Psychometric constant
g_b	${\rm m~s^{-1}}$	$2\cdot 10^{-2}$	Leaf boundary layer conductance
g_a	${\rm m~s^{-1}}$	$2 \cdot 10^{-2}$	Aerodynamic conductance
k_r	$\mathrm{m}^2 \mathrm{W}^{-1}$	$5 \cdot 10^{-3}$	Jarvis radiation parameter
k_t	K^{-2}	$1.6 \cdot 10^{-3}$	Jarvis temperature parameter
k_d	Pa^{-1}	$1.1 \cdot 10^{-3}$	Jarvis vapor pressure deficit parameter
h_{x50}	m	-130	Jarvis leaf water potential parameter
g_{smax}	${\rm m~s^{-1}}$	$10\cdot 10^{-3}$	Maximum leaf stomatal conductance
n_l	-	2	Jarvis leaf water potential parameter
E_{max}	${\rm m~s^{-1}}$	$1 \cdot 10^{-9}$	Maximum night time transpiration
q_z	-	9	Root distribution parameter
k_{max}	${\rm m~s^{-1}}$	$1 \cdot 10^{-5}$	Maximum conductivity of saturated stem xylem
k_{sax}	${\rm m~s^{-1}}$	$1\cdot 10^{-5}$	Specific axial conductivity for the root system
a_p	Pa^{-1}	$2\cdot 10^{-6}$	Xylem cavitation parameter
b_p	Pa	$-1.5\cdot10^{-6}$	Xylem cavitation parameter

Table 5.3: List of parameters used in the application of the model as in Verma et al. (2014)

water mass error was -0.16% of the total infiltration, and was calculated as the change in water storage (in the stem and root xylem) minus the difference between the fluxes entering (root water uptake) and exiting (transpiration) the tree. The model maintained a continuous water potential along roots and stem xylem (Figure 5.5b). At midday, in the roots, water potential decreases almost linearly with elevation, while in the stem xylem, because of the transpiration flux at the top of the tree, it is non linear. The change in the gradient at the soil surface is due to the sharp change in the axial hydraulic conductivity, since the xylem cross-sectional area index for the stem ($A_x/A_s=8.6 \cdot 10^{-4}$) is different from that of the roots ($A_r/A_s=1$).

For the days shown in Figure5.5b, when transpiration is peaking, the water potential fluctuates between a minimum of -2.2 MPa at the tree top and -0.8 MPa at the bottom of the roots. This range of values is in agreement with the results from the original studies and the published literature (Choat et al., 2012; Franks et al., 2007; Quijano and Kumar, 2015; Verma et al., 2014).



Figure 5.5: a) Comparison between measured (T_{obs}) and modelled (T_{mod}) daily sap flux rates excluding fluxes during the night. b) Root and stem xylem water potential (MPa) as a function of elevation (z) at midday. The vertical position of 5 m (above z=0, which is defined as the bottom of the soil column) represents the interface between between the roots and the stem.

A comparison of modelled and observed time series of transpiration rates for a week in January (summer) and April (autumn) is shown in Figure 5.6. The model is able to reproduce the temporal patterns of transpiration during the day, and does not show large fluxes at night because of the simplified modelling of the stomatal conductance at night, as in Verma et al. (2014) (see Section 4.4).

FETCH3 was able to accurately represent the nonlinear interactions between the above- and below-ground components of the SPAC. From Figure 5.6, we can verify that root water uptake and transpiration are coupled, meaning that below-above ground interface is correctly repres-

ented by the model. Below-ground, shallow soil layers generated maximum rates of root water uptake (RWU) during most days, caused by greater root density and low water stress when water is readily available. During dryer days, with the decrease of soil moisture at the surface, considerable RWU was found in the deeper layers (approximately 20 - 30 cm from the soil surface). Root water uptake from deeper layers can be characterized as a hydraulic compensation path generated by rapid reductions in the top layers radial hydraulic conductivity, as it can be seen in Figure 5.6a, during the last 3 days in January.



Figure 5.6: Comparison between modelled (black line) and observed (blue circles) transpiration rates and modelled root water uptake (colormap, $mm h^{-1}$) during one-week periods in (a) January and in (b) April. The vertical position of 5 m (above z=0, which is defined as the bottom of the soil column) represents the interface between the roots and the above-ground stem xylem.

5.4 Modelling LAD and water capacitance

FETCH3 is able to simulate the distribution of transpiration along the vertical axis, as well as a water capacitance function for the roots and stem xylem. In order to test this capability, we applied FETCH3 using the same parameters and setup as in section 5.3.2, but changed how the transpiration and xylem water capacitance are modelled in the case study. For this experiment, transpiration is not a boundary condition at the tree top, but is distributed along the stem as in Eq. (4.10), with S_x/A_s depending on the leaf area density (LAD). At the tree top, a no-flux condition is applied. An empirical LAD function described in Lalic and Mihailovic (2004) was used, and a LAD profile suitable for *Eucalyptus* stands can be written as:

$$l(z) = l_{max} \left(\frac{h - z_m}{h - z}\right)^{n_0} \exp\left[n_0 \left(1 - \frac{h - z_m}{h - z}\right)\right],$$
(5.18)

where *h* is the tree height (m), l_{max} (m² m⁻³) is the maximum value of leaf area density in a layer, z_m (m) is the corresponding above-ground height of l_{max} , and n_0 (-) is an empirical parameter defined as

$$n_0 = \begin{cases} 6 & 0 \le z < z_m, \\ 0.5 & z_m \le z \le h. \end{cases}$$
(5.19)

The value of l_{max} can be calculated from the LAI imposing

$$LAI = \int_0^h l(z)dz.$$
 (5.20)

Following Chuang et al. (2006) and Bohrer et al. (2005), the water capacitance of the roots and stem xylem are

$$C_x(\Phi_x) = \frac{A_x}{A_s} \frac{\partial \theta_x}{\partial \Phi_x} = \frac{A_x p \theta_{sat,x}}{A_s \Phi_d} \left(\frac{\Phi_d - \Phi_x}{\Phi_d}\right)^{-(p+1)},$$
(5.21)

$$C_r(\Phi_r) = \frac{A_r}{A_s} \frac{\partial \theta_r}{\partial \Phi_r} = \frac{A_r p \theta_{sat,r}}{A_s \Phi_d} \left(\frac{\Phi_d - \Phi_r}{\Phi_d}\right)^{-(p+1)},$$
(5.22)

where Φ_d (Pa) and p (-) are empirical coefficients for the hydraulic system, and $\theta_{sat,x}$ (-) and $\theta_{sat,r}$ (-) are the water content at saturation for the stem and roots xylem, respectively. The values of these parameters are shown in Table 5.4.

From Figure 5.7, the vertical distribution of transpiration follows the shape of the LAD, with larger values of transpiration where the LAD is also large. Accordingly, Φ decreases along the tree height, in accordance with the no-flux boundary condition applied at the top of the tree.

5.5 Model application: calibration against measured sap flux data

FETCH3 was calibrated against sap flow data collected in the young blue gum plantation described in Chapter 3, referred here as Digby plantation. For these simulations, FETCH3 used the environmental variables collected at the site as input to run the model.



Figure 5.7: a) Transpiration fluxes $(mm h^{-1})$ as a function of the elevation (z). b) Water potential (MPa) along z, considering z=0 at the bottom of the soil, and z=5 m equal the bottom of the stem.

5.5.1 Data description

Transpiration was measured in individual trees using commercially available sap flow sensors (SFM1, ICT International, Australia). The SFM1 uses the heat pulse velocity (HPV) technique to measure sap flow rates from the velocity of a short pulse of heat moving along the xylem tissue. Sap flow sensors were installed approximately at breast height in 5 trees and HPV recorded at half-hourly intervals. Velocities were corrected for wounding effects following Burgess et al. (2001), using compatible values for *Eucalyptus* trees from the literature (Burgess et al., 2001; Choat et al., 2012; Fraser, 2020). Additional measurements of tree diameter at breast height (DBH) were performed by dendrometers (DBL60, ICT International, Australia) installed on the same trees. Stem diameter variations were recorded every 30 minutes. Given that plantation trees are generally homogeneous and equally spaced, the measured trees were assumed to be representative of the entire plantation.

Measurements were converted to sap flux density (SFD, $m^3m^{-2}h^{-1}$), based on wood core measurements of dry wood density, gravimetric sapwood moisture content, and sapwood area. For the plantation, sapwood area was assumed to be the entire stem, as it is commonly the case for young *Eucalyptus* trees (Benyon and Doody, 2015; Drake et al., 2012), and also verified

Table 5.4: List of parameters used i	in the application	of the model co	onsidering a water	capacit-
ance and a leaf area density functio	n			

Parameters	Units	Value	Description
l _{max}	$\mathrm{m}^{2}\mathrm{m}^{-3}$	0.4	Maximum value of leaf area density
z_m	m	11	Corresponding above-ground height of l_{max}
n_0	-	6 or 0.5	Empirical parameter (Eq. 5.19)
Φ_d	Pa	$5.74\cdot 10^8$	Empirical parameter for water pressure of dry xylem
p	-	20	Empirical coefficient
$ heta_{sat,x}$	-	0.58	Water content at saturation for the stem xylem
$ heta_{sat,r}$	-	0.58	Water content at saturation for the root xylem

during the core sampling. From average SFD, plot-level transpiration $(mm h^{-1})$ was obtained using the total sapwood area calculated based on the measurements of DBH, following Kume et al. (2010) and Marchionni et al. (2019). Figure 5.8 shows the plot-level transpiration and diameter change. The measurement period was 2^{nd} of January 2021 to 1^{st} of July 2021.



Figure 5.8: Sap flux density (SFD) and diameter at breast height (DBH) during the monitored period.

5.5.2 Model setup

A 5 m deep sandy-loam soil layer was considered in the model, following Dresel et al. (2018). The boundary condition at the soil bottom was constant water potential equal to $-3.06 \cdot 10^{-4}$ MPa, corresponding to a water content of 0.30 m³m⁻³. At the surface, measured rainfall was used as a flux boundary condition to compute soil water infiltration, considering a constant interception of 20% of rainfall. The percentage of interception was derived using an equation described in Amenu and Kumar (2007), which considers the average LAI of the site to calculate the amount of water intercepted by leaves. The boundary conditions for the trees were a zero-flux condition at the bottom of the roots and, above-ground, a zero-flux condition applied at the top of the tree. Transpiration is modelled as in Section 5.4, with transpiration and capacitance

Parameters	Units	Value	Reference
Tree density	trees hec^{-1}	750	Measured
Plot area	m^2	10000	Measured
Mean sapwood area	m^2	$1.2 \cdot 10^{-2}$	Measured
A_x/A_s	-	$1.2 \cdot 10^{-3}$	Estimated from sapwood area
A_r/A_s	-	$1.2\cdot 10^{-3}$	Estimated from sapwood area
A_{ls}/A_s	-	1	Estimated from sapwood area
LAI	-	2	Estimated from alometric equations
h	m	9	Measured
C_p	${ m J}~{ m m}^{-3}~{ m K}^{-1}$	1200	From Vermaet al. (2014)
T_{opt}	Κ	293.65	from Vermaet al. (2014)
$\dot{\lambda}$	${ m J}~{ m m}^{-3}$	$2.51\cdot 10^9$	From Vermaet al. (2014)
γ	${\rm Pa}~{\rm K}^{-1}$	66.7	From Vermaet al. (2014)
g_b	${\rm m~s^{-1}}$	$2\cdot 10^{-2}$	From Vermaet al. (2014)
g_a	${\rm m~s^{-1}}$	$2\cdot 10^{-2}$	From Vermaet al. (2014)
k_r	$\mathrm{m}^2~\mathrm{W}^{-1}$	$5 \cdot 10^{-3}$	From Vermaet al. (2014)
k_t	K^{-2}	$1.6 \cdot 10^{-3}$	From Vermaet al. (2014)
k_d	Pa^{-1}	$1.1 \cdot 10^{-3}$	From Vermaet al. (2014)
n_l	-	2	From Vermaet al. (2014)
E_{max}	${\rm m~s^{-1}}$	$1 \cdot 10^{-8}$	From Vermaet al. (2014)
q_{z}	-	12	Adapted from Vermaet al. (2014)
k_{max}	${\rm m~s^{-1}}$	10^{-8}	Adapted from Vermaet al. (2014)
k_{sax}	${\rm m~s^{-1}}$	10^{-8}	Adapted from Vermaet al. (2014)
$ heta_{sat}$	_	0.41	From vanGenuchten (1980)
$ heta_{res}$	_	0.065	From vanGenuchten (1980)
lpha	m^{-1}	7.5	From vanGenuchten (1980)
n	_	1.89	From vanGenuchten (1980)
$ heta_1$	_	0.02	Adapted from Vermaet al. (2014)
$ heta_2$	_	0.2	Adapted from Vermaet al. (2014)

Table 5.5: List of parameters values used in the application of FETCH3 for the Digby plantation

distributed along the stem.

For the soil, initial conditions were assumed to be a soil water potential of -0.01 MPa corresponding to a water content of $0.12 \text{ m}^3 \text{m}^{-3}$ for the first 80 cm from the surface. Below a depth of 3.6 m, soil water potential was constant and corresponding to $0.28 \text{ m}^3 \text{m}^{-3}$. Between these two depths, water potential was interpolated linearly. Initial conditions for the trees consisted in water potential hydrostatically decreasing from the bottom of the roots to the top of the stem, assuming that the bottom of the roots has the same water potential as the soil at the same depth. A root depth of 1.2 m and a tree height of 9 m were set. Parameters were estimated from the literature on *Eucalyptus globulus* trees (Barotto et al., 2017; Barrett et al., 2005; David et al., 1997; Drake et al., 2009, 2012; Franks et al., 2007) and from plant physiology and traits databases (Choat et al., 2012; Falster et al., 2015; Fraser, 2020; Pausas et al., 2016), and are listed in Table 5.5. The spatial resolution used was 0.1 m, and the time step 20 s.

5.5.3 Automatic calibration using PSO

The methodology chosen for the calibration was the Particle Swarm Optimization (PSO) (Kennedy and Eberhart, 1995). PSO is a heuristic optimization method and a bio-inspired algorithm, mimicking the collective behavior of a flock of birds or a swarm of bees. In this approach, a population of particles, representing individual solutions to the problem, moves through the solution space by performing multiple iterations until they converge to an optimal solution of an objective function (Scheerlinck et al., 2009). For this study, PSO was applied using PySwarms (https://pyswarms.readthedocs.io/, accessed 31/03/2022), a widely used Python-based tool.

PSO requires the definition of the objective function. For the calibration, the root mean square error (RMSE) was chosen, as

$$RMSE = \sqrt{\frac{\sum_{i=1}^{n} (y_i - \hat{y}_i)^2}{n}},$$
(5.23)

with *n* the number of data points, *y* the observed sap flux data and \hat{y} the simulated transpiration data. In this application, the overall RMSE was calculated based on the half-hourly values of the variables, excluding nighttime data. PSO performed consecutive simulations of FETCH3 until the difference in the RMSE function between simulations was less than $10^{-8} \text{ mm } 0.5 \text{ h}^{-1}$. The parameters chosen for the calibration were $k_{s,rad}$, g_{smax} , a_p , b_p , h_{x50} , and $k_{s,sat}$. These parameters were chosen based on the model sensitivity, and because they represent plantfunctional traits, such as resistance to cavitation and root uptake behavior. With the exception of $k_{s,sat}$ the calibrated parameters can express considerable differences according to plant species, age, site climate, and management practices (Matheny et al., 2017; Pappas et al., 2016). The parameter $k_{s,sat}$ is associated with a physical property of the soil, and should be independent of soil moisture. However, given the high spatial variability of soil properties, which can change over only a few centimetres (Manzoni et al., 2013b), $k_{s,sat}$ was calibrated separately for the two periods of time under analysis.

PSO requires the definition of an upper and lower bound for each parameter that will be calibrated. The upper and lower bounds chosen were based on values for blue gum trees found in the databases described in Choat et al. (2012), Falster et al. (2015), and Fraser (2020). The dataset was divided into dry (February - March 2021) and wet (May - June 2021) periods. Table 5.6 contains the resultant parameters for each period after calibration using PSO.

5.5.4 Results

Table 5.7 shows the calibration period, RMSE for the calibration, and the R² for the simulation period. The calibrated parameters during the wet period generated good predictions of transpir-

Parameters	Units	Dry period	Wet period
k _{s,rad}	s^{-1}	$1.30 \cdot 10^{-13}$	$3.26 \cdot 10^{-11}$
g_{smax}	${\rm m~s^{-1}}$	$2.04 \cdot 10^{-2}$	$2.09 \cdot 10^{-2}$
a_p	Pa^{-1}	$1.70 \cdot 10^{-6}$	$3.45 \cdot 10^{-7}$
b_p	Pa	$-1.00\cdot10^5$	$-1.33\cdot10^{6}$
h_{x50}	m	$-1.17\cdot10^5$	$-2.11\cdot 10^6$
$k_{s,sat}$	${\rm m~s^{-1}}$	$1.11\cdot 10^{-5}$	$1.27\cdot 10^{-5}$

Table 5.6: List of calibrated values of parameters for each simulation

ation that compared well with the measured sap flow from Digby over the same period. During May and June 2021, the R² for the daily transpiration, excluding nighttime values, was 0.66 (Figure 5.9, b), which is close to the value registered for the case study from Verma et al. (2014). The RMSE for the calibration period was small, considering that it was lower than the standard deviation of the dataset, which was 0.0164 mm $0.5h^{-1}$. Looking at the simulated hourly rates, FETCH3 predictions overestimated transpiration during the majority of May (Figure 5.9, c), but reached good agreement in June (Figure 5.9, a).

Table 5.7: Calibration and Simulation period, with the respective RMSE $(mm 0.5h^{-1})$ from the calibration with PSO and R² for the simulation period.

Simulation	Dry period	Wet period
Calibration period	07 Feb - 07 Mar	26 May - 26 Jun
RMSE calibration	$9.50 \cdot 10^{-3}$	$1.33\cdot10^{-2}$
Simulation period	01 Feb - 01 Apr	01 May - 01 Jul
R ² simulation	0.43	0.66

During the dry period, even though the RMSE was still smaller than the standard deviation of the calibration dataset ($0.0127 \text{ mm } 0.5 \text{hr}^{-1}$), the results were not as good as during the wet period, registering a R² of 0.43 (Figure 5.10, b). FETCH3 was not able to replicate completely the water stress in March 2021 (Figure 5.10, a), and overestimated transpiration during most of this period. Contrarily, during early February, the model underestimated transpiration (Figure 5.10, c).

The considerable growth over the monitored period, especially during winter, made it harder to derive a single set of calibrated parameters for the site, since tree DBH from February (124 mm) was considerably smaller than in June (132 mm). The increasing DBH influenced the calculation of the sap flux density over time. This made it challenging for FETCH3 to reproduce the behavior of the trees, giving that the current version of the model does not account for tree growth (above and below ground) or variation in sapwood area during simulations. Additionally, the high variability in DBH over a short period of time might have increased the uncertainties in the calculation of the plot-scale sap flux density. Uncertainties and errors in the data also play a role. Potential errors in measurement of SFD using the HPV method may include up to 35% underestimation of the actual SFD after corrections. Errors can increase with SFD



Figure 5.9: Results for the wet period: a) comparison between measured (T_{obs}) and modelled (T_{mod}) sap flux rates for a period of 10 days in June, b) comparison between T_{obs} and T_{mod} daily sap flux rates for the entire simulation period excluding fluxes during the night, and c) comparison between T_{obs} and T_{mod} for the entire simulation period.

due to sap wood heterogeneity (Steppe et al., 2010). In addition, SFD measurements do not account for the lag in water withdrawn from storage, which can account for 10 to 15% of the daily transpiration (Matheny et al., 2015).

The high sap flux during winter (when compared to summer) and the large sap flux variability over summer at the growing plantation are difficult to model with the Penman-Monteith formulation of transpiration. The Penman-Monteith formulation used here (Eq. 4.57) assumes that transpiration is proportional to net radiation and VPD, which are usually higher during summer and decrease as winter approaches (refer to Figure 3.2, Chapter 3). The young trees at Digby were constantly growing and, differently from the the northern-hemisphere ecosystems, they grew considerably during winter, when it is wet in the plantation. As a result, considering the decreasing VPD and radiation during May and June, FETCH3 was not able to replicate the large transpiration variability from February to July as a single simulation. Additionally, the current version of FETCH3 applies a simplified night-time transpiration function, which may have also contributed to the poor performance of the model during summer. Despite the limitations, FETCH3 reached a R² of 0.66 during the wet period.



Figure 5.10: Results for the dry period: a) comparison between measured (T_{obs}) and modelled (T_{mod}) sap flux rates for a period of 10 days in March, b) comparison between T_{obs} and T_{mod} daily sap flux rates for the entire simulation period excluding fluxes during the night, and c) comparison between T_{obs} and T_{mod} for the entire simulation period.

Applying the same model set up for both dry and wet periods, only changing the five calibrated parameters, may also have contributed to the lower R² registered for the dry period. Even though blue gum trees are considered an evergreen species, the seasonal variation in LAI can be substantial and cause different interception rates, also affecting the root water uptake capacity (Battaglia et al., 1998; White et al., 2010). This variation can be even larger for young trees, as considerable stand development, below and above ground, can occur in a few months period (for example, DBH variation in Figure 5.8). As a result, differences in tree development and traits, when comparing the wet and dry period, were not included, likely affecting the model performance. For example, θ_1 , θ_2 , and q_z were the same for dry and wet periods, even though root water uptake capacity (influenced by the LAI) (Christoffersen et al., 2016; Matheny et al., 2017; Pereira et al., 1997); this might have resulted in contrasting results. This can be seen around mid March, when the sap flux data shows a reduction in transpiration not replicated by FETCH3, which consistently overestimated the sap flow rates during the drier days.

Therefore, general model set up and parameters might have been a better fit to the wet period, resulting in improved results for May - July.

5.6 Conclusions

The Finite-difference Ecosystem-scale Tree Crown Hydrodynamics version 3 (FETCH3) was tested and applied in this study. By using a porous-media approach, FETCH3 is able to simulate intradaily dynamics of transpiration and provides a fast response to environmental variables. FETCH3 allows fidelity in the representation of hydraulic traits, which can be used to explore plant responses to water stress and xylem processes rather than land-atmosphere interactions.

We tested FETCH3 against exact numerical solutions of the equations and observations of transpiration. The numerical scheme of the model was applied to two simplified exact non-steady state cases, reaching a maximum error of approximately 0.2% with respect to the exact solution at the tree top of a 6 m high tree, for a case in which transpiration is dependent on both time and elevation. For a steady-state scenario, considering a more complex formulation, the error approached 0.4% of the exact solution at the tree top.

Simulated transpiration rates from FETCH3 reached an R² of 0.74 in comparison to observed sap flow rates from a published case study. In addition, values of water potential were continuous along roots and stem xylem, showing that water flux in the soil, roots, and stem are correctly coupled along the entire tree structure. By using a hydrodynamic set of equations, FETCH3 resolves the temporal and vertical dynamics of root water uptake, and stem and root water transport and storage. This allows FETCH3 to simulate hydrodynamic phenomena such as root water compensation following reductions of soil moisture in the shallow soil layers.

FETCH3 produced good results when calibrated against measured sap flow collected from a growing plantation during the wet period of May - July 2021, with a R² of 0.66. However, it did not performed well during a drier summer period (February and March 2021). Limitations in the application of FETCH3, for this case, included the complexity of simulating growing trees with a model formulation that does not include tree growth. Additionally, as the same model set up was applied for both periods, FETCH3 might not have been able to accurately simulate the stage of tree development during each period, given that growing trees can sharply change in only few months.

By comparing the model predictions of transpiration and soil and xylem water storage with different sets of parameters (describing the whole-tree hydraulic strategy of the trees), and different environmental forcing (describing realistic or hypothetical conditions and stress), FETCH3 allows model-based studies of the consequences of hydraulic traits and strategies of different tree species for above- and below-ground water transport, with a range of stem and root xylem hydraulic characteristics.

Chapter 6

Conclusion

In the next sections, the results of the project are summarized, the contributions of the thesis are presented, the practical implications of this project are discussed, and the future work and limitations of the project are highlighted.

6.1 Summary of results

6.1.1 Experimental work

This project quantified the water and carbon fluxes in a young *Eucalyptus globulus* plantation during its first 4 years after establishment in a site located in southwest Victoria, Australia. The fluxes were measured using the eddy covariance method with the equipment raised above the canopy as the trees grew; additional measurements of tree height and diameter at breast height were taken to monitor tree growth and provide estimations of LAI.

Measured evapotranspiration (ET) at the site did not vary considerably over the years, even though trees were growing and developing continuously. However, the Net Ecosystem Production (NEP) and Gross Ecosystem Production (GEP) substantially increased over time, with the plantation switching from being an overall carbon source to a carbon sink in approximately 2 years. Increases in carbon fluxes were a result of tree growth and development, with DBH and estimated Leaf Area Index (LAI) substantially increasing in time. The dynamics of tree growth dictated how evapotranspiration was partitioned as trees developed, with soil evaporation and understory transpiration decreasing as trees grew, as a result of increasing LAI and DBH. Therefore, although ET did not increase substantially as the trees grew, the large increase in GEP resulted in a large increase in water use efficiency (WUE) of the plantation during the monitoring period.

The different dynamics in water and carbon fluxes for young stands captured by this study high-

CHAPTER 6. CONCLUSION

lighted the need to evaluate stand age and species-specific characteristics. The management practices currently applied to commercial plantations in Australia only refer to an estimated annual water use. Differently from what commonly assumed to provide licenses for plantation establishment, the plantation in our study did not see a dramatic increase in ET as the trees grew. Small increments in water used generated substantial increases in productivity, highlight-ing that management practices should be site specific and consider the trade-offs of water used by the trees against the gain in carbon assimilation.

6.1.2 FETCH3

The Finite-difference Ecosystem-scale Tree Crown Hydrodynamics (FETCH3) was developed. FETCH3 simulates water flowing through the soil-plant-atmosphere continuum by assuming that the water movement in the xylem resembles flow in porous media, and uses a macroscopic approach, lumping soil, roots, and stems along the vertical direction. FETCH3 solves a system of three partial differential equations in a 1D domain to describe the water flow through the soil, root xylem, and stem xylem.

FETCH3 presented a novel methodology to fully couple the soil, roots, and stem, assuring the continuity of the water fluxes between these three compartments. FETCH3 can simulate above- and below-ground water transport, accounting for root water uptake strategies, such as root water compensation and hydraulic redistribution along the vertical direction. The numerical formulation of the model was tested against exact solutions (non-steady and steady cases), transpiration observations from a published case study, and calibration was performed for the trees at the growing plantation experimental site.

The model presented considerably small errors when compared to the exact solutions and was able to correctly represent transpiration patterns when compared to experimental data from a case study that provided calibrated parameters. FETCH3 produced good results when calibrated against measured sap flow data from the growing plantation collected during the wet period, but it did not performed well during the drier period. As the model does not include a formulation to simulate tree growth, FETCH3 was not able to fully replicate the behavior and transpiration variability of the growing trees.

6.2 Contributions of research

The present research entailed experimental and modelling elements, which included unique components. Together, both parts provided novel findings on the dynamics of young commercial plantations and water flow along the soil-plant-atmosphere continuum.

There is ongoing debate regarding the sustainability and management practices of commercial

CHAPTER 6. CONCLUSION

plantations in Australia. Therefore, the present research produced useful findings for future policies and licensing procedures. Some specifics contributions are described below.

- Unique dataset of water and carbon fluxes in young plantation trees in the first 4 years after planting. Only a few studies have captured the dynamics of water use and carbon assimilation in ecosystems growing rapidly as *E. globulus*. As commercial plantations are extensively planted in Australia and around the world (e.g., Portugal, Brazil, Colombia), this unique dataset might serve as an example for land and water management in different countries.
- 2. Quantification of vegetation response to stand growth and climate variability, which can be used to evaluate current management practices. Measured evapotranspiration at the site did not vary considerably over the years. However, carbon uptake substantially increased over time, with the plantation switching from being an overall carbon source to a carbon sink in approximately 2 years. This challenge the common perception that evapotranspiration rates in plantations are much larger than in pastures already in the first years after establishment.
- 3. FETCH3 was developed to simulate water fluxes through the soil, roots, and stem xylem. This was achieved by restructuring the coupling between these three components of the system when lumping the processes into a single, continuous vertical dimension. The full-coupling between soil, roots, and stem xylem guarantees the continuity of the water fluxes between these three components. The numerical formulation of FETCH3 was verified against exact solutions of simplified expressions of the equations, and the model performance was evaluated against observations of sap flux data collected during five months from a case study.

6.3 Practical Implications

Understanding the trade-offs between water use and carbon assimilation in commercial plantations is important given the economic and ecological impacts caused by plantation establishment and expansion (Saadaoui et al., 2017; Tomé et al., 2021). Therefore, continued cultivation of large scale commercial plantations requires the application of efficient management practices (Stape et al., 2004; White et al., 2010, 2014).

Because most of Australia's commercial plantation trees are established in former agricultural lands (Downham and Gavran, 2020; White et al., 2010), there is a concern that plantation trees may alter the catchment water balance through higher evapotranspiration rates than the former vegetation (Adelana et al., 2014; Dresel et al., 2018). Catchment studies have tried to compare contrasting land uses under the same climate characteristics. Most of these studies focused

CHAPTER 6. CONCLUSION

on the quantification of evapotranspiration and hydrologic differences between these two land uses. Results from these studies are not definitive, with experimental and modelling studies documenting streamflow reduction with afforestation (Adelana et al., 2014; Azarnivand et al., 2020; Brown et al., 2005) and other studies not finding significant alterations of streamflow (Brown et al., 2015). Some studies reported a decline in the water-table levels, possibly suggesting that plantation trees are consuming more water, through a deeper root system, when compared to pastures (Dean et al., 2016; Dresel et al., 2018).

Large annual transpiration rates often exceeding annual precipitation were observed by Benyon et al. (2006) and Benyon and Doody (2015) across several plantations in the Green Triangle, across South Australia and Victoria. Transpiration rates were especially high where the water table was up to about 6 m from the surface. These observations were used to delineate regulations for commercial plantation in South Australia, where simple water allocation models are being applied in order to represent hydrological consequences of plantation establishments (Greenwood, 2013). Accordingly, *E. globulus* trees are assumed to reach canopy closure after 3 years with evapotranspiration rates increasing in time, thus reducing groundwater recharge, which is estimated to stop completely after 3 years (Figure 2.2). These assumptions were not satisfied in the plantation of the present study, where evapotranspiration rates only slightly increased as the trees grew towards canopy closure. This was caused by the contribution of the understory vegetation to the ET, which was more prominent when the trees were young with a sparse canopy, and gradually decreased and was replaced by tree transpiration. Conversely, the gain in carbon assimilation increased dramatically in the first 2 years after tree establishment, with a rapid increase in water use efficiency.

The results from our study show that local conditions are important in determining the water use of a plantation and that, in some cases, the increase in water consumption might be outweighed by the large gain in carbon assimilation.

6.4 Limitations and future research

6.4.1 Experimental work

The measurements represent a valuable dataset, given the lack of data from growing ecosystem, as discussed in Chapter 2. Limitations to the experimental work are related to the difficulties in measuring important variables such as LAI, soil properties, and interception of rainfall, which certainly would provide more insights on the behavior and productivity of plantations. Part of this limitation was overcome by using allometric equations developed for *E. globulus* trees. Considering that blue gum trees are extensively planted, these type of equations are broadly used in the forestry field. The limited access to the study site also represented a limitation, especially during the lockdown periods the state of Victoria underwent in 2020 and 2021. This culminated in data being lost due to equipment malfunction during late 2020 and early 2021.

The results of this project highlighted the structural and physiological differences between young and mature trees, when compared to previous studies on mature and well-established forests (Benyon et al., 2009; Benyon and Doody, 2015; Benyon et al., 2006; Pereira et al., 1986). To the global scale, these differences emphasize the need for extensive and continuous research on the ecosystem carbon and water fluxes, as pointed out by several studies (Baldocchi et al., 2021; Cleverly et al., 2020; White et al., 2021). For example, in a recent study, Wardlaw (2022) emphasised the need for ongoing long-term data to understand resilience and predict future trends of terrestrial carbon and water cycle in response to future changing climatic and environmental conditions. In addition, by analysing a comprehensive set of ecosystem functions across a variety of biomes located in different climate zones, Migliavacca et al. (2021) showed that most variability between ecosystem functions is captured by ecosystem productivity, which is mainly explained by vegetation structure.

Therefore, the demand for long-term ecosystem data from a large range of vegetation structures underline the need to monitor a diversity of sites that include variable age, climate, and species. The resultant data sets will provide more accurate estimations of the carbon budget, ecosystem impacts, and forest productivity, which can create the basis for robust management practices and predictions of global ecosystem efficiency.

6.4.2 FETCH3

From a modeling perspective, FETCH3, as the majority of physically-based models, simplifies highly-dynamic processes in order to allow them to be described by mathematical equations. Limitations related to data availability also lead to more simplifications into the model. For example, LAI and height were kept constant even when simulating young trees, which likely impacted the results of the simulations.

The optimization scheme applied, PSO, could be further developed to include more parameters and optimization functions. However, the inclusion of more parameters may not improve the model results since it could over-fit the dataset and add unnecessary complexity to the model. A balance between parametrization and simplicity should be evaluated, also considering that most of the empirical parameters may not always have physical meaning. Currently, there are efforts to calibrate FETCH3 against evapotranspiration data from the AmeriFlux network, using a different calibration approach. These new studies might provide further insights on the calibration and parametrization of FETCH3, certainly benefiting the model outputs. Additional efforts into the calibration and enhancement of FETCH3 might improve model performance during periods of high variable transpiration, such as the case for the dry period at Digby Plant-
ation.

Future developments into FETCH3 should also focus in improving model processes. A good option would be the implementation of a more comprehensive transpiration formulation, as the Ball et al. (1987), Leuning (1990), and Leuning (1995), and stomata conductance optimization as Qiu and Katul (2020) and Liu et al. (2020). These approaches mechanistically describe the stomatal behavior and reproduce the hydraulic dynamics in the leaf by assuming that stomata conductance is proportional to carbon assimilation. This can lead to a better trait representation and estimations of transpiration variability.

Another valuable future contribution into FETCH3 would be the inclusion of tree growth into the solution of the equations, including dynamic xylem area, root depth, root density, and tree height. Even though FETCH3 algorithm allows to update parameters in time and space, changing xylem area and other structural tree parameters during the simulation would create an increment of water unaccounted for in the water and mass balance. This would create numerical instabilities that could ultimately add errors in the numerical scheme of the model. Further studies are required to estimate those errors and identify the best approach to account for xylem area variability and growth. Tree growth is an important factor that contributes to tree transpiration and tree-trait representation, and it would allow a more realistic modelling of plantation structure and development. However, developing a tree growth module into FETCH3 might add more complexity in the solution of the equations, so future research might also focus on a balance between model complexity and performance.

Future directions for FETCH3 could also include the addition of coexistence of species. This would provide the model with more realism, considering that even in monocultures, the presence of understory vegetation, for example, can impact the water balance and increase ecosystem water use. This would likely improve simulations of young and developing trees, such as the plantation monitored in this project. Lastly, the modelling of soil temperature, and the production of CO_2 from soil microorganisms and plant roots could provide valuable insights in the behavior and trait-based representation of trees and also improve the description of the key mechanisms regulating root water uptake.

Bibliography

- M. Abramowitz and I. A. Stegun. *Handbook of Mathematical Functions with Formulas, Graphs, and Mathematical Tables.* JSTOR, dec 1964. doi: 10.2307/2282672.
- E. Acuña, R. Rubilar, J. Cancino, T. J. Albaugh, and C. A. Maier. Economic assessment of eucalyptus globulus short rotation energy crops under contrasting silvicultural intensities on marginal agricultural land. *Land Use Policy*, 76:329–337, jul 2018. ISSN 0264-8377. doi: 10.1016/j.landusepol.2018.05.028.
- S. M. Adelana, P. E. Dresel, P. Hekmeijer, H. Zydor, J. A. Webb, M. Reynolds, and M. Ryan. A comparison of streamflow, salt and water balances in adjacent farmland and forest catchments in south-western victoria, australia. *Hydrological Processes*, 29(6):1630–1643, 2014. ISSN 0885-6087. doi: 10.1002/hyp.10281.
- R. Allen, L. Pereira, D. Raes, and M. Smith. Crop evapotranspiration guidelines for computing crop water requirements fao irrigation and drainage paper 56. rome: Fao, 1998.
- M. Almagro, J. López, J. I. Querejeta, and M. Martínez-Mena. Temperature dependence of soil co2 efflux is strongly modulated by seasonal patterns of moisture availability in a mediterranean ecosystem. *Soil Biology and Biochemistry*, 41(3):594–605, 2009. ISSN 0038-0717. doi: https://doi.org/10.1016/j.soilbio.2008.12.021.
- M. Almagro, J. I. Querejeta, C. Boix-Fayos, and M. Martínez-Mena. Links between vegetation patterns, soil c and n pools and respiration rate under three different land uses in a dry mediterranean ecosystem. *Journal of Soils and Sediments*, 13(4):641–653, jan 2013. ISSN 1439-0108. doi: 10.1007/s11368-012-0643-5.
- A. C. Almeida and P. J. Sands. Improving the ability of 3-pg to model the water balance of forest plantations in contrasting environments. *Ecohydrology*, 9(4):610–630, sep 2016. ISSN 1936-0584. doi: 10.1002/eco.1661.
- A. C. Almeida, J. J. Landsberg, and P. J. Sands. Parameterisation of 3-pg model for fast-growing eucalyptus grandis plantations. *Forest Ecology and Management*, 193(1-2):179–195, may 2004. ISSN 0378-1127. doi: 10.1016/j.foreco.2004.01.029.

- A. C. Almeida, J. V. Soares, J. J. Landsberg, and G. D. Rezende. Growth and water balance of eucalyptus grandis hybrid plantations in brazil during a rotation for pulp production. *Forest Ecology and Management*, 251(1-2):10–21, oct 2007. ISSN 03781127 (ISSN). doi: 10.1016/ j.foreco.2007.06.009.
- A. C. Almeida, A. Siggins, T. R. Batista, C. Beadle, S. Fonseca, and R. Loos. Mapping the effect of spatial and temporal variation in climate and soils on eucalyptus plantation production with 3-pg, a process-based growth model. *Forest Ecology and Management*, 259(9):1730–1740, apr 2010. ISSN 0378-1127. doi: 10.1016/j.foreco.2009.10.008.
- G. G. Amenu and P. Kumar. A model for hydraulic redistribution incorporating coupled soil-root moisture transport. *Hydrology and Earth System Sciences Discussions*, 4(5):3719–3769, oct 2007. ISSN 1812-2116. doi: 10.5194/hessd-4-3719-2007.
- B. D. Amiro, A. G. Barr, J. G. Barr, T. A. Black, R. Bracho, M. Brown, J. Chen, K. L. Clark, K. J. Davis, A. R. Desai, S. Dore, V. Engel, J. D. Fuentes, A. H. Goldstein, M. L. Goulden, T. E. Kolb, M. B. Lavigne, B. E. Law, H. A. Margolis, T. Martin, J. H. McCaughey, L. Misson, M. Montes-Helu, A. Noormets, J. T. Randerson, G. Starr, and J. Xiao. Ecosystem carbon dioxide fluxes after disturbance in forests of north america. *Journal of Geophysical Research*, 115, 2010. ISSN 0148-0227. doi: 10.1029/2010jg001390.
- P. M. Anthoni, M. H. Unsworth, B. E. Law, J. Irvine, D. D. Baldocchi, S. V. Tuyl, and D. Moore. Seasonal differences in carbon and water vapor exchange in young and old-growth ponderosa pine ecosystems. *Agricultural and Forest Meteorology*, 111(3):203–222, may 2002. ISSN 01681923 (ISSN). doi: 10.1016/S0168-1923(02)00021-7.
- M. Aubinet, T. Vesala, and D. Papale. *Eddy covariance: a practical guide to measurement and data analysis.* Springer Science & Business Media, 2012.
- A. Azarnivand, M. Camporese, S. Alaghmand, and E. Daly. Modeling hydrological impacts of afforestation on intermittent streams. *Science of The Total Environment*, 728:138748, aug 2020. ISSN 0048-9697. doi: 10.1016/j.scitotenv.2020.138748.
- D. Baldocchi. 'breathing' of the terrestrial biosphere: lessons learned from a global network of carbon dioxide flux measurement systems. *Australian Journal of Botany*, 56(1):1, 2008. ISSN 0067-1924. doi: 10.1071/bt07151.
- D. Baldocchi, H. Chu, and M. Reichstein. Inter-annual variability of net and gross ecosystem carbon fluxes: A review. *Agricultural and Forest Meteorology*, 249:520–533, feb 2018. ISSN 0168-1923. doi: https://doi.org/10.1016/j.agrformet.2017.05.015.
- D. Baldocchi, S. Ma, and J. Verfaillie. On the inter- and intra-annual variability of ecosystem evapotranspiration and water use efficiency of an oak savanna and annual grassland subjec-

ted to booms and busts in rainfall. *Global Change Biology*, 27(2):359–375, nov 2021. ISSN 1354-1013. doi: 10.1111/gcb.15414.

- D. D. Baldocchi. Assessing the eddy covariance technique for evaluating carbon dioxide exchange rates of ecosystems: Past, present and future. *Global Change Biology*, 9(4):479–492, apr 2003. ISSN 13541013 (ISSN). doi: 10.1046/j.1365-2486.2003.00629.x.
- J. T. Ball, I. E. Woodrow, and J. A. Berry. A Model Predicting Stomatal Conductance and its Contribution to the Control of Photosynthesis under Different Environmental Conditions, pages 221–224. Springer Netherlands, Dordrecht, 1987. ISBN 978-94-017-0519-6. doi: 10.1007/978-94-017-0519-6_48.
- M. D. Baptista, S. J. Livesley, E. G. Parmehr, M. Neave, and M. Amati. Variation in leaf area density drives the rainfall storage capacity of individual urban tree species. *Hydrological Processes*, 32(25):3729–3740, oct 2018. ISSN 0885-6087. doi: 10.1002/hyp.13255.
- A. J. Barotto, S. Monteoliva, J. Gyenge, A. Martínez-Meier, K. Moreno, N. Tesón, and M. E. Fernández. Wood density and anatomy of three eucalyptus species: Implications for hydraulic conductivity. *Forest Systems*, 26(1), 2017. doi: 10.5424/fs/2017261-10446.
- A. G. Barr, A. D. Richardson, D. Y. Hollinger, D. Papale, M. A. Arain, T. A. Black, G. Bohrer, D. Dragoni, M. L. Fischer, L. Gu, B. E. Law, H. A. Margolis, J. H. McCaughey, J. W. Munger, W. Oechel, and K. Schaeffer. Use of change-point detection for friction–velocity threshold evaluation in eddy-covariance studies. *Agricultural and Forest Meteorology*, 171-172:31–45, apr 2013. ISSN 0168-1923. doi: 10.1016/j.agrformet.2012.11.023.
- M. S. Barrett, K. A. Preiss, and R. Sinclair. Influence of newly imposed salinity and waterlogging on eucalyptus gracilis in south australia. *Tree Physiology*, 25(10):1339–1346, oct 2005. ISSN 0829-318X. doi: 10.1093/treephys/25.10.1339.
- M. S. Bartlett, G. Vico, and A. Porporato. Coupled carbon and water fluxes in cam photosynthesis: modeling quantification of water use efficiency and productivity. *Plant and Soil*, 383 (1-2):111–138, jun 2014. ISSN 0032-079X. doi: 10.1007/s11104-014-2064-2.
- M. Battaglia and P. Sands. Modelling site productivity of eucalyptus globulus in response to climatic and site factors. *Functional Plant Biology*, 24(6):831, 1997. ISSN 1445-4408. doi: 10.1071/pp97065.
- M. Battaglia, M. L. Cherry, C. L. Beadle, P. J. Sands, and A. Hingston. Prediction of leaf area index in eucalypt plantations: effects of water stress and temperature. *Tree Physiology*, 18 (8-9):521–528, aug 1998. ISSN 0829-318X. doi: 10.1093/treephys/18.8-9.521.
- M. Battaglia, P. Sands, D. White, and D. Mummery. Cabala: a linked carbon, water and nitrogen model of forest growth for silvicultural decision support. *Forest Ecology and Management*, 193(1-2):251–282, may 2004. ISSN 0378-1127. doi: 10.1016/j.foreco.2004.01.033.

- R. Benyon, T. Doody, S. Theiveyanathan, and V. Koul. Plantation forest water use in southwest victoria. Technical report, Forest Wood Products Australia Limited, 2009.
- R. G. Benyon and T. M. Doody. Water use by tree plantations in south east south australia. Technical report, CSIRO Forestry and Forest Products, 2004.
- R. G. Benyon and T. M. Doody. Comparison of interception, forest floor evaporation and transpiration in pinus radiata and eucalyptus globulus plantations. *Hydrological Processes*, 29 (6):1173–1187, 2015. ISSN 0885-6087. doi: 10.1002/hyp.10237.
- R. G. Benyon, S. Theiveyanathan, and T. M. Doody. Impacts of tree plantations on groundwater in south-eastern australia. *Australian Journal of Botany*, 54(2):181–192, 2006.
- O. Bergeron, H. A. Margolis, C. Coursolle, and M.-A. Giasson. How does forest harvest influence carbon dioxide fluxes of black spruce ecosystems in eastern north america? *Agricultural and Forest Meteorology*, 148(4):537–548, apr 2008. ISSN 0168-1923. doi: https://doi.org/10.1016/j.agrformet.2007.10.012.
- J. Beringer, L. B. Hutley, I. McHugh, S. K. Arndt, D. Campbell, H. A. Cleugh, J. Cleverly, V. R. De Dios, D. Eamus, B. Evans, C. Ewenz, P. Grace, A. Griebel, V. Haverd, N. Hinko-Najera, A. Huete, P. Isaac, K. Kanniah, R. Leuning, M. J. Liddell, C. MacFarlane, W. Meyer, C. Moore, E. Pendall, A. Phillips, R. L. Phillips, S. M. Prober, N. Restrepo-Coupe, S. Rutledge, I. Schroder, R. Silberstein, P. Southall, M. Sun Yee, N. J. Tapper, E. Van Gorsel, C. Vote, J. Walker, and T. Wardlaw. An introduction to the australian and new zealand flux tower network ozflux. *Biogeosciences*, 13(21):5895–5916, 2016. doi: 10.5194/bg-13-5895-2016.
- C. Beverly. Models of the catchmentanalysis tool (cat1d version 22). Technical report, Department of Sustainability and Environment, 2007.
- C. Beverly, M. Bari, B. Christy, M. Hocking, and K. Smettem. Predicted salinity impacts from land use change: comparison between rapid assessment approaches and a detailed modelling framework. *Australian Journal of Experimental Agriculture*, 45(11):1453, 2005. ISSN 0816-1089. doi: 10.1071/ea04192.
- D. Binkley, J. L. Stape, W. L. Bauerle, and M. G. Ryan. Explaining growth of individual trees: Light interception and efficiency of light use by eucalyptus at four sites in brazil. *Forest Ecology and Management*, 259(9):1704–1713, apr 2010. ISSN 0378-1127. doi: 10.1016/j. foreco.2009.05.037.
- B. Bjarnadottir, B. D. Sigurdsson, and A. Lindroth. A young afforestation area in iceland was a moderate sink to co₂ only a decade after scarification and establishment. *Biogeosciences*, 6(12):2895–2906, dec 2009. ISSN 1726-4189. doi: 10.5194/ bg-6-2895-2009.

- T. M. Bleby, I. J. Colquhoun, and M. A. Adams. Hydraulic traits and water use of eucalyptus on restored versus natural sites in a seasonally dry forest in southwestern australia. *Forest Ecology and Management*, 274:58–66, jun 2012. ISSN 0378-1127. doi: 10.1016/j.foreco. 2012.02.029.
- G. Bohrer, H. Mourad, T. A. Laursen, D. Drewry, R. Avissar, D. Poggi, R. Oren, and G. G. Katul. Finite element tree crown hydrodynamics model (fetch) using porous media flow within branching elements: A new representation of tree hydrodynamics. *Water Resources Research*, 41(11), nov 2005. ISSN 0043-1397. doi: 10.1029/2005wr004181.
- G. Bohrer, G. G. Katul, R. L. Walko, and R. Avissar. Exploring the effects of microscale structural heterogeneity of forest canopies using large-eddy simulations. *Boundary-layer meteorology*, 132(3):351–382, 2009.
- G. Bonan. Ecological climatology, 2016.
- G. B. Bonan, M. Williams, R. A. Fisher, and K. W. Oleson. Modeling stomatal conductance in the earth system: linking leaf water-use efficiency and water transport along the soil-plant-atmosphere continuum. *Geoscientific Model Development*, 7(5):2193–2222, sep 2014. ISSN 1991-9603. doi: 10.5194/gmd-7-2193-2014.
- G. B. Bonan, E. G. Patton, I. N. Harman, K. W. Oleson, J. J. Finnigan, Y. Lu, and E. A. Burakowski. Modeling canopy-induced turbulence in the earth system: a unified parameterization of turbulent exchange within plant canopies and the roughness sublayer (clm-ml v0). *Geoscientific Model Development*, 11(4):1467–1496, apr 2018. ISSN 1991-9603. doi: 10.5194/gmd-11-1467-2018.
- G. B. Bonan, E. G. Patton, J. J. Finnigan, D. D. Baldocchi, and I. N. Harman. Moving beyond the incorrect but useful paradigm: reevaluating big-leaf and multilayer plant canopies to model biosphere-atmosphere fluxes a review. *Agricultural and Forest Meteorology*, 306:108435, 2021. ISSN 0168-1923. doi: 10.1016/j.agrformet.2021.108435.
- V. Bonnesoeur, B. Locatelli, M. R. Guariguata, B. F. Ochoa-Tocachi, V. Vanacker, Z. Mao, A. Stokes, and S.-L. Mathez-Stiefel. Impacts of forests and forestation on hydrological services in the andes: A systematic review. *Forest Ecology and Management*, 433:569–584, 2019. ISSN 0378-1127. doi: 10.1016/j.foreco.2018.11.033.
- R. Bracho, G. Starr, H. L. Gholz, T. A. Martin, W. P. Cropper, and H. W. Loescher. Controls on carbon dynamics by ecosystem structure and climate for southeastern u.s. slash pine plantations. *Ecological Monographs*, 82(1):101–128, feb 2012. ISSN 00129615 (ISSN). doi: 10.1890/11-0587.1.

- L. Bren, P. Lane, and D. McGuire. An empirical, comparative model of changes in annual water yield associated with pine plantations in southern australia. *Australian Forestry*, 69(4): 275–284, jan 2006. ISSN 0004-9158. doi: 10.1080/00049158.2006.10676248.
- P. Broadbridge, E. Daly, and J. Goard. Exact solutions of the richards equation with nonlinear plant-root extraction. *Water Resources Research*, 53(11):9679–9691, nov 2017. ISSN 0043-1397. doi: 10.1002/2017wr021097.
- A. E. Brown, L. Zhang, T. A. McMahon, A. W. Western, and R. A. Vertessy. A review of paired catchment studies for determining changes in water yield resulting from alterations in vegetation. *Journal of Hydrology*, 310(1-4):28–61, aug 2005. ISSN 0022-1694. doi: 10.1016/j.jhydrol.2004.12.010.
- A. E. Brown, P. B. Hairsine, and A. Freebairn. The development of the tasmanian land use change and stream flow (taslucas) tool. Technical report, Report to the Tasmanian Natural Resource Management Committees, CSIRO Land and Water Science Report 54/06, 2006.
- S. C. Brown, V. L. Versace, R. E. Lester, and M. T. Walter. Assessing the impact of drought and forestry on streamflows in south-eastern australia using a physically based hydrological model. *Environmental Earth Sciences*, 74(7):6047–6063, jun 2015. ISSN 1866-6280. doi: 10.1007/s12665-015-4628-8.
- C. Brümmer, T. A. Black, R. S. Jassal, N. J. Grant, D. L. Spittlehouse, B. Chen, Z. Nesic, B. D. Amiro, M. A. Arain, A. G. Barr, C. P.-A. Bourque, C. Coursolle, A. L. Dunn, L. B. Flanagan, E. R. Humphreys, P. M. Lafleur, H. A. Margolis, J. H. McCaughey, and S. C. Wofsy. How climate and vegetation type influence evapotranspiration and water use efficiency in canadian forest, peatland and grassland ecosystems. *Agricultural and Forest Meteorology*, 153:14–30, feb 2012. ISSN 0168-1923. doi: 10.1016/j.agrformet.2011.04.008.
- G. Burba and D. Anderson. Introduction to the Eddy Covariance Method: General Guidelines and Conventional Workflow. 01 2007. ISBN E-BOOK ID: 1951D438-A0374350AA92950EFF6457. doi: 10.13140/RG.2.1.3723.5683.
- S. S. O. Burgess, M. A. Adams, N. C. Turner, C. R. Beverly, C. K. Ong, A. A. H. Khan, and T. M. Bleby. An improved heat pulse method to measure low and reverse rates of sap flow in woody plants. *Tree Physiology*, 21(9):589–598, jun 2001. ISSN 0829-318X. doi: 10.1093/ treephys/21.9.589.
- O. M. R. Cabral, J. H. C. Gash, H. R. Rocha, C. Marsden, M. A. V. Ligo, H. C. Freitas, J. D. Tatsch, and E. Gomes. Fluxes of co2 above a plantation of eucalyptus in southeast brazil. *Agricultural and Forest Meteorology*, 151(1):49–59, 2011. doi: 10.1016/j.agrformet.2010.09. 003.

- O. M. R. Cabral, H. C. Freitas, S. V. Cuadra, C. A. de Andrade, N. P. Ramos, P. Grutzmacher, M. Galdos, A. P. C. Packer, H. R. da Rocha, and P. Rossi. The sustainability of a sugarcane plantation in brazil assessed by the eddy covariance fluxes of greenhouse gases. *Agricultural and Forest Meteorology*, 282-283, 2020. doi: 10.1016/j.agrformet.2019.107864.
- I. R. Calder. Forests and water—ensuring forest benefits outweigh water costs. *Forest Ecology* and Management, 251(1-2):110–120, oct 2007. ISSN 0378-1127. doi: 10.1016/j.foreco.2007. 06.015.
- O. C. Campoe, C. A. Alvares, R. L. Carneiro, D. Binkley, M. G. Ryan, R. M. Hubbard, J. Stahl, G. Moreira, L. F. Moraes, and J. L. Stape. Climate and genotype influences on carbon fluxes and partitioning in eucalyptus plantations. *Forest Ecology and Management*, 475, 2020. doi: 10.1016/j.foreco.2020.118445.
- M. A. Celia, E. T. Bouloutas, and R. L. Zarba. A general mass-conservative numerical solution for the unsaturated flow equation. *Water Resources Research*, 26:1483–1496, 1990. ISSN 0043-1397. doi: 10.1029/wr026i007p01483.
- F. C. C. Chan, M. Altaf Arain, M. Khomik, J. J. Brodeur, M. Peichl, N. Restrepo-Coupe, R. Thorne, E. Beamesderfer, S. McKenzie, B. Xu, H. Croft, M. Pejam, J. Trant, M. Kula, and R. Skubel. Carbon, water and energy exchange dynamics of a young pine plantation forest during the initial fourteen years of growth. *Forest Ecology and Management*, 410: 12–26, feb 2018. ISSN 0378-1127. doi: 10.1016/j.foreco.2017.12.024.
- F. Chapin, G. M. Woodwell, J. T. Randerson, E. B. Rastetter, G. M. Lovett, D. D. Baldocchi, D. A. Clark, M. E. Harmon, D. S. Schimel, R. Valentini, C. Wirth, J. D. Aber, J. J. Cole, M. L. Goulden, J. W. Harden, M. Heimann, R. W. Howarth, P. A. Matson, A. D. McGuire, J. M. Melillo, H. A. Mooney, J. C. Neff, R. A. Houghton, M. L. Pace, M. G. Ryan, S. W. Running, O. E. Sala, W. H. Schlesinger, and E. D. Schulze. Reconciling carbon-cycle concepts, terminology, and methods. *Ecosystems*, 9(7):1041–1050, 2006. ISSN 1435-0629. doi: 10.1007/s10021-005-0105-7.
- F. Chapin, P. A. Matson, and P. M. Vitousek. Principles of terrestrial ecosystem ecology. 2012.
- Y. Chen, J. Ryder, V. Bastrikov, M. J. McGrath, K. Naudts, J. Otto, C. Ottlé, P. Peylin, J. Polcher, A. Valade, A. Black, J. A. Elbers, E. Moors, T. Foken, E. van Gorsel, V. Haverd, B. Heinesch, F. Tiedemann, A. Knohl, S. Launiainen, D. Loustau, J. Ogée, T. Vessala, and S. Luyssaert. Evaluating the performance of land surface model orchidee-can v1.0 on water and energy flux estimation with a single- and multi-layer energy budget scheme. *Geoscientific Model Development*, 9(9):2951–2972, 2016. ISSN 1991-9603. doi: 10.5194/gmd-9-2951-2016.
- B. Choat, S. Jansen, T. J. Brodribb, H. Cochard, S. Delzon, R. Bhaskar, S. J. Bucci, T. S. Feild, S. M. Gleason, U. G. Hacke, A. L. Jacobsen, F. Lens, H. Maherali, J. Martínez-Vilalta, S. Mayr,

M. Mencuccini, P. J. Mitchell, A. Nardini, J. Pittermann, R. B. Pratt, J. S. Sperry, M. Westoby, I. J. Wright, and A. E. Zanne. Global convergence in the vulnerability of forests to drought. *Nature*, 491(7426):752–755, nov 2012. ISSN 0028-0836. doi: 10.1038/nature11688.

- M. Christina, Y. Nouvellon, J. P. Laclau, J. L. Stape, O. C. Campoe, and G. le Maire. Sensitivity and uncertainty analysis of the carbon and water fluxes at the tree scale in eucalyptus plantations using a metamodeling approach. *Canadian Journal of Forest Research*, 46(3): 297–309, 2016. ISSN 0045-5067. doi: 10.1139/cjfr-2015-0173.
- B. O. Christoffersen, M. Gloor, S. Fauset, N. M. Fyllas, D. R. Galbraith, T. R. Baker, B. Kruijt, L. Rowland, R. A. Fisher, O. J. Binks, S. Sevanto, C. Xu, S. Jansen, B. Choat, M. Mencuccini, N. G. McDowell, and P. Meir. Linking hydraulic traits to tropical forest function in a size-structured and trait-driven model (tfs v.1-hydro). *Geoscientific Model Development*, 9(11): 4227–4255, 2016. doi: 10.5194/gmd-9-4227-2016.
- Y. L. Chuang, R. Oren, A. L. Bertozzi, N. Phillips, and G. G. Katul. The porous media model for the hydraulic system of a conifer tree: Linking sap flux data to transpiration rate. *Ecological Modelling*, 191(3-4):447–468, feb 2006. ISSN 0304-3800. doi: 10.1016/j.ecolmodel.2005.03. 027.
- K. L. Clark, H. L. Gholz, and M. S. Castro. Carbon dynamics along a chronosequence of slash pine plantations in north florida. *Ecological Applications*, 14(4):1154–1171, aug 2004. ISSN 1051-0761. doi: 10.1890/02-5391.
- J. Cleverly, C. Vote, P. Isaac, C. Ewenz, M. Harahap, J. Beringer, D. I. Campbell, E. Daly, D. Eamus, L. He, J. Hunt, P. Grace, L. B. Hutley, J. Laubach, M. McCaskill, D. Rowlings, S. Rutledge Jonker, L. A. Schipper, I. Schroder, B. Teodosio, Q. Yu, P. R. Ward, J. P. Walker, J. A. Webb, and S. P. P. Grover. Carbon, water and energy fluxes in agricultural systems of australia and new zealand. *Agricultural and Forest Meteorology*, 287:107934, 2020. ISSN 0168-1923. doi: https://doi.org/10.1016/j.agrformet.2020.107934.
- C. Coursolle, H. A. Margolis, M. A. Giasson, P. Y. Bernier, B. D. Amiro, M. A. Arain, A. G. Barr, T. A. Black, M. L. Goulden, J. H. McCaughey, J. M. Chen, A. L. Dunn, R. F. Grant, and P. M. Lafleur. Influence of stand age on the magnitude and seasonality of carbon fluxes in canadian forests. *Agricultural and Forest Meteorology*, 165:136–148, nov 2012. ISSN 0168-1923. doi: 10.1016/j.agrformet.2012.06.011.
- V. Couvreur, J. Vanderborght, and M. Javaux. A simple three-dimensional macroscopic root water uptake model based on the hydraulic architecture approach. *Hydrology and Earth System Sciences Discussions*, 9(4):4943–4987, apr 2012. ISSN 1812-2116. doi: 10.5194/ hessd-9-4943-2012.

- V. Couvreur, G. Ledder, S. Manzoni, D. A. Way, E. B. Muller, and S. E. Russo. Water transport through tall trees: A vertically explicit, analytical model of xylem hydraulic conductance in stems. *Plant, Cell & Environment*, 41(8):1821–1839, 2018. doi: 10.1111/pce.13322.
- I. R. Cowan. Transport of water in the soil-plant-atmosphere system. *Journal of Applied Ecology*, 2(1):221–239, 1965. ISSN 00218901, 13652664. doi: 10.2307/2401706.
- P. M. Cristiano, M. V. E. Díaz Villa, M. S. De Diego, M. V. Lacoretz, N. Madanes, and G. Goldstein. Carbon assimilation, water consumption and water use efficiency under different land use types in subtropical ecosystems: from native forests to pine plantations. *Agricultural and Forest Meteorology*, 291:108094, 2020. ISSN 0168-1923. doi: https://doi.org/10.1016/j.agrformet.2020.108094.
- P. Cruiziat, H. Cochard, and T. Améglio. Hydraulic architecture of trees: main concepts and results. *Annals of Forest Science*, 59(7):723–752, nov 2002. ISSN 1286-4560. doi: 10.1051/ forest:2002060.
- B. B. da Silva, B. P. Wilcox, V. P. R. da Silva, S. M. G. L. Montenegro, and L. M. M. de Oliveira. Changes to the energy budget and evapotranspiration following conversion of tropical savannas to agricultural lands in são paulo state, brazil. *Ecohydrology*, 8(7):1272–1283, 2015. doi: 10.1002/eco.1580.
- E. Daly, A. Porporato, and I. Rodriguez-Iturbe. Coupled dynamics of photosynthesis, transpiration, and soil water balance. part ii: Stochastic analysis and ecohydrological significance. *Journal of Hydrometeorology*, 5(3):559–566, jun 2004a. ISSN 1525-755X. doi: 10.1175/1525-7541(2004)005<0559:cdopta>2.0.co;2.
- E. Daly, A. Porporato, and I. Rodriguez-Iturbe. Coupled dynamics of photosynthesis, transpiration, and soil water balance. part i: Upscaling from hourly to daily level. *Journal of Hydrometeorology*, 5(3):546–558, 2004b. doi: 10.1175/1525-7541(2004)005<0546:CDOPTA>2.0.CO; 2.
- E. Daly, S. Calabrese, J. Yin, and A. Porporato. Linking parametric and water-balance models of the budyko and turc spaces. *Advances in Water Resources*, 134:103435, dec 2019. ISSN 0309-1708. doi: 10.1016/j.advwatres.2019.103435.
- H. Daneshmand, S. Alaghmand, M. Camporese, A. Talei, P. J.-F. Yeh, and E. Daly. Long-term impacts of partial afforestation on water and salt dynamics of an intermittent catchment under climate change. *Water*, 12(4):1067, apr 2020. ISSN 2073-4441. doi: 10.3390/w12041067.
- T. S. David, M. I. Ferreira, J. S. David, and J. S. Pereira. Transpiration from a mature eucalyptus globulus plantation in portugal during a spring-summer period of progressively higher water deficit. *Oecologia*, 110(2):153–159, 1997. ISSN 00298549 (ISSN). doi: 10.1007/PL00008812.

- E. A. Davidson, E. Belk, and R. D. Boone. Soil water content and temperature as independent or confounded factors controlling soil respiration in a temperate mixed hardwood forest. *Global Change Biology*, 4(2):217–227, 1998. ISSN 13541013 (ISSN). doi: 10.1046/j.1365-2486. 1998.00128.x.
- A. C. de Araújo, A. J. Dolman, M. J. Waterloo, J. H. C. Gash, B. Kruijt, F. B. Zanchi, J. M. E. de Lange, R. Stoevelaar, A. O. Manzi, A. D. Nobre, R. N. Lootens, and J. Backer. The spatial variability of co2 storage and the interpretation of eddy covariance fluxes in central amazonia. *Agricultural and Forest Meteorology*, 150(2):226–237, feb 2010. ISSN 0168-1923. doi: 10.1016/j.agrformet.2009.11.005.
- Q. de Jong van Lier, J. C. van Dam, K. Metselaar, R. de Jong, and W. H. M. Duijnisveld.
 Macroscopic root water uptake distribution using a matric flux potential approach. *Vadose Zone Journal*, 7(3):1065–1078, 2008. ISSN 1539-1663. doi: 10.2136/vzj2007.0083.
- J. F. Dean, M. Camporese, J. A. Webb, S. P. Grover, P. E. Dresel, and E. Daly. Water balance complexities in ephemeral catchments with different land uses: Insights from monitoring and distributed hydrologic modeling. *Water Resources Research*, 52(6):4713–4729, jun 2016. ISSN 0043-1397. doi: 10.1002/2016wr018663.
- R. Downham and M. Gavran. Australian plantation statistics 2020 update. Technical report, ABARES, 2020.
- J. E. Drake, M. G. Tjoelker, A. Vårhammar, B. Medlyn, P. B. Reich, A. Leigh, S. Pfautsch, C. J. Blackman, R. López, M. J. Aspinwall, K. Y. Crous, R. A. Duursma, D. Kumarathunge, M. G. D. Kauwe, M. Jiang, A. B. Nicotra, D. T. Tissue, B. Choat, O. K. Atkin, and C. V. M. Barton. Trees tolerate an extreme heatwave via sustained transpirational cooling and increased leaf thermal tolerance. *Global Change Biology*, 24(6):2390–2402, feb 2018. ISSN 1354-1013. doi: 10.1111/gcb.14037.
- P. L. Drake, D. S. Mendham, D. A. White, and G. N. Ogden. A comparison of growth, photosynthetic capacity and water stress in eucalyptus globulus coppice regrowth and seedlings during early development. *Tree Physiology*, 29(5):663–674, feb 2009. ISSN 0829-318X. doi: 10.1093/treephys/tpp006.
- P. L. Drake, D. S. Mendham, D. A. White, G. N. Ogden, and B. Dell. Water use and water-use efficiency of coppice and seedling eucalyptus globulus labill.: a comparison of stand-scale water balance components. *Plant and Soil*, 350(1-2):221–235, jul 2012. ISSN 0032-079X. doi: 10.1007/s11104-011-0897-5.
- P. E. Dresel, J. F. Dean, F. Perveen, J. A. Webb, P. Hekmeijer, S. M. Adelana, and E. Daly. Effect of eucalyptus plantations, geology, and precipitation variability on water resources in upland intermittent catchments. *Journal of Hydrology*, 564:723–739, 2018. ISSN 0022-1694. doi: 10.1016/j.jhydrol.2018.07.019.

- D. T. Drewry, P. Kumar, S. Long, C. Bernacchi, X. Z. Liang, and M. Sivapalan. Ecohydrological responses of dense canopies to environmental variability: 1. interplay between vertical structure and photosynthetic pathway. *Journal of Geophysical Research: Biogeosciences*, 115 (G4), 2010. ISSN 0148-0227. doi: 10.1029/2010JG001340.
- P. Dye. Modelling growth and water use in four pinus patula stands with the3-pg model. *The Southern African Forestry Journal*, 191(1):53–63, jul 2001. ISSN 1029-5925. doi: 10.1080/20702620.2001.10434151.
- P. Dye. A review of changing perspectives on eucalyptus water-use in south africa. *Forest Ecology and Management*, 301:51–57, 2013. ISSN 03781127 (ISSN). doi: 10.1016/j.foreco. 2012.08.027.
- P. Dye and D. Versfeld. Managing the hydrological impacts of south african plantation forests: An overview. *Forest Ecology and Management*, 251(1-2):121–128, oct 2007. ISSN 0378-1127. doi: 10.1016/j.foreco.2007.06.013.
- P. J. Dye. Water use efficiency in south african eucalyptus plantations: A review. *Southern African Forestry Journal*, 189(189):17–26, 2000.
- P. J. Dye, S. Jacobs, and D. Drew. Verification of 3-pg growth and water-use predictions in twelve eucalyptus plantation stands in zululand, south africa. *Forest Ecology and Management*, 193(1-2):197–218, may 2004. ISSN 0378-1127. doi: 10.1016/j.foreco.2004.01.030.
- D. Eamus, J. Cleverly, N. Boulain, N. Grant, R. Faux, and R. Villalobos-Vega. Carbon and water fluxes in an arid-zone acacia savanna woodland: An analyses of seasonal patterns and responses to rainfall events. *Agricultural and Forest Meteorology*, 182-183:225–238, dec 2013. ISSN 0168-1923. doi: 10.1016/j.agrformet.2013.04.020.
- E. Falge, D. Baldocchi, R. Olson, P. Anthoni, M. Aubinet, C. Bernhofer, G. Burba, R. Ceulemans, R. Clement, H. Dolman, A. Granier, P. Gross, T. Grünwald, D. Hollinger, N.-O. Jensen, G. Katul, P. Keronen, A. Kowalski, C. T. Lai, B. E. Law, T. Meyers, J. Moncrieff, E. Moors, J. Munger, K. Pilegaard, Üllar Rannik, C. Rebmann, A. Suyker, J. Tenhunen, K. Tu, S. Verma, T. Vesala, K. Wilson, and S. Wofsy. Gap filling strategies for defensible annual sums of net ecosystem exchange. *Agricultural and Forest Meteorology*, 107(1):43–69, mar 2001. ISSN 0168-1923. doi: 10.1016/s0168-1923(00)00225-2.
- D. S. Falster, R. A. Duursma, M. I. Ishihara, D. R. Barneche, R. G. FitzJohn, A. Vårhammar, M. Aiba, M. Ando, N. Anten, M. J. Aspinwall, J. L. Baltzer, C. Baraloto, M. Battaglia, J. J. Battles, B. Bond-Lamberty, M. van Breugel, J. Camac, Y. Claveau, L. Coll, M. Dannoura, S. Delagrange, J.-C. Domec, F. Fatemi, W. Feng, V. Gargaglione, Y. Goto, A. Hagihara, J. S. Hall, S. Hamilton, D. Harja, T. Hiura, R. Holdaway, L. S. Hutley, T. Ichie, E. J. Jokela, A. Kantola, J. W. G. Kelly, T. Kenzo, D. King, B. D. Kloeppel, T. Kohyama, A. Komiyama,

J.-P. Laclau, C. H. Lusk, D. A. Maguire, G. le Maire, A. Mäkelä, L. Markesteijn, J. Marshall, K. McCulloh, I. Miyata, K. Mokany, S. Mori, R. W. Myster, M. Nagano, S. L. Naidu, Y. Nouvellon, A. P. O'Grady, K. L. O'Hara, T. Ohtsuka, N. Osada, O. O. Osunkoya, P. L. Peri, A. M. Petritan, L. Poorter, A. Portsmuth, C. Potvin, J. Ransijn, D. Reid, S. C. Ribeiro, S. D. Roberts, R. Rodríguez, A. Saldaña-Acosta, I. Santa-Regina, K. Sasa, N. G. Selaya, S. C. Sillett, F. Sterck, K. Takagi, T. Tange, H. Tanouchi, D. Tissue, T. Umehara, H. Utsugi, M. A. Vadeboncoeur, F. Valladares, P. Vanninen, J. R. Wang, E. Wenk, R. Williams, F. de Aquino Ximenes, A. Yamaba, T. Yamada, T. Yamakura, R. D. Yanai, and R. A. York. Baad: a biomass and allometry database for woody plants. *Ecology*, 96(5):1445–1445, 2015. ISSN 0012-9658. doi: 10.1890/14-1889.1.

FAO and UNEP. The state of the world's forests 2020. Technical report, 2020.

- K. A. Farley, E. G. Jobbagy, and R. B. Jackson. Effects of afforestation on water yield: a global synthesis with implications for policy. *Global Change Biology*, 11(10):1565–1576, oct 2005. ISSN 1354-1013. doi: 10.1111/j.1365-2486.2005.01011.x.
- S. Fatichi, C. Pappas, and V. Y. Ivanov. Modeling plant–water interactions: an ecohydrological overview from the cell to the global scale. *WIREs Water*, 3(3):327–368, nov 2016. ISSN 2049-1948. doi: 10.1002/wat2.1125.
- S. Fatichi, C. Pappas, J. Zscheischler, and S. Leuzinger. Modelling carbon sources and sinks in terrestrial vegetation. *New Phytologist*, 221(2):652–668, 2019. ISSN 0028-646X. doi: 10.1111/nph.15451.
- R. A. Feddes, P. Kowalik, K. Kolinska-Malinka, and H. Zaradny. Simulation of field water uptake by plants using a soil water dependent root extraction function. *Journal of Hydrology*, 31 (1-2):13–26, sep 1976. ISSN 0022-1694. doi: 10.1016/0022-1694(76)90017-2.
- P. M. Feikema, J. D. Morris, C. R. Beverly, J. J. Collopy, T. G. Baker, and P. N. Lane. Validation of plantation transpiration in south-eastern australia estimated using the 3pg+ forest growth model. *Forest Ecology and Management*, 260(5):663–678, 2010.
- S. F. d. B. Ferraz, C. B. Rodrigues, L. G. Garcia, C. A. Alvares, and W. d. P. Lima. Effects of eucalyptus plantations on streamflow in brazil: Moving beyond the water use debate. *Forest Ecology and Management*, 453:117571, 2019. ISSN 0378-1127. doi: https://doi.org/10. 1016/j.foreco.2019.117571.
- T. Foken, R. Leuning, S. R. Oncley, M. Mauder, and M. Aubinet. *Corrections and Data Quality Control*, pages 85–131. Springer Netherlands, Dordrecht, 2012. ISBN 978-94-007-2351-1. doi: 10.1007/978-94-007-2351-1_4.

- D. I. Forrester, J. Bauhus, and A. L. Cowie. Carbon allocation in a mixed-species plantation of eucalyptus globulus and acacia mearnsii. *Forest Ecology and Management*, 233(2-3): 275–284, 2006. ISSN 03781127 (ISSN). doi: 10.1016/j.foreco.2006.05.018.
- D. I. Forrester, J. J. Collopy, and J. D. Morris. Transpiration along an age series of eucalyptus globulus plantations in southeastern australia. *Forest Ecology and Management*, 259(9): 1754–1760, 2010a. ISSN 0378-1127. doi: https://doi.org/10.1016/j.foreco.2009.04.023.
- D. I. Forrester, S. Theiveyanathan, J. J. Collopy, and N. E. Marcar. Enhanced water use efficiency in a mixed eucalyptus globulus and acacia mearnsii plantation. *Forest Ecology and Management*, 259(9):1761–1770, 2010b. ISSN 0378-1127. doi: https://doi.org/10.1016/j. foreco.2009.07.036.
- P. J. Franks, P. L. Drake, and R. H. Froend. Anisohydric but isohydrodynamic: seasonally constant plant water potential gradient explained by a stomatal control mechanism incorporating variable plant hydraulic conductance. *Plant, Cell and Environment*, 30(1):19–30, jan 2007. ISSN 0140-7791. doi: 10.1111/j.1365-3040.2006.01600.x.
- L. H. Fraser. Try—a plant trait database of databases. *Global Change Biology*, 26(1):189–190, dec 2020. ISSN 1354-1013. doi: 10.1111/gcb.14869.
- T. Fruh and W. Kurth. The hydraulic system of trees: Theoretical framework and numerical simulation. *Journal of Theoretical Biology*, 201(4):251–270, dec 1999. ISSN 0022-5193. doi: 10.1006/jtbi.1999.1028.
- W. R. Gardner. Dynamic aspects of water availability to plants. *Soil Science*, 89(2):63–73, feb 1960. ISSN 0038-075X. doi: 10.1097/00010694-196002000-00001.
- J. L. M. Gonçalves, C. A. Alvares, J. H. T. Rocha, C. B. Brandani, and R. Hakamada. Eucalypt plantation management in regions with water stress. *Southern Forests: a Journal of Forest Science*, 79(3):169–183, may 2017. ISSN 2070-2620. doi: 10.2989/20702620.2016.1255415.
- A. Granier, N. Bréda, B. Longdoz, P. Gross, and J. Ngao. Ten years of fluxes and stand growth in a young beech forest at hesse, north-eastern france. *Annals of Forest Science*, 65(7): 704–704, jan 2008. ISSN 12864560 (ISSN). doi: 10.1051/forest:2008052.
- A. Greenwood. The application and limitations of using mean annual evapotranspirationand rainfall in the assessment of the impact of plantation forestry on runoff. Technical report, Department of Water, Land and Biodiversity Conservation Technical note 2007/14, 2007.
- A. J. B. Greenwood. The first stages of australian forest water regulation: National reform and regional implementation. *Environmental Science & Policy*, 29:124–136, may 2013. ISSN 1462-9011. doi: 10.1016/j.envsci.2013.01.012.

- A. Griebel, L. T. Bennett, D. Metzen, E. Pendall, P. N. J. Lane, and S. K. Arndt. Trading water for carbon: Maintaining photosynthesis at the cost of increased water loss during high temperatures in a temperate forest. *Journal of Geophysical Research: Biogeosciences*, 125(1), 2020. doi: 10.1029/2019JG005239.
- M. B. Gush, D. F. Scott, G. P. W. Jewitt, R. E. Schulze, L. A. Hallowes, and A. H. M. Görgens. A new approach to modelling streamflow reductions resulting from commercial afforestation in south africa. *South African Forestry Journal*, 196(1):27–36, nov 2002. ISSN 1029-5925. doi: 10.1080/20702620.2002.10434615.
- S. K. Hamilton, M. Z. Hussain, C. Lowrie, B. Basso, and G. P. Robertson. Evapotranspiration is resilient in the face of land cover and climate change in a humid temperate catchment. *Hydrological Processes*, 32(5):655–663, 2018. ISSN 0885-6087. doi: 10.1002/hyp.11447.
- R. J. Harper, S. J. Sochacki, K. R. J. Smettem, and N. Robinson. Managing water in agricultural landscapes with short-rotation biomass plantations. *GCB Bioenergy*, 6(5):544–555, aug 2014. ISSN 1757-1693. doi: 10.1111/gcbb.12090.
- S. Hartzell, M. S. Bartlett, and A. Porporato. The role of plant water storage and hydraulic strategies in relation to soil moisture availability. *Plant and Soil*, 419(1-2):503–521, 2017. ISSN 0032-079X. doi: 10.1007/s11104-017-3341-7.
- S. Hartzell, M. S. Bartlett, and A. Porporato. Unified representation of the c3, c4, and cam photosynthetic pathways with the photo3 model. *Ecological Modelling*, 384:173–187, sep 2018. ISSN 0304-3800. doi: 10.1016/j.ecolmodel.2018.06.012.
- L. He, J. M. Chen, Y. Pan, R. Birdsey, and J. Kattge. Relationships between net primary productivity and forest stand age in u.s. forests. *Global Biogeochemical Cycles*, 26(3), aug 2012. ISSN 0886-6236. doi: 10.1029/2010GB003942.
- M. M. P. D. Heijmans, W. J. Arp, and F. S. Chapin. Carbon dioxide and water vapour exchange from understory species in boreal forest. *Agricultural and Forest Meteorology*, 123(3-4):135– 147, jun 2004. ISSN 0168-1923. doi: 10.1016/j.agrformet.2003.12.006.
- W. N. Herkelrath, E. E. Miller, and W. R. Gardner. Water uptake by plants: li. the root contact model. *Soil Science Society of America Journal*, 41(6):1039–1043, nov 1977. ISSN 0361-5995. doi: 10.2136/sssaj1977.03615995004100060004x.
- N. Hinko-Najera, P. Isaac, J. Beringer, E. van Gorsel, C. Ewenz, I. McHugh, J.-F. Exbrayat, S. J. Livesley, and S. K. Arndt. Net ecosystem carbon exchange of a dry temperate eucalypt forest. *Biogeosciences*, 14(16):3781–3800, aug 2017. ISSN 1726-4189. doi: 10.5194/ bg-14-3781-2017.

- C. W. Huang, J. C. Domec, E. J. Ward, T. Duman, G. Manoli, A. J. Parolari, and G. G. Katul. The effect of plant water storage on water fluxes within the coupled soil–plant system. *New Phytologist*, 213(3):1093–1106, nov 2017. ISSN 0028-646X. doi: 10.1111/nph.14273.
- R. M. Hubbard, M. G. Ryan, C. P. Giardina, and H. Barnard. The effect of fertilization on sap flux and canopy conductance in a eucalyptus saligna experimental forest. *Global Change Biology*, 10(4):427–436, 2004. ISSN 1354-1013. doi: 10.1111/j.1529-8817.2003.00741.x.
- R. M. Hubbard, J. Stape, M. G. Ryan, A. C. Almeida, and J. Rojas. Effects of irrigation on water use and water use efficiency in two fast growing eucalyptus plantations. *Forest Ecology and Management*, 259(9):1714–1721, apr 2010. ISSN 0378-1127. doi: 10.1016/j.foreco.2009.10. 028.
- J. Irvine, B. E. Law, P. M. Anthoni, and F. C. Meinzer. Water limitations to carbon exchange in old-growth and young ponderosa pine stands. *Tree Physiology*, 22(2-3):189–196, feb 2002. ISSN 0829-318X. doi: 10.1093/treephys/22.2-3.189.
- J. Irvine, B. E. Law, M. R. Kurpius, P. M. Anthoni, D. Moore, and P. A. Schwarz. Age-related changes in ecosystem structure and function and effects on water and carbon exchange in ponderosa pine. *Tree Physiology*, 24(7):753–763, jul 2004. ISSN 0829-318X. doi: 10.1093/ treephys/24.7.753.
- R. B. Jackson. Trading water for carbon with biological carbon sequestration. *Science*, 310 (5756):1944–1947, dec 2005. ISSN 0036-8075. doi: 10.1126/science.1119282.
- M. Janott, S. Gayler, A. Gessler, M. Javaux, C. Klier, and E. Priesack. A one-dimensional model of water flow in soil-plant systems based on plant architecture. *Plant and Soil*, 341 (1-2):233–256, nov 2011. ISSN 0032-079X. doi: 10.1007/s11104-010-0639-0.
- N. J. Jarvis. The interpretation of the variations in leaf water potential and stomatal conductance found in canopies in the field. *Philosophical Transactions of the Royal Society of London. B, Biological Sciences*, 273(927):593–610, feb 1976. ISSN 0080-4622. doi: 10.1098/rstb.1976. 0035.
- N. J. Jarvis. A simple empirical model of root water uptake. *Journal of Hydrology*, 107(1-4): 57–72, may 1989. ISSN 0022-1694. doi: 10.1016/0022-1694(89)90050-4.
- R. S. Jassal, T. A. Black, D. L. Spittlehouse, C. Brümmer, and Z. Nesic. Evapotranspiration and water use efficiency in different-aged pacific northwest douglas-fir stands. *Agricultural and Forest Meteorology*, 149(6-7):1168–1178, jun 2009. ISSN 0168-1923. doi: 10.1016/j. agrformet.2009.02.004.
- S. Jiang, C. Liang, N. Cui, L. Zhao, C. Liu, Y. Feng, X. Hu, D. Gong, and Q. Zou. Water use efficiency and its drivers in four typical agroecosystems based on flux tower measurements.

Agricultural and Forest Meteorology, 295:108200, 2020. ISSN 0168-1923. doi: https://doi. org/10.1016/j.agrformet.2020.108200.

- H. G. Jones. *Plants and Microclimate*. Cambridge University Press, 2009. doi: 10.1017/ cbo9780511845727. cited By 980.
- H. G. Jones. *Plants and microclimate: A quantitative approach to environmental plant physiology*. Plants and Microclimate: A Quantitative Approach to Environmental Plant Physiology. Cambridge University Press, 2013. doi: 10.1017/CBO9780511845727.
- R. J. Keenan, M. Parsons, A. Gerrand, E. O'Loughlin, S. Beavis, D. Gunawardana, M. Gavran, and A. Bugg. Plantations and water use: a review prepared for the forest and wood products, research and development corporation. Technical report, Bureau of Rural Sciences, 2004.
- D. Kennedy, S. Swenson, K. W. Oleson, D. M. Lawrence, R. Fisher, A. C. L. da Costa, and P. Gentine. Implementing plant hydraulics in the community land model, version 5. *Journal of Advances in Modeling Earth Systems*, 11(2):485–513, 2019. ISSN 1942-2466. doi: 10. 1029/2018ms001500.
- J. Kennedy and R. Eberhart. Particle swarm optimization. In *Proceedings of ICNN'95 International Conference on Neural Networks*, volume 4, pages 1942–1948 vol.4, 1995. doi: 10.1109/ICNN.1995.488968.
- P. Krishnan, T. A. Black, R. S. Jassal, B. Chen, and Z. Nesic. Interannual variability of the carbon balance of three different-aged douglas-fir stands in the pacific northwest. *Journal* of Geophysical Research: Biogeosciences, 114(4), 2009. ISSN 01480227 (ISSN). doi: 10.1029/2008JG000912.
- T. Kumagai. Modeling water transportation and storage in sapwood model development and validation. *Agricultural and Forest Meteorology*, 109(2):105–115, aug 2001. ISSN 0168-1923. doi: 10.1016/s0168-1923(01)00261-1.
- T. Kume, K. Tsuruta, H. Komatsu, T. Kumagai, N. Higashi, Y. Shinohara, and K. Otsuki. Effects of sample size on sap flux-based stand-scale transpiration estimates. *Tree Physiology*, 30 (1):129–138, 2010. doi: 10.1093/treephys/tpp074.
- W. L. Kutsch, M. Aubinet, N. Buchmann, P. Smith, B. Osborne, W. Eugster, M. Wattenbach, M. Schrumpf, E. D. Schulze, E. Tomelleri, E. Ceschia, C. Bernhofer, P. Béziat, A. Carrara, P. D. Tommasi, T. Grünwald, M. Jones, V. Magliulo, O. Marloie, C. Moureaux, A. Olioso, M. J. Sanz, M. Saunders, H. Søgaard, and W. Ziegler. The net biome production of full crop rotations in europe. *Agriculture, Ecosystems & Environment*, 139(3):336–345, nov 2010. ISSN 0167-8809. doi: 10.1016/j.agee.2010.07.016.

- H. Kwon, B. E. Law, C. K. Thomas, and B. G. Johnson. The influence of hydrological variability on inherent water use efficiency in forests of contrasting composition, age, and precipitation regimes in the pacific northwest. *Agricultural and Forest Meteorology*, 249:488–500, feb 2018. ISSN 0168-1923. doi: 10.1016/j.agrformet.2017.08.006.
- B. Lalic and D. T. Mihailovic. An empirical relation describing leaf-area density inside the forest for environmental modeling. *Journal of Applied Meteorology*, 43(4):641–645, apr 2004. ISSN 0894-8763. doi: 10.1175/1520-0450(2004)043<0641:AERDLD>2.0.CO;2.
- J. J. Landsberg and R. H. Waring. A generalised model of forest productivity using simplified concepts of radiation-use efficiency, carbon balance and partitioning. *Forest Ecology and Management*, 95(3):209–228, aug 1997. ISSN 0378-1127. doi: 10.1016/s0378-1127(97) 00026-1.
- G. Lasslop, M. Reichstein, D. Papale, A. D. Richardson, A. Arneth, A. Barr, P. Stoy, and G. Wohlfahrt. Separation of net ecosystem exchange into assimilation and respiration using a light response curve approach: critical issues and global evaluation. *Global Change Biology*, 16 (1):187–208, 2010. ISSN 1354-1013. doi: https://doi.org/10.1111/j.1365-2486.2009.02041.x.
- B. E. Law, D. D. Baldocchi, and P. M. Anthoni. Below-canopy and soil co2 fluxes in a ponderosa pine forest. *Agricultural and Forest Meteorology*, 94(3-4):171–188, may 1999. ISSN 0168-1923. doi: 10.1016/s0168-1923(99)00019-2.
- B. E. Law, A. H. Goldstein, P. M. Anthoni, M. H. Unsworth, J. A. Panek, M. R. Bauer, J. M. Fracheboud, and N. Hultman. Carbon dioxide and water vapor exchange by young and old ponderosa pine ecosystems during a dry summer. *Tree Physiology*, 21(5):299–308, mar 2001. ISSN 0829-318X. doi: 10.1093/treephys/21.5.299.
- X. Lee, J. Finnigan, and K. T. Paw U. Coordinate Systems and Flux Bias Error, pages 33–66. Springer Netherlands, Dordrecht, 2005. ISBN 978-1-4020-2265-4. doi: 10.1007/ 1-4020-2265-4_3.
- R. Leuning. Modelling stomatal behaviour and and photosynthesis of eucalyptus grandis. *Functional Plant Biology*, 17(2):159, 1990. ISSN 1445-4408. doi: 10.1071/pp9900159.
- R. Leuning. A critical appraisal of a combined stomatal-photosynthesis model for c3 plants. *Plant, Cell and Environment*, 18(4):339–355, apr 1995. ISSN 0140-7791. doi: 10.1111/j. 1365-3040.1995.tb00370.x.
- R. Leuning, E. van Gorsel, W. J. Massman, and P. R. Isaac. Reflections on the surface energy imbalance problem. *Agricultural and Forest Meteorology*, 156:65–74, apr 2012. ISSN 0168-1923. doi: 10.1016/j.agrformet.2011.12.002.

- D. Li, S. Niu, and Y. Luo. Global patterns of the dynamics of soil carbon and nitrogen stocks following afforestation: a meta-analysis. *New Phytologist*, 195(1):172–181, apr 2012. ISSN 0028-646X. doi: 10.1111/j.1469-8137.2012.04150.x.
- L. Li, Z.-L. Yang, A. M. Matheny, H. Zheng, S. C. Swenson, D. M. Lawrence, M. Barlage, B. Yan, N. G. McDowell, and L. R. Leung. Representation of plant hydraulics in the noah-mp land surface model: Model development and multiscale evaluation. *Journal of Advances in Modeling Earth Systems*, 13(4):e2020MS002214, 2021.
- M. Littleboy, D. Silburn, D. Freebairn, D. Woodruff, and G. Hammer. Perfect, a computer simulation model of productivity, erosion, runoff functions to evaluate conservation techniques. Technical report, Queensland Department of PrimaryIndustries, 1989.
- W. Liu, J. Wu, H. Fan, H. Duan, Q. Li, Y. Yuan, and H. Zhang. Estimations of evapotranspiration in an age sequence of eucalyptus plantations in subtropical china. *PLOS ONE*, 12(4): e0174208, 2017. doi: 10.1371/journal.pone.0174208.
- Y. Liu, M. Kumar, G. G. Katul, X. Feng, and A. G. Konings. Plant hydraulics accentuates the effect of atmospheric moisture stress on transpiration. *Nature Climate Change*, 10(7):691– 695, jun 2020. ISSN 1758-678X. doi: 10.1038/s41558-020-0781-5.
- Y. Luo and X. Zhou. Chapter 2 importance and roles of soil respiration. In Y. Luo and X. Zhou, editors, *Soil Respiration and the Environment*, pages 17 32. Academic Press, Burlington, 2006. ISBN 978-0-12-088782-8. doi: https://doi.org/10.1016/B978-012088782-8/50002-4.
- J. Ma, T. Zha, X. Jia, Y. Tian, C. P. A. Bourque, P. Liu, Y. Bai, Y. Wu, C. Ren, H. Yu, F. Zhang, C. Zhou, and W. Chen. Energy and water vapor exchange over a young plantation in northern china. *Agricultural and Forest Meteorology*, 263:334–345, dec 2018. ISSN 0168-1923. doi: 10.1016/j.agrformet.2018.09.004.
- J. Ma, X. Jia, T. Zha, C. P.-A. Bourque, Y. Tian, Y. Bai, P. Liu, R. Yang, C. Li, C. Li, J. Xie, H. Yu, F. Zhang, and C. Zhou. Ecosystem water use efficiency in a young plantation in northern china and its relationship to drought. *Agricultural and Forest Meteorology*, 275:1–10, sep 2019. ISSN 0168-1923. doi: 10.1016/j.agrformet.2019.05.004.
- A. Mahabbati, J. Beringer, M. Leopold, I. McHugh, J. Cleverly, P. Isaac, and A. Izady. A comparison of gap-filling algorithms for eddy covariance fluxes and their drivers. *Geoscientific Instrumentation, Methods and Data Systems*, 10(1):123–140, 2021.
- G. Manoli, S. Bonetti, J.-C. Domec, M. Putti, G. Katul, and M. Marani. Tree root systems competing for soil moisture in a 3d soil–plant model. *Advances in Water Resources*, 66: 32–42, apr 2014. ISSN 0309-1708. doi: 10.1016/j.advwatres.2014.01.006.

- S. Manzoni, G. Vico, G. Katul, S. Palmroth, R. B. Jackson, and A. Porporato. Hydraulic limits on maximum plant transpiration and the emergence of the safety–efficiency trade-off. *New Phytologist*, 198(1):169–178, 2013a.
- S. Manzoni, G. Vico, A. Porporato, and G. Katul. Biological constraints on water transport in the soil-plant-atmosphere system. *Advances in Water Resources*, 51:292–304, 2013b. doi: 10.1016/j.advwatres.2012.03.016.
- V. Marchionni, A. Guyot, N. Tapper, J. P. Walker, and E. Daly. Water balance and tree water use dynamics in remnant urban reserves. *Journal of Hydrology*, 575:343–353, aug 2019. ISSN 0022-1694. doi: 10.1016/j.jhydrol.2019.05.022.
- B. Marcolla, A. Cescatti, G. Manca, R. Zorer, M. Cavagna, A. Fiora, D. Gianelle, M. Rodeghiero, M. Sottocornola, and R. Zampedri. Climatic controls and ecosystem responses drive the inter-annual variability of the net ecosystem exchange of an alpine meadow. *Agricultural and Forest Meteorology*, 151(9):1233–1243, sep 2011. ISSN 0168-1923. doi: https://doi.org/10.1016/j.agrformet.2011.04.015.
- A. M. Matheny, G. Bohrer, S. R. Garrity, T. H. Morin, C. J. Howard, and C. S. Vogel. Observations of stem water storage in trees of opposing hydraulic strategies. *Ecosphere*, 6(9):art165, sep 2015. ISSN 2150-8925. doi: 10.1890/es15-00170.1.
- A. M. Matheny, G. Mirfenderesgi, and G. Bohrer. Trait-based representation of hydrological functional properties of plants in weather and ecosystem models. *Plant Diversity*, 39(1): 1–12, feb 2017. ISSN 2468-2659. doi: https://doi.org/10.1016/j.pld.2016.10.001.
- I. D. McHugh, J. Beringer, S. C. Cunningham, P. J. Baker, T. R. Cavagnaro, R. MacNally, and R. M. Thompson. Interactions between nocturnal turbulent flux, storage and advection at an "ideal" eucalypt woodland site. *Biogeosciences*, 14(12):3027–3050, 2017. doi: 10.5194/ bg-14-3027-2017.
- D. Medvigy, S. C. Wofsy, J. W. Munger, D. Y. Hollinger, and P. R. Moorcroft. Mechanistic scaling of ecosystem function and dynamics in space and time: Ecosystem demography model version 2. *Journal of Geophysical Research*, 114(G1), 2009. ISSN 0148-0227. doi: 10.1029/2008jg000812.
- M. Mencuccini, S. Manzoni, and B. Christoffersen. Modelling water fluxes in plants: from tissues to biosphere. *New Phytologist*, 222(3):1207–1222, feb 2019. doi: 10.1111/nph.15681.
- M. Mendel, S. Hergarten, and H. J. Neugebauer. On a better understanding of hydraulic lift: A numerical study. *Water Resources Research*, 38(10):1–1–1–10, oct 2002. ISSN 0043-1397. doi: 10.1029/2001wr000911.

- K. R. Mendes, S. Campos, L. L. da Silva, P. R. Mutti, R. R. Ferreira, S. S. Medeiros, A. M. Perez-Marin, T. V. Marques, T. M. Ramos, M. M. de Lima Vieira, C. P. Oliveira, W. A. Gonçalves, G. B. Costa, A. C. D. Antonino, R. S. C. Menezes, B. G. Bezerra, and C. M. S. e Silva. Seasonal variation in net ecosystem co2 exchange of a brazilian seasonally dry tropical forest. *Scientific Reports*, 10(1), jun 2020. ISSN 2045-2322. doi: 10.1038/s41598-020-66415-w.
- F. Meunier, Y. Rothfuss, T. Bariac, P. Biron, P. Richard, J.-L. Durand, V. Couvreur, J. Vanderborght, and M. Javaux. Measuring and modeling hydraulic lift of lolium multiflorum using stable water isotopes. *Vadose Zone Journal*, 17(1):160134, may 2018. ISSN 1539-1663. doi: 10.2136/vzj2016.12.0134.
- P. Miehle, S. J. Livesley, P. M. Feikema, C. Li, and S. K. Arndt. Assessing productivity and carbon sequestration capacity of eucalyptus globulus plantations using the process model forest-dndc: Calibration and validation. *Ecological Modelling*, 192(1-2):83–94, 2006. ISSN 03043800 (ISSN). doi: 10.1016/j.ecolmodel.2005.07.021.
- M. Migliavacca, T. Musavi, M. D. Mahecha, J. A. Nelson, J. Knauer, D. D. Baldocchi, O. Perez-Priego, R. Christiansen, J. Peters, K. Anderson, M. Bahn, T. A. Black, P. D. Blanken, D. Bonal, N. Buchmann, S. Caldararu, A. Carrara, N. Carvalhais, A. Cescatti, J. Chen, J. Cleverly, E. Cremonese, A. R. Desai, T. S. El-Madany, M. M. Farella, M. Fernández-Martínez, G. Filippa, M. Forkel, M. Galvagno, U. Gomarasca, C. M. Gough, M. Göckede, A. Ibrom, H. Ikawa, I. A. Janssens, M. Jung, J. Kattge, T. F. Keenan, A. Knohl, H. Kobayashi, G. Kraemer, B. E. Law, M. J. Liddell, X. Ma, I. Mammarella, D. Martini, C. Macfarlane, G. Matteucci, L. Montagnani, D. E. Pabon-Moreno, C. Panigada, D. Papale, E. Pendall, J. Penuelas, R. P. Phillips, P. B. Reich, M. Rossini, E. Rotenberg, R. L. Scott, C. Stahl, U. Weber, G. Wohlfahrt, S. Wolf, I. J. Wright, D. Yakir, S. Zaehle, and M. Reichstein. The three major axes of terrestrial ecosystem function. *Nature*, 2021. ISSN 1476-4687. doi: 10.1038/s41586-021-03939-9.
- G. Mirfenderesgi, G. Bohrer, A. M. Matheny, S. Fatichi, R. P. de Moraes Frasson, and K. V. R. Schäfer. Tree level hydrodynamic approach for resolving aboveground water storage and stomatal conductance and modeling the effects of tree hydraulic strategy. *Journal of Geophysical Research: Biogeosciences*, 121(7):1792–1813, jul 2016. ISSN 2169-8953. doi: 10.1002/2016jg003467.
- G. Mirfenderesgi, A. M. Matheny, and G. Bohrer. Hydrodynamic trait coordination and costbenefit trade-offs throughout the isohydric-anisohydric continuum in trees. *Ecohydrology*, 12 (1):e2041, sep 2018. ISSN 1936-0584. doi: 10.1002/eco.2041.
- M. T. Moroni, D. Worledge, and C. L. Beadle. Root distribution of eucalyptus nitens and e. globulus in irrigated and droughted soil. *Forest Ecology and Management*, 177(1-3):399–407, apr 2003. ISSN 0378-1127. doi: 10.1016/s0378-1127(02)00410-3.

- P. S. Nobel. *Chapter 9 Plants and Fluxes*, pages 438–505. Academic Press, San Diego, 2009. ISBN 978-0-12-374143-1. doi: https://doi.org/10.1016/B978-0-12-374143-1.00009-0.
- M. D. Nosetto, E. G. Jobbagy, and J. M. Paruelo. Land-use change and water losses: the case of grassland afforestation across a soil textural gradient in central argentina. *Global Change Biology*, 11(7):1101–1117, jul 2005. ISSN 1354-1013. doi: 10.1111/j.1365-2486.2005.00975.x.
- M. D. Nosetto, E. Luna Toledo, P. N. Magliano, P. Figuerola, L. J. Blanco, and E. G. Jobbágy. Contrasting co2 and water vapour fluxes in dry forest and pasture sites of central argentina. *Ecohydrology*, 13(8), 2020. ISSN 19360584 (ISSN). doi: 10.1002/eco.2244.
- Z. Ntshidi, S. Dzikiti, D. Mazvimavi, and N. T. Mobe. Contribution of understorey vegetation to evapotranspiration partitioning in apple orchards under mediterranean climatic conditions in south africa. *Agricultural Water Management*, 245:106627, feb 2021. ISSN 0378-3774. doi: 10.1016/j.agwat.2020.106627.
- A. P. O'Grady, D. Eamus, and L. B. Hutley. Transpiration increases during the dry season: patterns of tree water use in eucalypt open-forests of northern australia. *Tree Physiology*, 19 (9):591–597, jul 1999. ISSN 0829-318X. doi: 10.1093/treephys/19.9.591.
- C. Pappas, S. Fatichi, and P. Burlando. Modeling terrestrial carbon and water dynamics across climatic gradients: Does plant trait diversity matter? *New Phytologist*, 209(1):137–151, 2016. doi: 10.1111/nph.13590.
- G. Pastorello, C. Trotta, E. Canfora, H. Chu, D. Christianson, Y. W. Cheah, C. Poindexter, J. Chen, A. Elbashandy, M. Humphrey, P. Isaac, D. Polidori, A. Ribeca, C. van Ingen, L. Zhang, B. Amiro, C. Ammann, M. A. Arain, J. Ardö, T. Arkebauer, S. K. Arndt, N. Arriga, M. Aubinet, M. Aurela, D. Baldocchi, A. Barr, E. Beamesderfer, L. B. Marchesini, O. Bergeron, J. Beringer, C. Bernhofer, D. Berveiller, D. Billesbach, T. A. Black, P. D. Blanken, G. Bohrer, J. Boike, P. V. Bolstad, D. Bonal, J. M. Bonnefond, D. R. Bowling, R. Bracho, J. Brodeur, C. Brümmer, N. Buchmann, B. Burban, S. P. Burns, P. Buysse, P. Cale, M. Cavagna, P. Cellier, S. Chen, I. Chini, T. R. Christensen, J. Cleverly, A. Collalti, C. Consalvo, B. D. Cook, D. Cook, C. Coursolle, E. Cremonese, P. S. Curtis, E. D'Andrea, H. da Rocha, X. Dai, K. J. Davis, B. De Cinti, A. de Grandcourt, A. De Ligne, R. C. De Oliveira, N. Delpierre, A. R. Desai, C. M. Di Bella, P. di Tommasi, H. Dolman, F. Domingo, G. Dong, S. Dore, P. Duce, E. Dufrêne, A. Dunn, J. Dušek, D. Eamus, U. Eichelmann, H. A. M. ElKhidir, W. Eugster, C. M. Ewenz, B. Ewers, D. Famulari, S. Fares, I. Feigenwinter, A. Feitz, R. Fensholt, G. Filippa, M. Fischer, J. Frank, M. Galvagno, M. Gharun, D. Gianelle, B. Gielen, et al. The fluxnet2015 dataset and the oneflux processing pipeline for eddy covariance data. Scientific data, 7(1):225, 2020. doi: 10.1038/s41597-020-0534-3.
- J. G. Pausas, R. B. Pratt, J. E. Keeley, A. L. Jacobsen, A. R. Ramirez, A. Vilagrosa, S. Paula, I. N. Kaneakua-Pia, and S. D. Davis. Towards understanding resprouting at the global scale.

New Phytologist, 209(3):945–954, 2016. ISSN 0028-646X. doi: https://doi.org/10.1111/nph.13644.

- M. Peichl, M. A. Arain, and J. J. Brodeur. Age effects on carbon fluxes in temperate pine forests. *Agricultural and Forest Meteorology*, 150(7-8):1090–1101, jul 2010. ISSN 0168-1923. doi: 10.1016/j.agrformet.2010.04.008.
- J. M. C. Pereira, M. Tome, J. M. B. Carreiras, J. A. Tome, J. S. Pereira, J. S. David, and A. M. D. Fabiao. Leaf area estimation from tree allometrics in eucalyptus globulus plantations. *Canadian Journal of Forest Research*, 27(2):166–173, 1997. doi: 10.1139/x96-179.
- J. S. Pereira, J. D. Tenhunen, O. L. Lange, W. Beyschlag, A. Meyer, and M. M. David. Seasonal and diurnal patterns in leaf gas exchange of eucalyptus globulus trees growing in portugal. *Canadian Journal of Forest Research*, 16(2):177–184, 1986. ISSN 00455067 (ISSN). doi: 10.1139/x86-033.
- J. S. Pereira, J. A. Mateus, L. M. Aires, G. Pita, C. Pio, J. S. David, V. Andrade, J. Banza, T. S. David, T. A. Paço, and A. Rodrigues. Net ecosystem carbon exchange in three contrasting mediterranean ecosystems the effect of drought. *Biogeosciences*, 4(5):791–802, sep 2007. ISSN 1726-4189. doi: 10.5194/bg-4-791-2007.
- G. Pita, B. Gielen, D. Zona, A. Rodrigues, S. Rambal, I. A. Janssens, and R. Ceulemans. Carbon and water vapor fluxes over four forests in two contrasting climatic zones. *Agricultural and Forest Meteorology*, 180:211–224, oct 2013. ISSN 0168-1923. doi: https://doi.org/10. 1016/j.agrformet.2013.06.003.
- A. D. Polyanin. *Handbook of Linear Partial Differential Equations for Engineers and Scientists*. Chapman and Hall/CRC, nov 2001. doi: 10.1201/9781420035322.
- K. S. Pregitzer and E. S. Euskirchen. Carbon cycling and storage in world forests: biome patterns related to forest age. *Global Change Biology*, 10(12):2052–2077, oct 2004. ISSN 1354-1013. doi: 10.1111/j.1365-2486.2004.00866.x.
- R. Qiu and G. G. Katul. Maximizing leaf carbon gain in varying saline conditions: An optimization model with dynamic mesophyll conductance. *The Plant Journal*, 101(3):543–554, nov 2020. ISSN 0960-7412. doi: 10.1111/tpj.14553.
- J. C. Quijano and P. Kumar. Numerical simulations of hydraulic redistribution across climates: The role of the root hydraulic conductivities. *Water Resources Research*, 51(10):8529–8550, 2015. ISSN 0043-1397. doi: 10.1002/2014WR016509.
- M. Reichstein, E. Falge, D. Baldocchi, D. Papale, M. Aubinet, P. Berbigier, C. Bernhofer, N. Buchmann, T. Gilmanov, A. Granier, T. Grünwald, K. Havránková, H. Ilvesniemi, D. Janous, A. Knohl, T. Laurila, A. Lohila, D. Loustau, G. Matteucci, T. Meyers, F. Miglietta, J.-M.

Ourcival, J. Pumpanen, S. Rambal, E. Rotenberg, M. Sanz, J. Tenhunen, G. Seufert, F. Vaccari, T. Vesala, D. Yakir, and R. Valentini. On the separation of net ecosystem exchange into assimilation and ecosystem respiration: review and improved algorithm. *Global Change Biology*, **11**(9):1424–1439, 2005a. ISSN 1354-1013. doi: https://doi.org/10.1111/j.1365-2486. 2005.001002.x.

- M. Reichstein, T. Kätterer, O. Andrén, P. Ciais, E.-D. Schulze, W. Cramer, D. Papale, and R. Valentini. Temperature sensitivity of decomposition in relation to soil organic matter pools: critique and outlook. *Biogeosciences*, 2(4):317–321, nov 2005b. ISSN 1726-4189. doi: 10.5194/bg-2-317-2005.
- M. G. Reis, A. Ribeiro, E. E. N. Alves, Y. P. Rody, R. A. Loos, A. A. Vasconcelos, and W. L. Araújo. Maximum co2 assimilation in young eucalyptus plantations is higher than in brazilian savanna trees during dry field seasons. *Trees Structure and Function*, 2019. doi: 10.1007/s00468-018-1800-2.
- A. A. Renchon, A. Griebel, D. Metzen, C. A. Williams, B. Medlyn, R. A. Duursma, C. V. M. Barton, C. Maier, M. M. Boer, P. Isaac, D. Tissue, V. Resco De Dios, and E. Pendall. Upside-down fluxes down under: Co2 net sink in winter and net source in summer in a temperate ever-green broadleaf forest. *Biogeosciences*, 15(12):3703–3716, 2018. ISSN 17264170 (ISSN). doi: 10.5194/bg-15-3703-2018.
- P. Rengasamy. World salinization with emphasis on australia. *Journal of Experimental Botany*, 57(5):1017–1023, mar 2006. ISSN 1460-2431. doi: 10.1093/jxb/erj108.
- A. Rodrigues, G. Pita, J. Mateus, C. Kurz-Besson, M. Casquilho, S. Cerasoli, A. Gomes, and J. Pereira. Eight years of continuous carbon fluxes measurements in a portuguese eucalypt stand under two main events: Drought and felling. *Agricultural and Forest Meteorology*, 151 (4):493–507, 2011. ISSN 0168-1923. doi: https://doi.org/10.1016/j.agrformet.2010.12.007.
- M. G. Ryan, R. M. Hubbard, S. Pongracic, R. J. Raison, and R. E. McMurtrie. Foliage, fineroot, woody-tissue and stand respiration in pinus radiata in relation to nitrogen status. *Tree Physiology*, 16(3):333–343, mar 1996. ISSN 0829-318X. doi: 10.1093/treephys/16.3.333.
- M. G. Ryan, D. Binkley, J. H. Fownes, C. P. Giardina, and R. S. Senock. An experimental test of the causes of forest growth decline with stand age. *Ecological Monographs*, 74(3):393–414, aug 2004. ISSN 00129615 (ISSN). doi: 10.1890/03-4037.
- M. G. Ryan, J. L. Stape, D. Binkley, S. Fonseca, R. A. Loos, E. N. Takahashi, C. R. Silva, S. R. Silva, R. E. Hakamada, J. M. Ferreira, A. M. N. Lima, J. L. Gava, F. P. Leite, H. B. Andrade, J. M. Alves, and G. G. C. Silva. Factors controlling eucalyptus productivity: How water availability and stand structure alter production and carbon allocation. *Forest Ecology and Management*, 259(9):1695–1703, apr 2010. ISSN 03781127 (ISSN). doi: 10.1016/j. foreco.2010.01.013.

- E. Saadaoui, K. B. Yahia, S. Dhahri, M. L. B. Jamaa, and M. L. Khouja. An overview of adaptative responses to drought stress in eucalyptus spp. *Forestry Studies*, 67(1):86–96, dec 2017. ISSN 1736-8723. doi: 10.1515/fsmu-2017-0014.
- P. J. Sands and J. J. Landsberg. Parameterisation of 3-pg for plantation grown eucalyptus globulus. *Forest Ecology and Management*, 163(1-3):273–292, jun 2002. ISSN 0378-1127. doi: 10.1016/s0378-1127(01)00586-2.
- K. Scheerlinck, V. R. N. Pauwels, H. Vernieuwe, and B. D. Baets. Calibration of a water and energy balance model: Recursive parameter estimation versus particle swarm optimization. *Water Resources Research*, 45(10), oct 2009. ISSN 0043-1397. doi: 10.1029/2009wr008051.
- D. F. Scott. On the hydrology of industrial timber plantations. *Hydrological Processes*, 19(20): 4203–4206, 2005. ISSN 0885-6087. doi: 10.1002/hyp.6104.
- R. H. Shaw and U. Schumann. Large-eddy simulation of turbulent flow above and within a forest. *Boundary-Layer Meteorology*, 61(1-2):47–64, oct 1992. ISSN 0006-8314. doi: 10. 1007/bf02033994.
- R. Skubel, M. A. Arain, M. Peichl, J. J. Brodeur, M. Khomik, R. Thorne, J. Trant, and M. Kula. Age effects on the water-use efficiency and water-use dynamics of temperate pine plantation forests. *Hydrological Processes*, 29(18):4100–4113, 2015. doi: 10.1002/hyp.10549.
- P. Smith, G.-J. Nabuurs, I. A. Janssens, S. Reis, G. Marland, J.-F. Soussana, T. R. Christensen, L. Heath, M. Apps, V. Alexeyev, J. Fang, J.-P. Gattuso, J. P. Guerschman, Y. Huang, E. Jobbagy, D. Murdiyarso, J. Ni, A. Nobre, C. Peng, A. Walcroft, S. Q. Wang, Y. Pan, and G. S. Zhou. Sectoral approaches to improve regional carbon budgets. *Climatic Change*, 88(3-4): 209–249, jan 2008. ISSN 0165-0009. doi: 10.1007/s10584-007-9378-5.
- F. Somma, J. W. Hopmans, and V. Clausnitzer. Transient three-dimensional modeling of soil water and solute transport with simultaneous root growth, root water and nutrient uptake. *Plant and Soil*, 202(2):281–293, 1998. doi: 10.1023/A:1004378602378.
- J. S. Sperry, V. Stiller, and U. G. Hacke. Xylem hydraulics and the soil–plant–atmosphere continuum: Opportunities and unresolved issues. *Agronomy Journal*, 95(6):1362–1370, 2003. ISSN 0002-1962. doi: 10.2134/agronj2003.1362.
- J. L. Stape, D. A. N. Binkley, M. G. Ryan, and A. Do Nascimento Gomes. Water use, water limitation, and water use efficiency in a eucalyptus plantation. *Bosque (Valdivia)*, 25:35–41, 2004. ISSN 0717-9200.
- K. Steppe, D. J. W. De Pauw, T. M. Doody, and R. O. Teskey. A comparison of sap flux density using thermal dissipation, heat pulse velocity and heat field deformation methods. *Agricultural and Forest Meteorology*, 150(7-8):1046–1056, 2010. doi: 10.1016/j.agrformet.2010.04. 004.

- E. Steudle. The cohesion-tension mechanism and the acquisition of water by plant roots. Annual Review of Plant Physiology and Plant Molecular Biology, 52:847–875, 2001. doi: 10.1146/annurev.arplant.52.1.847.
- G. Sun, A. Noormets, J. Chen, and S. G. McNulty. Evapotranspiration estimates from eddy covariance towers and hydrologic modeling in managed forests in northern wisconsin, usa. *Agricultural and Forest Meteorology*, 148(2):257–267, feb 2008. ISSN 0168-1923. doi: 10. 1016/j.agrformet.2007.08.010.
- G. Sun, A. Noormets, M. J. Gavazzi, S. G. McNulty, J. Chen, J. C. Domec, J. S. King, D. M. Amatya, and R. W. Skaggs. Energy and water balance of two contrasting loblolly pine plantations on the lower coastal plain of north carolina, usa. *Forest Ecology and Management*, 259(7):1299–1310, mar 2010. ISSN 0378-1127. doi: https://doi.org/10.1016/j.foreco.2009. 09.016.
- V. Suseela, R. T. Conant, M. D. Wallenstein, and J. S. Dukes. Effects of soil moisture on the temperature sensitivity of heterotrophic respiration vary seasonally in an old-field climate change experiment. *Global Change Biology*, 18(1):336–348, sep 2012. ISSN 1354-1013. doi: 10.1111/j.1365-2486.2011.02516.x.
- T. Tarin, R. H. Nolan, D. Eamus, and J. Cleverly. Carbon and water fluxes in two adjacent australian semi-arid ecosystems. *Agricultural and Forest Meteorology*, 281:107853, 2020. ISSN 0168-1923. doi: https://doi.org/10.1016/j.agrformet.2019.107853.
- B. Teodosio, V. R. N. Pauwels, S. P. Loheide, and E. Daly. Relationship between root water uptake and soil respiration: A modeling perspective. *Journal of Geophysical Research: Biogeosciences*, 122(8):1954–1968, aug 2017. ISSN 2169-8953. doi: 10.1002/2017jg003831.
- C. M. Tfwala, L. D. V. Rensburg, R. Schall, P. C. Zietsman, and P. Dlamini. Whole tree water use: Effects of tree morphology and environmental factors. *Ecological Indicators*, 102:366–373, jul 2019. ISSN 1470-160X. doi: 10.1016/j.ecolind.2019.02.054.
- P. E. Thornton, B. E. Law, H. L. Gholz, K. L. Clark, E. Falge, D. S. Ellsworth, A. H. Goldstein, R. K. Monson, D. Hollinger, M. Falk, J. Chen, and J. P. Sparks. Modeling and measuring the effects of disturbance history and climate on carbon and water budgets in evergreen needleleaf forests. *Agricultural and Forest Meteorology*, 113(1-4):185–222, 2002. ISSN 01681923 (ISSN). doi: 10.1016/S0168-1923(02)00108-9.
- M. Tomé, M. H. Almeida, S. Barreiro, M. R. Branco, E. Deus, G. Pinto, J. S. Silva, P. Soares, and R. Rodríguez-Soalleiro. Opportunities and challenges of eucalyptus plantations in europe: the iberian peninsula experience. *European Journal of Forest Research*, 140(3):489–510, feb 2021. ISSN 1612-4669. doi: 10.1007/s10342-021-01358-z.

- X. Tong, J. Zhang, P. Meng, J. Li, and N. Zheng. Ecosystem water use efficiency in a warmtemperate mixed plantation in the north china. *Journal of Hydrology*, 512:221–228, may 2014. ISSN 0022-1694. doi: 10.1016/j.jhydrol.2014.02.042.
- A. T. Trugman, N. J. Fenton, Y. Bergeron, X. Xu, L. R. Welp, and D. Medvigy. Climate, soil organic layer, and nitrogen jointly drive forest development after fire in the north american boreal zone. *Journal of Advances in Modeling Earth Systems*, 8(3):1180–1209, 2016. doi: https://doi.org/10.1002/2015MS000576.
- T. H. van den Honert. Water transport in plants as a catenary process. *Discussions of the Faraday Society*, 3:146, 1948. ISSN 0366-9033. doi: 10.1039/df9480300146.
- A. van Dijk. Extension to 3d of "the effect of line averaging on scalar flux measurements with a sonic anemometer near the surface" by kristensen and fitzjarrald. *Journal of Atmospheric and Oceanic Technology*, 19(1):80–82, jan 2002. ISSN 0739-0572. doi: 10.1175/1520-0426(2002) 019<0080:etoteo>2.0.co;2.
- A. I. J. M. van Dijk, H. E. Beck, R. S. Crosbie, R. A. M. de Jeu, Y. Y. Liu, G. M. Podger, B. Timbal, and N. R. Viney. The millennium drought in southeast australia (2001-2009): Natural and human causes and implications for water resources, ecosystems, economy, and society. *Water Resources Research*, 49(2):1040–1057, feb 2013. ISSN 0043-1397. doi: 10.1002/wrcr.20123.
- M. T. van Genuchten. A closed-form equation for predicting the hydraulic conductivity of unsaturated soils. *Soil Science Society of America Journal*, 44(5):892–898, sep 1980. ISSN 0361-5995. doi: 10.2136/sssaj1980.03615995004400050002x.
- E. van Gorsel, J. A. J. Berni, P. Briggs, A. Cabello-Leblic, L. Chasmer, H. A. Cleugh, J. Hacker, S. Hantson, V. Haverd, D. Hughes, C. Hopkinson, H. Keith, N. Kljun, R. Leuning, M. Yebra, and S. Zegelin. Primary and secondary effects of climate variability on net ecosystem carbon exchange in an evergreen eucalyptus forest. *Agricultural and Forest Meteorology*, 182-183: 248–256, dec 2013. ISSN 0168-1923. doi: 10.1016/j.agrformet.2013.04.027.
- P. Veiga. Allometric biomass equations for plantations of eucalyptus globulus and eucalyptus nitens in australia. Master's thesis, 2008.
- P. Verma, S. P. Loheide, D. Eamus, and E. Daly. Root water compensation sustains transpiration rates in an australian woodland. *Advances in Water Resources*, 74:91–101, dec 2014. ISSN 0309-1708. doi: 10.1016/j.advwatres.2014.08.013.
- R. A. Vertessy, F. G. R. Watson, and S. K. O_Sullivan. Factors determining relations between stand age and catchment water balance in mountain ash forests. *Forest Ecology and Management*, 143(1-3):13–26, 2001. ISSN 0378-1127. doi: 10.1016/s0378-1127(00)00501-6.

- D. Vickers, C. Thomas, C. Pettijohn, J. G. Martin, and B. Law. Five years of carbon fluxes and inherent water-use efficiency at two semi-arid pine forests with different disturbance histories. *Tellus B: Chemical and Physical Meteorology*, 64(1):17159, jan 2012. ISSN 1600-0889. doi: 10.3402/tellusb.v64i0.17159.
- J. A. Vrugt, M. T. van Wijk, J. W. Hopmans, and J. Šimunek. One-, two-, and three-dimensional root water uptake functions for transient modeling. *Water Resources Research*, 37(10):2457–2470, oct 2001. ISSN 0043-1397. doi: 10.1029/2000WR000027.
- T. J. Wardlaw. Eucalyptus obliqua tall forest in cool, temperate tasmania becomes a carbon source during a protracted warm spell in november 2017. *Sci Rep*, 12, 2022. ISSN 2045-2322. doi: 10.1038/s41598-022-06674-x.
- E. K. Webb, G. I. Pearman, and R. Leuning. Correction of flux measurements for density effects due to heat and water vapour transfer. *Quarterly Journal of the Royal Meteorological Society*, 106(447):85–100, jan 1980. ISSN 0035-9009. doi: 10.1002/qj.49710644707.
- D. A. White, M. Battaglia, D. S. Mendham, D. S. Crombie, J. Kinal, and J. F. McGrath. Observed and modelled leaf area index in eucalyptus globulus plantations: tests of optimality and equilibrium hypotheses. *Tree Physiology*, 30(7):831–844, 2010. ISSN 0829-318X. doi: 10.1093/treephys/tpq037.
- D. A. White, J. F. McGrath, M. G. Ryan, M. Battaglia, D. S. Mendham, J. Kinal, G. M. Downes, D. S. Crombie, and M. E. Hunt. Managing for water-use efficient wood production in euclyptus globulus plantations. *Forest Ecology and Management*, 331:272–280, 2014. doi: 10.1016/j.foreco.2014.08.020.
- D. A. White, C. L. Beadle, D. Worledge, and J. L. Honeysett. Wood production per evapotranspiration was increased by irrigation in plantations of eucalyptus globulus and e. nitens. *New Forests*, 47(2):303–317, 2016. ISSN 1573-5095. doi: 10.1007/s11056-015-9516-2.
- D. A. White, R. P. Silberstein, F. Balocchi-Contreras, J. J. Quiroga, D. F. Meason, J. H. N. Palma, and P. R. de Arellano. Growth, water use, and water use efficiency of eucalyptus globulus and pinus radiata plantations compared with natural stands of roble-hualo forest in the coastal mountains of central chile. *Forest Ecology and Management*, 501:119676, dec 2021. ISSN 0378-1127. doi: 10.1016/j.foreco.2021.119676.
- D. Whitehead and C. L. Beadle. Physiological regulation of productivity and water use in eucalyptus: a review. *Forest Ecology and Management*, 193(1-2):113–140, 2004.
- S. Wiesner, C. L. Staudhammer, C. L. Javaheri, J. K. Hiers, L. R. Boring, R. J. Mitchell, and G. Starr. The role of understory phenology and productivity in the carbon dynamics of longleaf pine savannas. *Ecosphere*, 10(4):e02675, apr 2019. ISSN 2150-8925. doi: 10.1002/ecs2. 2675.

- T. Wutzler, A. Lucas-Moffat, M. Migliavacca, J. Knauer, K. Sickel, L. Šigut, O. Menzer, and M. Reichstein. Basic and extensible post-processing of eddy covariance flux data with reddyproc. *Biogeosciences*, 15(16):5015–5030, 2018. ISSN 1726-4189. doi: 10.5194/ bg-15-5015-2018.
- X. Xu, D. Medvigy, J. S. Powers, J. M. Becknell, and K. Guan. Diversity in plant hydraulic traits explains seasonal and inter-annual variations of vegetation dynamics in seasonally dry tropical forests. *New Phytologist*, 212(1):80–95, 2016. ISSN 0028-646X. doi: 10.1111/nph. 14009.
- I. A. M. Yunusa, S. Zolfaghar, M. J. B. Zeppel, Z. Li, A. R. Palmer, and D. Eamus. Fine root biomass and its relationship to evapotranspiration in woody and grassy vegetation covers for ecological restoration of waste storage and mining landscapes. *Ecosystems*, 15(1):113–127, oct 2012. ISSN 1432-9840. doi: 10.1007/s10021-011-9496-9.
- J. C. Yuste, M. Nagy, I. A. Janssens, A. Carrara, and R. Ceulemans. Soil respiration in a mixed temperate forest and its contribution to total ecosystem respiration. *Tree Physiology*, 25(5): 609–619, may 2005. ISSN 0829-318X. doi: 10.1093/treephys/25.5.609.
- M. Zeppel, C. Macinnis-Ng, A. Palmer, D. Taylor, R. Whitley, S. Fuentes, I. Yunusa, M. Williams, and D. Eamus. An analysis of the sensitivity of sap flux to soil and plant variables assessed for an australian woodland using a soil - plant - atmosphere model. *Functional Plant Biology*, 35(6):509, 2008. ISSN 1445-4408. doi: 10.1071/fp08114.
- T. Zha, A. G. Barr, T. A. Black, J. H. McCaughey, J. Bhatti, I. Hawthorne, P. Krishnan, J. Kidston, N. Saigusa, A. Shashkov, and Z. Nesic. Carbon sequestration in boreal jack pine stands following harvesting. *Global Change Biology*, 15(6):1475–1487, 2009. doi: 10.1111/j.1365-2486.2008.01817.x.
- L. Zhang. Tree planting and catchment water yield. heartlands, towards sustainable land use in the murray-darling basin. Technical report, Fact Sheet No 5. CSIRO and MDBC, 2004.
- L. Zhang, W. R. Dawes, and G. Walker. Predicting the effect of vegetation changes on catchment average water balance. Technical report, Cooperative Research Centre for Catchment Hydrology - CSIRO Land and Water, 1999.
- L. Zhang, W. R. Dawes, and G. R. Walker. Response of mean annual evapotranspiration to vegetation changes at catchment scale. *Water Resources Research*, 37(3):701–708, mar 2001. ISSN 0043-1397. doi: 10.1029/2000wr900325.
- L. Zhang, T. Dowling, M. Hocking, J. Morris, G. Adams, K. Hickel, A. Best, and R. Vertessy. Predicting the effects of large-scale afforestation on annual flow regime and water allocation: an example for thegoulburn-broken catchments. Technical report, Cooperative Research Centre for Catchment Hydrology, 2003.

- L. Zhang, R. Vertessy, G. Walker, M. Gilfedder, and P. Hairsine. Afforestation in a catchment context: understanding the impacts on water yield and salinity. Technical report, Industry report 1/07, eWater CRC, Melbourne, Australia, CSIRO and eWater Limited, 2007.
- L. Zhang, F. Zhao, Y. Chen, and R. N. M. Dixon. Estimating effects of plantation expansion and climate variability on streamflow for catchments in australia. *Water Resources Research*, 47 (12), 2011. ISSN 0043-1397. doi: 10.1029/2011wr010711.
- L. Zhang, F. F. Zhao, and A. E. Brown. Predicting effects of plantation expansion on streamflow regime for catchments in australia. *Hydrology and Earth System Sciences*, 16(7):2109–2121, 2012. ISSN 1607-7938. doi: 10.5194/hess-16-2109-2012.